

IntechOpen

Complex Systems, Sustainability and Innovation

Edited by Ciza Thomas





COMPLEX SYSTEMS, SUSTAINABILITY AND INNOVATION

Edited by Ciza Thomas

Complex Systems, Sustainability and Innovation

http://dx.doi.org/10.5772/62659 Edited by Ciza Thomas

Contributors

Kazutoshi Fujihira, Rabha Ibrahim, Alex Kindler, Sorin Solomon, Natasa Golo, Dasheng Liu, Hiroshi Honda, Yu-Chen Lo, Ren Gui, Jorge Torres, Shwn-Meei Lee, Ciza Thomas, Rendhir RajendraPrasad, Minu Mathew

© The Editor(s) and the Author(s) 2016

The moral rights of the and the author(s) have been asserted.

All rights to the book as a whole are reserved by INTECH. The book as a whole (compilation) cannot be reproduced, distributed or used for commercial or non-commercial purposes without INTECH's written permission. Enquiries concerning the use of the book should be directed to INTECH rights and permissions department (permissions@intechopen.com).

Violations are liable to prosecution under the governing Copyright Law.

CC BY

Individual chapters of this publication are distributed under the terms of the Creative Commons Attribution 3.0 Unported License which permits commercial use, distribution and reproduction of the individual chapters, provided the original author(s) and source publication are appropriately acknowledged. If so indicated, certain images may not be included under the Creative Commons license. In such cases users will need to obtain permission from the license holder to reproduce the material. More details and guidelines concerning content reuse and adaptation can be foundat http://www.intechopen.com/copyright-policy.html.

Notice

Statements and opinions expressed in the chapters are these of the individual contributors and not necessarily those of the editors or publisher. No responsibility is accepted for the accuracy of information contained in the published chapters. The publisher assumes no responsibility for any damage or injury to persons or property arising out of the use of any materials, instructions, methods or ideas contained in the book.

First published in Croatia, 2016 by INTECH d.o.o. eBook (PDF) Published by IN TECH d.o.o. Place and year of publication of eBook (PDF): Rijeka, 2019. IntechOpen is the global imprint of IN TECH d.o.o. Printed in Croatia

Legal deposit, Croatia: National and University Library in Zagreb

Additional hard and PDF copies can be obtained from orders@intechopen.com

Complex Systems, Sustainability and Innovation Edited by Ciza Thomas p. cm. Print ISBN 978-953-51-2841-0 Online ISBN 978-953-51-2842-7 eBook (PDF) ISBN 978-953-51-6688-7

We are IntechOpen, the world's leading publisher of Open Access books Built by scientists, for scientists

3,750+

115,000+

International authors and editors

119M+

151 Countries delivered to Our authors are among the Top 1%

most cited scientists

12.2%

Contributors from top 500 universities



WEB OF SCIENCE

Selection of our books indexed in the Book Citation Index in Web of Science™ Core Collection (BKCI)

Interested in publishing with us? Contact book.department@intechopen.com

Numbers displayed above are based on latest data collected. For more information visit www.intechopen.com



Meet the editor



Dr. Ciza Thomas is currently working as a professor in Electronics and Communication Department of College of Engineering, Trivandrum, India. Her area of expertise is network security with research interest in the fields of information security, data mining, sensor fusion, pattern recognition, information retrieval, digital signal processing, and image processing. She has publications in more

than 40 International Journals and International Conference Proceedings. She has edited two books in the field of sensor fusion and published five book chapters in the field of network security and pattern recognition. She is a reviewer of more than ten reputed international journals including *IEEE Transactions on Signal Processing*, *IEEE Transactions on Neural Networks*, *International Journal of Network Security*, *International Journal of Network Management*, and *IEEE-John Wiley International Journal on Security and Communications Network*. She is a recipient of achievement award in 2010 and the e-learning IT award in 2014 from the Government of Kerala.

Contents

Preface XI

Chapter 1	Introduction to Complex Systems, Sustainability and Innovation 1 Ciza Thomas, Rendhir R. Prasad and Minu Mathew
Chapter 2	Measurement of the Communication Possibility of Service Requests for Multiservers in Parallel Connection in Cloud Computing Systems 21 Rabha W. Ibrahim
Chapter 3	System Control for Sustainability: Application to Building Design 37 Kazutoshi Fujihira
Chapter 4	Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic Robot Using Delayed Feedback Control 57 Dasheng Liu
Chapter 5	Quantitative Methods in System-Based Drug Discovery 85 Yu-Chen Lo, Ren Gui, Hiroshi Honda and Jorge Z. Torres
Chapter 6	Stochastic Agent-Based Simulation of the Role of Labor in the Fconomy 99

the Economy 99 Alexander Kindler, Natasa Golo and Sorin Solomon

Preface

This book on complex systems, sustainability, and innovation explores a broad set of ideas and presents some of the state-of-the-art research in this field. The book is triggered by ubiquitous applications of real-world complex problems. Complex systems may have many diverse and autonomous components that interact and create a functioning whole. In a complex system, it is difficult to know exactly how the individual components contribute to an observed behavior and the extent of each component's contributions. It is the interactions of the individual components that determine the emergent functionalities. This makes it difficult to understand and predict the behavior of complex systems.

In the technological front, significant progress has been made to increase the system performance and thereby the living standards. However, all innovations where their impact far exceeds the knowledge are identified as one of the quandaries of sustainability. This is because humans have through decades experimented and created ever-increasing complex systems that outperformed every system that existed at that time. However, as time passed by, the complexity has exponentially increased, and now, that complexity itself is the problem. Hence, a new age of innovations has to emerge with the main focus on user orientation and sustainability. An advanced dynamical understanding of complex systems and their dependence on the environment is essential for a long-term perspective and innovations supporting practical applications that aid in a sustainable society. Present studies need to focus on sustainability and innovation that include awareness of the consequences of complex systems. It is necessary to learn from the past, constructing causative narratives; at the same time, we need to learn for the future, which requires thinking differently and no longer linearly but in terms of alternatives.

Complex systems span the entire spectrum of science and technology including basic sciences to life sciences and medicine, to social, economic, and cognitive sciences, to engineering and computers. The book consists of six chapters. In Chapter I, the introduction to complex systems and their evolution are discussed. The knowledge of a complex system allows for sustainability when various innovations occur in the field. Chapter II to Chapter VI include the problems associated with various complex systems and the methods to address each one of them with the focus on sustainability and innovation. Chapter II includes a method to minimize the cost function of the cloud by looking at the communication possibility of service requests for multiserver in its parallel connection. Chapter III proposes a methodology for sustainable home design by applying control science. Chapter IV proposes stabilization control strategies to address the stabilization problems of periodic motions for the higher-order continuous-time chaotic systems. Chapter V discusses the quantitative methods for system-based drug discovery and design to engineer safer and more effective drugs. Chapter VI presents a platform for stochastic agent–based simulation of the role of labor in econo-

my. The intended audience of this book will mainly consist of researchers, research students, and practitioners in the field of complex systems and sustainability.

I would like to convey my appreciation to all the contributors of this book including the authors of the accepted chapters. My special thanks to the Publishing Process Manager, Ms. Martina Usljebrka, and other staff of InTech publishing for their kind support and great efforts in bringing the book to fruition. In addition, I also thank the reviewer, Dr. T. John Tharakan, for providing very critical and constructive comments to the introductory chapter of this book. His comments helped to tremendously improve the content of the chapter.

Prof. Ciza Thomas College of Engineering Trivandrum, India

Chapter 1

Introduction to Complex Systems, Sustainability and Innovation

Ciza Thomas, Rendhir R. Prasad and Minu Mathew

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/66453

Abstract

The technological innovations have always proved the impossible possible. Humans have all the time obliterated barriers and set records with astounding regularity. However, there are issues springing up in terms of complexity and sustainability in this context, which we were ignoring for long. Today, in every walk of life, we encounter complex systems, whether it is the Internet, communication systems, electrical power grids, or the financial markets. Due to its unpredictable behavior, any creative change in a complex system poses a threat of systemic risks. This is because an innovation is always introducing something new, introducing a change, possibly to solve an existing problem, the effect of which is nonlinear. Failure to predict the future states of the system due to the nonlinear nature makes any system unsustainable. This necessitates the need for any development to be sustainable by meeting the needs of people today without destroying the potential of future generations to meet their needs. This chapter, which studies systems that are complex due to intricateness in their connectivity, gives insights into their ways of emergence and the nonlinear cause and effects pattern the complex systems use to follow, effectively paving way for sustainable innovation.

Keywords: chaos, complex systems, complex networks, complexity theory, information systems security, sustainable networks

1. Introduction

Today, in every walk of life, we encounter complex networks, whether it is the Internet, which is a global system of interconnected computer networks; or the communication systems, which is a collection of individual communications networks, transmission systems, relay



© 2016 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. stations, tributary stations, and data terminal equipment usually capable of interconnection and interoperation to form an integrated whole; or the electrical power grids, which is an interconnected network for delivering electricity from suppliers to consumers; or the financial markets, which is any collection of traders, firms, banks, and financial exchanges; or the food web, which is a system of interlocking and interdependent food chains; or the metabolic networks, which is the complete set of metabolic and physical processes that determine the physiological and biochemical properties of a cell; or the social networks, which is a network of social interactions and personal relationships. The interacting units and intricate connections make these systems complex. Though these complex systems perform diverse functions, studies show that there are similarities in its structure.

Complexity has been defined in various ways in different fields. Moses [1] defines complex systems as "A system is complex when it is composed of many parts that interconnect in intricate ways." According to this definition, the number and types of parts of a system decide its complexity. The systems become complex when these parts are connected in a nonregular way. The information about the nature of interconnections is usually insufficient and this makes modeling difficult.

Rechtin and Maier [2] define a complex system to consist of a set of different elements so connected or related as to perform a unique function not performable by the elements alone. The complex system acts as "whole is greater than parts," which means a complete knowledge about the parts will not be sufficient to predict the working of a system as a whole. Parts show different behavior than when working together as a whole. The traditional scientific method of studying a system is to dissect a system into its parts, model them into a formal system, study individually, and then combine them to get the whole information about the system. This works on the assumption that the parts combined together will correctly give complete information about the system as a whole and the behavior of the system can be predicted completely. However, in the case of complex systems, it is not possible to get the complete information about the system just by combining parts, because information is generated, not in parts per se but in the intricate connections between them. The component parts of a complex system are more or less complex entities linked with one another and organized within an ordered structure of mutual interaction. It is not mandatory that the web of linkages is visible-instead, it may consist of cause and effect relations which originate by the way of communication. Hence, studying the behavior of complex systems, in effect, is the study of the system as a network of individual parts and the emergence of complex networks from the interconnection between individual parts.

Strogatz [3] describes the complications in studying complex systems. Complex systems show structural complexity because the nature of connection is intricate, with connections changing dynamically over time as in the case of World Wide Web, where pages and links are created and deleted all the time, making the evolution of network very dynamic and hence complex. These links are not homogenous and they show considerable diversion in terms of weights, direction, and signs. The nodes itself can be nonlinear dynamic systems since the state of nodes is varying in time. The nodes can be of diverse types as in the case of biochemical networks. Moreover, they affect each other and this interdependency result from networked

risks, which means a small disturbance caused by a component, can result in a catastrophic failure of the system. Global financial meltdown of 2008 can be cited as an example of this type of cascading failure. In today's highly interconnected world, coupling of different kinds of systems makes the effects of failure uncontrollable in a global scale. Any naturally occurring complex system has a tendency to evolve automatically and those networks are quite resistant to small initial variations causing a tremendous effect on the entire system.

Helbing [4] argues that global interdependent systems produced by intricate networking are prone to cascading global-level failures even when external shocks are absent. He discusses various anecdotes of failures and calls for a new approach called Global Systems Science for the study of these phenomena and gives a detailed description of drivers and examples of systemic instabilities.

Networks have interconnected the distributed, heterogeneous systems and offer a wide variety of applications that were never dreamt of. However, there are issues springing up in terms of complexity and sustainability in this context, which we were ignoring for long. Any creative change in a complex system poses a threat of systemic risks. The exploitation of the technologies has of late led to the verge of a complexity crisis. Due to its unpredictable behavior, these networks are vulnerable to a cascade of overload failures, which can in turn cause the entire or a substantial part of the network to collapse. It is the artificial or the man-made complex systems like the Internet that gets affected by cascading failures capable of disabling the network almost entirely. Even though humans have all the time obliterated barriers and set records with astounding regularity, the irony is that an innovation is always introducing something new, introducing a change, possibly to solve an existing problem, the effect of which is nonlinear. Failure to predict the future states of the system due to the nonlinear nature makes any innovation vulnerable. Hence, there is a high possibility that an innovation can make a system unsustainable. Many examples like invention of plastics, usage of fossil fuels, etc., can be cited especially from global climate change studies. A disturbing question arises in this context: Is there a contradiction existing between innovation and sustainability? However, innovation is the only way to progress, and hence in the case of complex systems, innovation has to happen in the backdrop of sustainability. This necessitates the need for any development to be sustainable by meeting the needs of people today without destroying the potential of future generations to meet their needs. Hence, the study of complex systems, stability of complex systems, and the interdependence of the complex system components is inevitable today and is discussed in detail in this chapter. The remaining part of this chapter is to study systems that are complex due to intricateness in its connectivity and provides insights into its ways of emergence, and the nonlinear cause and effects pattern the complex systems use to follow, effectively paving way for sustainable innovation.

This chapter is organized as follows. Section 2 gives an introduction to reductionism and discusses the insufficiency of reductive approach in addressing the behavior of chaotic systems. Emergent behavior and order in chaotic system is addressed in Section 3. Section 4 discusses the relation between chaotic systems and the structure of complex systems. Section 5 looks at the Internet as a typical example of complex system with information system security as the main focus. Self-organized criticality and interdependence and its effects are also discussed in this section. Section 6 concludes the chapter.

2. Reductive approach

Nature and all its phenomena are innately mathematical. That was the confidence history accepted for a long time. All of nature's intricate and beautiful phenomena were simplified into mathematical abstractions and discussed extensively. This notion got a firmer foothold when Sir Isaac Newton discovered a common law that governed an apple falling from a tree and the splendid "planetary motion." The Newtonian theory came with all intelligibility for an acceptable explanation. There was no notion of a complex system as all observable phenomena were represented by mathematical equations, more precisely, differential equations. Newtonian physics took up the center stage and paved way to reductionism. Reductionism is the practice of simplifying or minimizing a complex issue or condition, and it deals with the theory that every complex phenomenon can be explained by analyzing the most basic and simple component that exists during the phenomenon.

Reductionism is the first footstep of modern science. The reductionist philosophy holds that a complex system is nothing but the sum of its parts and a complex phenomenon can be explained completely in terms of the interactions between more fundamental phenomena. To put it simple, analysis is done by breaking down the larger system into pieces or individual parts and determining the connections between the parts. We credit this procedure to the mathematical concept of linearity and additivity. If we add the small individual parts together, the complexity will increase in a linear manner resulting in the same original complex system.

Furthermore, reductionism infers that there exists a point-to-point connection from the starting state to the mature state of a complex system. If we know the starting state, we have 100 percent predictability of the mature state. If we know the rule that governs a system to go from state 1 to state 2, we can apply the same rule and calculate the parameter changes from state 2 to state 3. This notion is the concept of extrapolation in mathematics. The concept nullifies going stage by stage to find out the mature state of a system. Instead, it demands that applying the same rule for as many iterations as needed, we can find out the mature state just by knowing the starting state of a system and vice versa [5].

With all these rules of reductionism and definition of procedures in terms of mathematics taken seriously, the Universe becomes a clockwork machine. This type of Universe (also known as the Laplace demon), made famous by Laplace, enjoyed an established set of rules. According to Laplace, it is possible to both calculate the entire history of the Universe and to predict its entire future if there is sufficient information about the current state of the Universe in addition to all of its fundamental and unchanging laws. However, scientists have always scrambled to find a mathematical model for the natural phenomena like weather, tides, clouds, and the human body.

Last century saw physicists investigating the dynamics of atoms and everything smaller than them, as if matter and all its wonderful properties can be explained by protons, neutrons, and going further deep into quarks and gluons [6]. Subsequent research in medical sciences devoted time in understanding and developing the tools for comprehending the workings of genes. The reductionist mindset has pervaded the field of molecular biology for half a century. Since the biological systems are composed solely of atoms and molecules, it is assumed that it should be possible to explain them using the physicochemical properties of their individual components, down to the atomic level, until and unless there is the influence of "alien" or "spiritual" forces.

For science, measurement of properties of minute particles was obviously the next step. However, physicists encountered variability, or noise as it is known in the reductive world, in those measurements of fundamental particles. Variability was stated as the discrepancy from the actual true measure. This noise represented system error and it is something that has to be avoided to get the true result. The way to reduce noise is to become more and more reductive. Reductionists argued that the closer we look at a system, the closer the obtained result will be to the actual result.

Human body is a scale-free system. Consider the tree-like bifurcating structure of the nervous system, or the circulatory system, or the pulmonary system. If the genes specify the bifurcations in a human body, there is not enough genome to code for this entire scale-free structure and all the activities that each cell is capable of. The neurons, arteries, and dendrites have a bifurcating structure which cannot be coded for and each individual part given a specific function. For a point-to-point connection to exist between the genes and the various things that a human is capable of, there is a lack of genes in our body. All the information is not coded in a single cell; it is coded in the interactions between different cells. Biological problems cannot be solved in a reductive manner [5]. We know the qualities of oxygen atoms and we know the qualities of carbon atoms. By reductive philosophy, we should be able to determine the qualities of carbon dioxide. Actually we cannot. We are familiar with the qualities of carbon dioxide only because we have observed them empirically. This does not concede with the additive philosophy. Brownian motion of molecules cannot be explained by reductionism. Reductive philosophy does not count for a role of chance. With randomness in a system, we will not be able to have absolute knowledge of its mature state from the initial state. In the reductive world, anything without an established absolute certainty becomes invalid.

3. Chaos in nature

Today, advances in both physics and biology have exploded the myths laid out by the reductionist approach. Accordingly, how inanimate and animate things work, and are made, cannot be explained simply by breaking it down into their components. Scientific developments obviously state that the specificity of a complex biological activity does not arise from the specificity of the individual molecules that are involved. Experiments have shown time and again that reality does not obey classical deterministic rules. General relativity is a good conceptual tool that describes many phenomena very accurately on fairly small scales. This being a deterministic rule, it is often argued that it serves as a limitation for reaching out to a unified theory. As opposed to Laplace's claim, mathematics begins to fall apart—as indicated by infinities and time anomalies (what they call variabilities) that pop up—when it is (mis) applied or when trying to "solve" the state of the entire Universe [7]. When some very fundamental questions on life and Universe come to the fore, the general relativity theory remained inept to answer. For as long as science inquired about the laws of nature, it has suffered ignorance about the sudden changes and the disorder in the atmosphere, the turbulence of the wind and the sea, the fluctuations in wildlife and human populations, and the variations of signals from the brain and heart. The most simple and interesting things could not be regulated in a simple reductive way. No consensus on a satisfactory answer has led to the emergence of the concept of chaos.

"As far as the laws of mathematics refer to reality, they are not certain, and as far as they are certain, they do not refer to reality."

– Albert Einstein

3.1. Chaos and complexity

In the scientific world, we realize chaos everywhere. A rising column of smoke breaks into wild swirls. A flag snaps back and forth in the wind. A dripping faucet goes from a steady pattern to a random one. Similar is the fluttering of leaves in the wind or the trajectory followed by a double rod pendulum or the ball in a pinball machine. Chaos appears in the behavior of weather, behavior of an airplane in flight, behavior of road traffic, behavior of oil flowing in underground pipes, population changes, or swarm formation of bees. Chaotic systems are not reductive. If a cloud does not rain, we cannot examine the cloud's component parts and come up with a solution. If we divide up a complex system into its pieces and add them up again, the resultant can be another system as the individual parts may add up differently. Knowing the initial state of a complex system will not result in a 100 percent predictability of the final state. The concept of extrapolation, linearity, and additivity does not apply to a complex system. The only way to know the mature state of a complex system is to go step by step. There are no patterns that the system follows over and over again. Chaos breaks across the lines that separate scientific disciplines. That realization has begun to change the way neurologists look at brain activity, the way business executives make decisions about insurance, the way astronomers look at the solar system, the way political theorists talk about the stresses leading to armed conflict, and the way economists look at organizational structure. Last 20 years, science saw an increase in specialization, looking for an order in all of nature. Dramatically, that hunt for specialization and order has reversed because of chaos. As Glieck [7] mentions in his book of *Chaos-Making a New Science*, "Where chaos begins, classical science stops." Chaos is a science of process rather than state, of becoming rather than being. Being a science of the global nature of systems, it has brought together thinkers from fields that had been widely separated.

Traditionally, when physicists saw complex systems, they tried to model it by complex functions and interactions. For complex results, they looked for complex causes. When they encounter a seemingly random relationship between an input to a system and its output, they simulated it by introducing artificial noise or error. They were ignorant of the fact that variability is intrinsic to the system and that quite simple mathematical equations could model systems every bit as violent as a thunder storm. Senge [8] defines complex systems based on cause and effect in it as "A system presents dynamic complexity when cause and effect are subtle, over time." The cause and effect pattern is not static over the time, but it changes along with the emergence of the system. A local effect may not give a clue about the global effect, so the interventions produce complex unpredictable effects. Tiny differences in input could quickly escalate and become overwhelming differences in output. This phenomenon of chaotic systems being highly sensitive to initial conditions is termed as the "butterfly effect" — a notion that a butterfly flapping its wings in Perkin would result in a thunder storm in New York a month later [7].

3.2. The butterfly effect

The butterfly effect was first observed by Edward Lorenz, a research meteorologist at the Massachusetts Institute of Technology. In 1960, Lorenz created a weather model on his computer [9]. It consisted of an extensive array of complex formulas, but the computer system had neither the speed nor the memory to manage a realistic simulation of the earth's atmosphere and oceans. Yet the program behaved quite like the real weather; it never seemed to repeat a sequence. One day Lorentz had let the program run on certain parameters to generate a certain weather pattern, and he wanted to take a better look at the outcome. Instead of starting the whole run over, he started midway through. To give the machine its initial conditions, he typed the numbers that had come up in the earlier run. This new run should have exactly duplicated the old as he inputted the exact results from the preceding run. The program had not changed. Yet the weather result diverged so rapidly from the previous run such that all resemblance to the old one was gone. It was not a malfunctioning of the system, but an infinitesimal difference in the input that scaled exponentially to cause an overwhelming difference in the output. In the computer's memory, six decimal places were stored: 0.506127. On the printout of the result, to save space, just three appeared: 0.506. Lorenz had entered the shorter, rounded-off numbers. And that puny little inaccuracy appeared to amplify and cause the entire system to swing out of whack. Lorenz weather toy was implementing the classical program. It used a purely deterministic system of equations and the small errors proved catastrophic. Given a particular starting point, the weather would unfold exactly the same way each time. Given a slightly different starting point, the weather at the next moment would unfold in a slightly different way. This is the new initial condition which results in totally different mature state values compared to the previous run. This variability continues in a scale-free manner to result in a weather pattern with no similarities. Now the question arises: what if we give Lorenz weather machine results accurate to six decimal places? Remember, the Heisenberg's uncertainty principle prohibits accuracy of measurement. An infinitesimal change in the input values can result in a totally different outcome. Therefore, the initial situation of a complex system cannot be accurately determined, and the evolution of a complex system can therefore not be accurately predicted. This shattered the classical theory that one can calculate the approximate behavior of the system given an approximate knowledge of a system's initial conditions and an understanding of the natural law [7]. As chaos theory progressed, the research community saw journals which compared strange dynamics of a ball bouncing on a table side by side with articles on quantum physics. The simplest systems now seemed to be unpredictable.

3.3. Patterns in chaos

The second law of thermodynamics can be stated as follows, "In an isolated system, entropy never decreases." The second law of thermodynamics is just another way of saying that all physical systems tend to move toward their most probable states, which shows up as increased entropy. Viewed in that context, entropy can be thought of as measuring disorder or randomness. The Universe is governed by entropy and constant changes and growing toward greater and greater disorder. Yet, chaos is not simply disorder. Chaos explores the transitions between order and disorder, which often occur in surprising ways. An order arises from the ever-growing disorder of the Universe — chaos and order together. Let us take a close look at nature in **Figure 1**.



Figure 1. Pattern of river networks, lightning bolts, trees, snowflakes, metal crystals, and pinecones.

River networks, lightning bolts, and trees have a self-similar pattern. Snowflakes, metal crystals, and pinecones have similar structural pattern. In all these structures, the same pattern gets repeated. They are said to have fractal-like features. Fractal, by definition, is a curve or geometrical figure, each part of which has the same statistical character as the whole. Fractallike patterns are found throughout the nature. It can be seen in the formation of ice crystals and veins in the human body, in leaves, in water drops, in air bubbles, etc. It is often been said that nature is an excellent designer and fractals can be thought of as the design principle nature follows when putting things together. Fractals are the best existing mathematical descriptions of many natural forms, such as coastlines, mountains, or parts of living organisms. Fractals are infinitely complex patterns that are self-similar across different scales and infinitely detailed.

Most physical systems of nature and many human artifacts are not regular geometric shapes of the standard geometry derived from Euclid. Fractal geometry offers almost unlimited ways of describing and measuring these natural phenomena. Scientists discovered that the basic architecture of a chromosome is tree-like; every chromosome consists of many "mini-chromosomes" and therefore can be treated as fractal. This self-similarity has also been found in DNA sequences. Even the rhythm of our heart is a fractal pattern as a function of time. Many image compression schemes use fractal algorithms to compress computer graphic files to less than a quarter of their original size. Recent study shows that the Universe and our brain cell follow a very similar pattern as shown in **Figure 2**.

Figure 2(a) is an image, which is only micrometers wide and shows three neuron cells and their connections in the brain of a mouse, whereas **Figure 2(b)** is an image, which is billions of light-years across and is a computer-simulated image, which shows how the Universe grew and evolved. Together, they suggest the surprisingly similar patterns found in vastly different natural phenomena.

Fractals are related to chaos because they are complex systems that have definite properties. Chaotic systems and fractals are both iterated functions. Each iteration takes the state of the previous iteration as an input to produce the next state. Although fractals can show



Figure 2. (a) Pattern of the brain cell. (b) Pattern of the Universe.

beautiful patterns, we cannot actually predict what their value is going to be of a specific iteration other than by calculating all the preceding iterations. So in a sense, fractals are just beautiful chaos.

The process of making a fractal is a type of self-ordering process. These self-organizing processes are very local in nature but lead to highly organized structures on very large scales. An order arises from the ever-growing disorder of the Universe. We cannot see the order in chaos if we are looking moment to moment. We cannot see order if we are managing individual behaviors. However, if we stand back far enough, if we wait over time, scale, or distance, we will observe the order that is in chaos, a pattern emerges from the otherwise disordered system.

3.3.1. The strange attractor

In a deterministic linear system, variability is not significant. There is a definite answer, a mathematical proof and solution for its characteristics and problems. A perfectly periodic system has the concept of a regular attractor. The damped pendulum has two invariant points; one is the point of minimum height and the other of maximum height. Due to the dissipation of energy, point of minimum height behaves as a regular attractor. The system tends to evolve toward the attractor. Into a structured linear system, if the amount of input force is increased, the system undergoes a transition to a chaotic system. The periodically repeating pattern previously present in the linear system breaks apart and gives rise to a pattern that never repeats. The system loses its predictability. A system in chaos is totally unpredictable from moment to moment. When converted into multidimensional space on high-speed computers, one could track many variables at once and could plot the movement of a system in chaos. The system is in total disorder when it is observed from moment to moment. Prediction of the next state is impossible. However, over a period of time, we come to realize that the system conforms to a boundary. There is an inherent shape that it never violates and never will it move out of a defined boundary. The plots that are based on nonlinear equations may appear to be random or chaotic but over many iterations will evolve to be simple and tend to produce patterns called strange attractors that confine to a boundary. These strange attractors need not be symmetrical always.

The behavior strange attractor can be seen from the first chaotic system discovered by Edward Lorenz, the Lorenzian (or chaotic) waterwheel. A chaotic waterwheel is just like a normal waterwheel except for the fact that the buckets leak. The entire waterwheel system is mounted under a water sprout. In the working of the waterwheel, water pours in from the top at a steady state and gives the system energy while water leaks out from each bucket at a steady state and removes energy from the system. Water trickles into the top bucket and immediately trickles out through the hole in the bottom at low speeds. With an increase in the inflow of water, the waterwheel will begin to revolve as the buckets fill up faster than they can empty. The heavier buckets containing more water on descending will let the water out, and when empty, it ascends on the other side to be refilled. The system is in a steady state; the wheel will continue to spin at a fairly constant rate. On increasing the input flow of water, increasing the system transitions into a chaotic behavior. Rather than spinning in one direction at a constant speed, the wheel will speed up, slow down, change directions, and



Figure 3. Chaotic patterns followed by the center of mass of the bucket in Lorenzian waterwheel with x, y, and z being the coordinates of the center of mass in (a) x-y plane, (b) x-z plane and (c) y-z plane [10].

oscillate back and forth, between the combinations of behaviors in an unpredictable manner. However, the entire system is within some bounds. Because the system never exactly repeats itself, the trajectory of a particular bucket never intersects itself. Instead it loops around and around forever in a strange pattern. It seems to be attracted toward a point in space. The system does not reach the ideal attraction point but is constantly pulled to it. There emerges an order in the chaotic system as shown in **Figure 3**. Hence, simply put, a chaotic system is a deterministic and unpredictable system.

4. Complex systems

Complex systems exhibit chaotic behavior that is characterized by extreme sensitivity to initial conditions, fractal geometry, and self-organized criticality [11]. The concept of sensitive dependence on initial conditions has been popularized in the "butterfly effect": minute differences in the initial conditions for such a system result in extremely different outcomes.

Fractal geometry refers to self-similarity, which implies that an object looks the same at any scale. Mathematically, self-similarity requires a power law distribution of representative objects. In other words, as the size of an object decreases, there is a corresponding power law increase in the number of objects of that size [12].

Self-organized criticality (SOC) is the concept that a dynamic system will move toward a critical, an emergent, or a metastable state, by natural processes [13]. A metastable system is in a delicate state of equilibrium where even a small change in conditions may precipitate a major change in the system. The chaotic nature of complex systems and its sensitivity to initial conditions make it prone to be in a metastable state and not in an equilibrium state.

The knowledge about the emergent properties of chaotic system and similarities across various natural systems opens up new possibilities. First, it boosts the effort to explore the secrets of the nature in the sense that these conspicuous similarities encourage to think about the common mechanisms connecting all these systems. Second, it gives a tool to study the chaotic system. Chaotic systems are highly complex systems due to the randomness and nonlinearity existing in these systems. Chaotic systems are usually defined as deterministic systems whose trajectories diverge exponentially over time. It involves few parameters and the dynamics of their values. Complex systems are essentially chaotic systems, and hence, all these properties are relevant to complex systems too. However, as complex systems comprise of many interacting parts, they are ever-growing dynamic systems. It always interacts with the environment. So, the scientific study of chaos is the root of the modern study of complex systems [14]. The big question is how we can study the complex systems based on the insights obtained from the study of chaotic systems.

Complex systems manifest themselves as complex networks because complexity stems from intricacies that exist in the interconnections. Hence, the study of complex systems is the statistical study of network matrices as graphs representing complex networks. Networks can be represented as regular or nonregular graphs. Regular graphs are those having regular topologies, and algorithms in graph theory are used to analyze their properties. However, many complex systems, for example, World Wide Web or social networks, do not follow a regular graph structure. There is randomness associated with the existence of a connection between two nodes in a complex network. The random graph as proposed by Erdös and Rényi [15] tries to capture this randomness. It is observed by Strogatz [3] that the small-world effect is exhibited in complex networks. When a rewiring is done in a regular lattice with some random connections, short paths are produced between any pair of nodes and the network is highly clustered than random graphs. Many complex systems like social networks show this property of high significance in the way people are connected with each other.

The quest for finding the underlying laws in the emergence of complex systems is a very important one. With the knowledge that a small change during the emergence plays a big role in forming the final form, the question arises as to whether there exists any simple law that resulted in the systems that exist now. This enquiry starts with the analysis of complex networks and tries to find the similarities in them. Barabási and Albert [16] have observed that the degrees of distribution among the nodes are dissimilar. Some nodes are highly connected than others. This means, if p_k denote the fraction of nodes that have k links where k is the degree and p_k is the degree distribution, the random graph model predicts a bell-shaped Poisson distribution for pk. However, in the case of real networks, p_k is highly skewed and decays much more slowly than Poisson's distribution. For instance, the distribution decays as a power law $p_k \sim k$ for the Internet backbone, metabolic reaction networks, the telephone call graph, and the World Wide Web. These networks are called "scale-free" networks by analogy with fractals, phase transitions, and other situations where power laws arise and no single characteristic scale can be defined. Though power law distribution is not universal, it is quite common among real networks. Hence, this result is quite significant when the laws for emergence of these networks are concerned. Turcotte [11] has shown that this emerges due to the existence of two simple rules during the emergence: the growth of network and its preferential attachment. This means that if the network expands continuously by the addition of new vertices and the new vertices attach preferentially to the sites that are already well connected, a power law distribution in degree distribution will occur. In complex systems improbable events are orders of magnitude more likely than events that follow the Gaussian distribution. Complex systems exhibit non-Gaussian outputs since the estimates and predictions of system's future behavior, particularly Gaussian estimates, formed by observations collected over short time periods provide an incorrect picture of large-scale fluctuations [17].

5. Internet as a complex system

After the Internet became widely available in the early nineties, it has enabled new services and transformed access to information, improved economic efficiency, and facilitated greater collaboration among individuals, businesses, and government. The Internet and the internetworking infrastructures underpin every aspect of our lives. More than 50% of the world's population now uses the Internet, a proportion that is growing every year. The Internet provides the supporting platform for an emerging digital economy, which depends on broadband networks and services like social media and cloud computing.

The computers and network which form the information system is an important and natural application domain for complex systems. Those systems are dynamic in nature, with continuously changing patterns of behavior, programs getting modified or removed or new programs included, users being removed or added, and configurations getting changed over time. The information system security problems are quite challenging due to the growing complexity and interconnected nature of these systems. It happens to be additionally complex with its multidimensionality as problems at different layers of the system are to be addressed. These problems include the physical layer security problems, network layer routing problems, application layer privacy, authentication and encryption problems, information security management problems, etc.

The Internet is identified as a critical enabler for sustainable development since it aids in unlocking human capabilities and enables communication and integration of the large information society. Forecasting the development of innovative and new technologies in this field is very difficult. The developments of the Internet were not anticipated by anyone. Most forecasts turn out to be too optimistic about the short-term introduction and too conservative about the societal consequences in the long term. Sustainability necessitates the reliability and resilience of the Internet if governments and businesses are to use the Internet to deliver services and for economic prosperity. Participation in the digital economy, including cloud computing, requires uninterrupted and safe access to high-speed networks.

5.1. Information systems security

The Internet evolves at a rapid rate in terms of the number and variety of the individual components and the possible applications that were never expected at all. The main reason for instability of the Internet is that many of the engineering innovative applications for ease of use are not being matched by engineering for ease of security. Now, large number of people, even with little technical knowledge or skill, all over the world are using the power of application software and new hardware to work with greater effectiveness and efficiency. However, it is difficult to securely configure and operate many of these services or products. This mismatch between the knowledge needed to operate a system and that needed to keep it secure results in increasing the number of system exploits. Additionally, the Internet has evolved so rapidly that any service provider or application developer will be concentrating on reducing the time taken to bring a new innovation to market. Hence, security is given the least priority in a competitive market for new products. This results in a high chance of exploits and attacks on the Internet that involve direct participation of many components including people and systems (software and hardware including networks). Hence, there is a fundamental relationship between the networking components and network security problems.

Additionally, the complexity of the Internet adds to the risk in the system usage as it allows attackers to find novel ways to gain improper access to the information on computer systems or intrude into network systems. The security analysis in such a complex context has to be investigated rather than depending only on heuristics. This paradigm shift leads to employ complexity theory to model the interactions of networking components in security problems. In these decision-making security models, the decision makers play the role of either the attacker or the defender. They have conflicting goals, with attacker attempting to breach security of the system so as to interrupt the services offered or damage the information available on systems, whereas a defender takes either preventive or corrective measures to enhance the system performance.

There are situations of uncertainty in predicting the behavior, interaction, and intention of both the malicious attackers and the defenders. It is necessary to understand the interaction between the attackers and defenders. Also, there is an increasing interaction and collaboration between various organizations that are interconnected as there are security interdependencies among them. This is because the vulnerability of one organization may result in compromises for others and in cascading failures affecting many nodes. The equilibrium analysis of the entire system provides insights on the decisions like security investment and patch management. Complexity theory aids in analyzing and investigating the decision process by the proper prediction of the behavior of networking components.

It is therefore necessary to give a top priority to analyze, model, and better understand the interconnected large-scale complex systems due to the increasing vulnerability of these network infrastructures. The Leontief-based input-output infrastructure model [18] can be made use of to represent the added and ever-increasing interdependence, intraconnectedness, and interconnectedness among the networking infrastructures. Additionally, the entire system becomes complex due to its architecture involving decision-making at multiple levels. The Leontief-based model can be used in managing the vulnerabilities of these complex network systems and the threats on these infrastructures in a more cost-effective manner. This will ensure the continued performance of these complex infrastructures. This will enable the total protection of the information stored in the systems or the information that gets communicated over the network.

Sustainable development has been a focus of international public policy since the Earth Summit in 1992. It identifies that economic growth, social inclusion, and environmental sustainability are essential to achieve human development that meets the needs of the present

without compromising the ability of future generations to meet their own needs. Any innovation happening in a field becomes sustainable if it can meet the needs of people without adversely affecting the future generations to meet their needs. Thus, sustainable innovation has a close linkage with value choices. Any naturally occurring complex system has a tendency to evolve automatically, and those networks are quite resistant to small initial variations causing a tremendous effect on the entire system. It is only the artificial or the man-made complex networks like the Internet that get affected by cascading failures capable of disabling the network almost entirely.

5.2. Self-organized criticality

As many vital systems for the existence of human beings follow chaotic pattern, the quest for finding out the measure of the stability and predictability of a complex system is a very active research field. The stability of the Internet system is getting affected both from the internal and external threats. Internal threats stem from the growing instability inherent in the system. Self-organized criticality introduced by Bak et al. [13] addresses this issue. SOC is a property of dynamic systems to organize its microscopic behavior to be spatial (and/or temporal) scale independent. SOC is typically observed in slowly driven nonequilibrium systems with extended degrees of freedom and a high level of nonlinearity. Phenomena of strikingly different backgrounds were claimed to exhibit SOC behavior: sand piles, earthquakes, forest fires, rivers, mountains, cities, literary texts, electric breakdown, motion of magnetic flux lines in superconductors, water droplets on surfaces, dynamics of magnetic domains, growing surfaces, human brains, etc. [19]. SOC models are useful in analyzing systemic instabilities inherently present in the system. Growth, i.e., addition of the new components to the system, is a ubiquitous feature of the complex system. SOC models study whether this growth will reach critical state and in which further addition of a component leads to failure of the system. This scenario may be in direct relation to the issue of sustainability of the Internet system. Many concerns of the Internet usage in modern life are the unprecedented growth that raises the concerns of sustainability. The studies are conducted using SOC model to analyze the growth dynamics of these systems and effect of human interventions in the system. For example, the effect of introducing a new species into an ecosystem [20], external threats to the system, attack from outside targeting a random member or targeting a special member of the system, etc.

Peculiar structure existing in the Internet influences the effectiveness of these attacks. In many complex systems like social networks or the Internet, very few nodes are rich in their connections, whereas most of the nodes are poor. This gives rise to the formation of hubs in these kinds of networks. These hubs are important and the significance of their security is integral to the stability of the whole system. Albert et al. [21] have studied about the attack tolerance of scale-free network and have shown that scale-free networks are more tolerant to random attacks. Change in diameter (longest geodesic) is analyzed for both random attacks and targeted attacks, and their study shows that scale-free network is less resilient against targeted attack. This shows that the nature of possible attack and the resultant failure that can happen are highly important in systems like the Internet. A similar result is shown by Xiao et al. [22] when they examine the robustness of communication networks under intentional attacks from a complex network point of view. They have found that incomplete information

may significantly degrade the efficiency of intentional attack, especially if a big hub is missed. After analyzing local information–based distributed attacks such as greedy sequential attack, coordinated attack, and lower-bounded parallel attack, it is shown that distributed attacks can be highly effective, sometimes almost as efficient as an accurate global information–based attack. This observation shows that attack is possible even when complete information about a hub is not available.

The question of how effectively information can be spread in a complex network or conversely how fatal a malicious node or the information it spreads is very significant. In this regard, Kitsak et al. [23] investigated on the question of which nodes can be the most influential spreaders in a complex network. The most efficient spreaders were identified by the k-shell decomposition analysis to be those located within the core of the network. When multiple spreaders are considered simultaneously, the distance between them becomes the crucial parameter that determines the extent of the spreading. Basaras et al. [24] explore the possibility of blocking virus propagation in complex networks. They found that removing edges is more practical than removing nodes to prevent the spread of a malware infection. Identification of edges that have to be removed is discussed and they propose an algorithm that detects critical connections based on their diffusion capabilities, namely, critical edge detector (CED).

Due to the nonlinear nature of complex networks, a failure starting at some point can cause unpredictable damage to the whole system. An attack can cascade through the nodes too. Lai et al. [25] review two problems in the security of complex networks: *cascades of overload failures on nodes and range-based attacks on links*. They argue that many scale-free networks are more sensitive to attacks on short-range than on long-range links because short-range links tend to link together highly connected nodes, while long-range links tend to connect nodes with very few links. Another finding is that the small-world phenomenon in these scale-free networks is caused by short-range links. The removal of a node with a tremendously large load is likely to significantly affect the loads at other nodes. There is hence a chance to trigger a sequence of overload failures. Thus, in networks with some degree of randomness, the distribution of loads and the distribution of links are highly correlated.

Buldyrev et al. [26] developed a framework for understanding the robustness of interacting networks that are subject to cascading failures. They have presented the analytical solutions for finding the critical fraction of nodes that, on removal, will lead to a cascading failure and to a complete fragmentation of two interdependent networks. It is also found that a broader distribution with the same average degree implies that there are more low-degree nodes. The advantage of a broad distribution for a single network becomes a disadvantage for interdependent networks as the low-degree nodes are more easily disconnected.

5.3. Interdependence and its effects

Complexity is multiplied when one complex network intricately depends on other complex networks. The nonlinearity in the effect discussed so far makes most of the systems vulnerable to even the smallest perturbation. In a highly networked world, interdependence causes many

uncertainties. The effects of climate system on financial system, social network on political system, power grid on the Internet, etc., are some of the examples. Buldyrev et al. [26] discuss about the problem of interdependence failure. Vespignani [27] analyzes the number of nodes that must fail before network is totally fragmented. Using percolation theory it is shown that the number of nodes is very less for interdependent networks compared to isolated networks. Chen et al. [28] investigate the emergence of extreme events in interdependent networks and found that when the number of network layers and/or the overlap among the layers is increased, extreme events can emerge in a cascading manner on a global scale. Thorough analysis has been carried out to understand the emergence of global-scale extreme events based on the concept of effective betweenness. A cost-effective control scheme to suppress the cascading process of extreme events so as to protect the entire multilayer infrastructure against global-scale breakdown is undertaken in his work by increasing the capacity of very few hubs. Buldyrev et al. [26] developed a framework to understand the robustness of inter-acting networks that are subject to cascading failures.

6. Conclusion

The possibility of studying systems that are complex gives insights into the ways of emergence and the nonlinear cause and effects pattern that are followed. Most natural systems follow a complex structure. The interdependence of seemingly disparate systems is getting more evident with a tiny change in one parameter producing disastrous effect in many interconnected systems. A system becomes unsustainable when it causes those effects, which are never expected out of it. This is exactly because of the inability to predict the effect of a system on other systems.

Sustainability is crucial in organizational and technological innovations that yield both bottom-line and top-line returns. Sustainability in a sense is being environment friendly with a reduced consumption of resources or, in simple terms, by reducing the inputs used by a system, in terms lowering the cost of production or innovation. Additionally, the entire process generates extra revenues from better products or enables to create new businesses. In fact, it is smart to treat sustainability as innovation's new frontier. Hence, complex systems are to be taken as a whole for a long-term analysis. Complexity theory, which studies about the emergence of systems with a "whole is greater than part" approach, is the very suitable tool to attack the effects.

Research challenges in the field of complex systems are about creating models and finding metrics that can capture the true nature of any real-world complex system. Representing complex systems as a network provides a powerful tool to decipher the secrets of their complexities. This inquiry is yet another epoch of comprehending the difficult riddles nature poses before humans, the most intelligent life form of the Universe. Maybe, victory in this through a proper understanding of the system gives the ability to innovate and sustain.

Finally poised between chaos and order, sustainability, and innovation, chaos theory has an edge that promises to clarify the simple and yet mysteriously complex nature of this Universe

and all its beings and phenomena. Its ramifications are the information available, strong claims made, more lucid answers sought and driven by the sustainability of nature, and the innovative and problem-solving capacity of humans. Driven by the need for better problem solving, we continue to structure knowledge, recognize new problems, and create new knowledge with innovation at the center to sustain the flow.

Author details

Ciza Thomas*, Rendhir R. Prasad and Minu Mathew

*Address all correspondence to: cizathomas@gmail.com

College of Engineering Trivandrum, Thiruvananthapuram, Kerala, India

References

- Moses, J. (2000). Complexity and flexibility. Quoted in Ideas on Complexity in Systems Twenty Views, by JM Sussman. February. http://web. mit. edu/esd, 83.
- [2] Rechtin, E., & Maier, M. W. (2010). The Art of Systems Architecting. CRC Press, UK, ISBN 9781420058529 - CAT# E0440.
- [3] Strogatz, S. H. (2001). Exploring complex networks. Nature, 410(6825), 268-276.
- [4] Helbing, D. (2013). Globally networked risks and how to respond. Nature, 497(7447), 51–59.
- [5] Stanford University. "Chaos and Reductionism". WeTube. WeTube, 01 Feb. 2011. Web. 30 Sept. 2016.
- [6] Bhat, R., & Salingaros, N. (2013). Reductionism Undermines Both Science and Culture. New English Review Press.
- [7] Gleick, J. (1987). Chaos: Making a New Science. Viking Press, New York.
- [8] Senge, P. M. (2006). The Fifth Discipline: The Art and Practice of the Learning Organization. Broadway Business, ISBN 0-385-51725-4 (second edition).
- [9] Lorenz, E. N. (1963). Deterministic nonperiodic flow. *Journal of the Atmospheric Sciences*, 20(2), 130–141.
- [10] Wang, X., & Zhang, H. (2013). Bivariate module-phase synchronization of a fractionalorder Lorenz system in different dimensions. *Journal of Computational Nonlinear Dynamics*, 8(3), 031017 (7 pages) Paper No: CND-12-1202; doi: 10.1115/1.4023438
- [11] Turcotte, D. L. (2006). Modeling geocomplexity: "A new kind of science". Geological Society of America Special Papers, 413, 39–50.
- [12] Mandelbrot, B. B. (1967). How long is the coast of Britain. Science, 156(3775), 636-638.

- [13] Bak, P., Tang, C., & Wiesenfeld, K. (1987). Self-organized criticality: An explanation of the 1/f noise. *Physical Review Letters*, 59(4), 381.
- [14] Bar-Yam, Y. (1997). Dynamics of Complex Systems (Vol. 213). Addison-Wesley, Reading, MA.
- [15] Erdös, P., & Rényi, A. (1959). On random graphs, I. Publicationes Mathematicae Debrecen, 6, 290–297.
- [16] Barabási, A. L., & Albert, R. (1999). Emergence of scaling in random networks. *Science*, 286(5439), 509–512.
- [17] Herbert, B. E. (2006). Student understanding of complex earth systems. *Geological Society of America Special Papers*, 413, 95–104.
- [18] Leontief, W. W. (1986). Input-Output Economics. Oxford University Press, Oxford, England.
- [19] Jensen, H. J., & Criticality, S. O. (1998). Emergent complex behavior in physical and biological systems. *Cambridge Lecture Notes in Physics*, 10.
- [20] Wang, Y., Fan, H., Lin, W., Lai, Y. C., & Wang, X. (2016). Growth, collapse, and selforganized criticality in complex networks. Scientific Reports, 6, [24445], pg.1-12, DOI: 10.1038/srep24445
- [21] Albert, R., Jeong, H., & Barabási, A. L. (2000). Error and attack tolerance of complex networks. *Nature*, 406(6794), 378–382.
- [22] Xiao, S., Xiao, G., & Cheng, T. H. (2008). Tolerance of intentional attacks in complex communication networks. *IEEE Communications Magazine*, 46(1), 146–152.
- [23] Kitsak, M., Gallos, L. K., Havlin, S., Liljeros, F., Muchnik, L., Stanley, H. E., & Makse, H. A. (2010). Identification of influential spreaders in complex networks. *Nature Physics*, 6(11), 888–893.
- [24] Basaras, P., Katsaros, D., & Tassiulas, L. (2015, June). Dynamically Blocking Contagions in Complex Networks by Cutting Vital Connections. In: 2015 IEEE International Conference on Communications (ICC) (pp. 1170–1175). IEEE, London.
- [25] Lai, Y. C., Motter, A. E., & Nishikawa, T. (2004). Attacks and cascades in complex networks. In: *Complex Networks* (pp. 299–310). Springer, Berlin Heidelberg.
- [26] Buldyrev, S. V., Parshani, R., Paul, G., Stanley, H. E., & Havlin, S. (2010). Catastrophic cascade of failures in interdependent networks. *Nature*, 464(7291), 1025–1028.
- [27] Vespignani, A. (2010). Complex networks: The fragility of interdependency. *Nature*, 464(7291), 984–985.
- [28] Chen, Y. Z., Huang, Z. G., Zhang, H. F., Eisenberg, D., Seager, T. P., & Lai, Y. C. (2015). Extreme events in multilayer, interdependent complex networks and control. Scientific Reports, 5, 17277, pp.1–13, http://doi.org/10.1038/srep17277

Measurement of the Communication Possibility of Service Requests for Multiservers in Parallel Connection in Cloud Computing Systems

Rabha W. Ibrahim

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/65285

Abstract

Newly, growing amount of data-demanding applications arrangement with continuous fluctuating data substances, have been investigated by many researchers recently. In these applications, the underlying data management system must support new types of the space-time changing that indicates to the paths of the cloud computing system (CCS). The time-space changing causes change in the dimension of data and, consequently, in the CCS. One of the solutions regarding this case is suggesting an integrated cloud computing system (ICCS). In this effort, we introduce a new ICCS, based on fractional formal operators, taking into account the symmetrical delay in it. This model is useful for higher dimensional data, moving data, and chaos data. Moreover, we employ a fractional differential method to discover the paths (outcomes) of the system by minimizing the cost function. The proposed system delivers a sequence of paths that converge to the optimal path. The theoretical technique is supported by the applications.

Keywords: fractional calculus, fractional formal, fractional dynamical system, cloud computing

1. Introduction

Cloud computing system (CCS) is the technique industries and organizations use to do their trades. In that dynamically accessible and visualized resources are delivered as a service over the Internet. This system constructs a new type of chance for organizations. CCS is developing as one of the most important factors for the business industry: it can convert the traditional industrial business prototypical, aid it to support product improvement with business



© 2016 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. approach, and construct able manufacturing networks that reassure effective cooperation. Various categories of CCS adoptions in the industrial area have been studied with direct adoption utilizing technologies and manufacturing. CCS has been in some of the important parts of developed Information Technology (IT). In CCS, entirety is preserved as a service such as platform, infrastructure, and software. These services express a layered system assembly for CCS. There are different kinds of CCS: private (single-tenant setting), public (idea of sharing), hybrid (mixed internal and external clouds), and community (multitenant setting shared by numerous organizations) clouds. The integrated CCS may refer to one of them. These cloud services are permeating as a particular plug of access. Various types of placement simulations outfit various states. The main study in CCS is to minimize the cost of the cloud, keeping the good and stable quality with less time [1–5].

The cost assistance of accepting clouds in a characteristic developed prize can be manifold. The savings acquired from the exclusion of selected functions that were necessary in customary Information Technology (IT) can be important. With cloud paths (solutions), some applications may be distributed with the company's Information Technology (IT) region along CCS technologies. Thus, the Information Technology (IT) control can create the modification that occurs seamlessly to reduce the cost. scheming is significant for creating schedules and controlling decisions. Cost itemization analysis, finding the developed movement, adaptive cost scheming, clearness of feeding, and billings are imperative concerns. Virtualization states to the idea of rational resources from their fundamental physical features in order to progress and enhance flexibility and decrease cost. Integrated cloud computing system (ICCS) platform paths show a significant character in converting prize systems, funding to cost reduction [1, 6–9].

A diversity of smart phones and handheld computers prepared with a GPS and further sensory devices previously exists for clients. A collective sensor that can discover and report separate motivations plots a sequence of data arguments in a higher dimensional data (HDD) time-space (one time dimension and n data dimensions). Consequently, to improve a practical database application, one requires to examine the following categories: the nature, the presentation, the store, the management, the environment, and the support of data. The challenging phase of HDD organization is considerably more obvious in training, particularly when inadequate recognition, such as intervals, reduces the frequency of accelerators and GPS interval. The HDD is adopted to deal with partial capacity, highly dynamic clients, and partial scalability to larger data sets. Recently, numerous application path simulations have been presented. These models suggested the point-time and location of data in the information space. If the item transfers farther, it reports its new position and changes the inform threshold for the next apprise [10–14].

Fractional calculus is a field of various studies with applications not only in mathematics but also in other sciences, engineering, economics, and social presentations. It agrees with differential and integral operators, including random powers: real and complex [15, 16]. The essential benefit of the constructing technique of fractional differential equations (ordinary and partial) in methodical forming is their nonlocal property. It is standard that the normal derivative is a local, linear operator, while the derivative of arbitrary order is nonlocal and

nonlinear. This proceeds that the successive recognized by a system is not only susceptible upon its present proper but also reliable upon all of its ancient formals. This additionally defined why fractional calculus has industrialized more and more extensively in mechanical, engineering, information theory, and industrial areas. Fractional calculus is received in information theory [17].

In this study, we impose a new ICCS, based on fractional formal operators, taking into account the symmetrical delay in it. This model is useful for higher dimensional data, moving data, and chaos data. Moreover, we employ a fractional differential method to discover the paths (outcomes) of the system by minimizing the cost function. The proposed system delivers a sequence of paths that converge to the optimal path. The theoretical technique is supported by applications.

2. Seeking

In this section, we illustrate the algorithm that we use to minimize the cost of the Information Technology (IT) in the cloud. Here, we suppose that the agents have nonhomogeneous options and pay the cost in order to translate from a position to another one in the state space $X \in [0, 1]$, where X is the level of utilizing Information Technology (IT) of the CCS. All agents have the common cost function Φ to have the request χ . Every user applies his optimal decision on the utility of the CCS by minimizing the cost function. Thus, the issue for each user is the minimized control function U(t,X).

The method is based on a new fractional differential transform (FDT). It is well known that this method has solved many problems not only in mathematics but also in physics and engineering (see [18–20]). We need the following fractional formal: the Riemann-Liouville operators are defined by the formal

$$D_s^{\nu}\wp(t) = \frac{d}{ds} \int_0^t \frac{(t-\varsigma)^{-\nu}}{\Gamma(1-\nu)} \wp(\varsigma) d\varsigma, \tag{1}$$

which coincides with the fractional integral operator

$$I_a^{\nu}\wp(t) = \int_a^t \frac{(t-\varsigma)^{\nu-1}}{\Gamma(\nu)} \wp(\varsigma) d\varsigma,$$
⁽²⁾

such that $v \in (0, 1)$ and \wp is a continuous function on $J = [0, T], T < \infty$. We use the notation $I_0^{\nu} \wp(t) = I^{\nu} \wp(t)$.

In this effort, we deal with the FCCS of the form

minimize:

$$\frac{\partial}{\partial t}U(t,\chi) = \Phi(t,\chi(\tau_0 t), D^{\nu}\chi(\tau_1 t), D^{2\nu}\chi(\tau_2 t), ..., D^{n\nu}\chi(\tau_n t), I^{\nu}\phi(t)),$$
Initial condition : $(\chi_i(\tilde{t}) = y_i(\tilde{t})), \forall i = 1, ..., n, \tilde{t} < t \in J,$
(3)

where

$$\phi(t) := \phi(t,\varsigma,\chi(\rho_0\varsigma), D^\nu\chi(\rho_1\varsigma), D^{2\nu}\chi(\rho_2\varsigma), \dots, D^{n\nu}\chi(\rho_n\varsigma)), \quad t \ge 0, \varsigma \le t,$$
(4)

is the objective function (we suggested integrable function in order to make the minimization of the cost function), $\chi = (\chi_1, ..., \chi_n)^T$ is the request function for n – agent in the cloud, and Φ is the cost function of the cloud system. We aim for a given company a degree of CCS in its present operation, $X(\tilde{t}) = \chi(\tilde{t}) = y(\tilde{t}) \in [0,1]$. Therefore, $\chi = 0$ is represented to a company that is not yet translated to advanced position. Therefore, the dynamic of the company's position develops depending on Eq. (3). We proposed the system (3) in terms of the fractional formal operator to include the delay in the CCS. The objective function deals with optimal state of the usage of the Information Technology (IT) in the cloud system depending on the cost.

CCS influences notions from utility computing to deliver elements the services used. Such elements are at the fundamental of the public cloud paid-per-use representations. Moreover, measured services are a critical part of the reactive loop in autonomic computing, agreeing services to scale on-demand and to achieve automatic disappointment recovery. Also, CCS delivers the tools and technologies to measure data/compute demanding parallel applications with much more reasonable prices compared to traditional computing system techniques. Our aim is to minimize (3). Therefore, we proceed to introduce a solution for the system (3). In this section, we introduce a numerical solution for Eq. (3) by using a new technique based on the fractional formal concept. The CCS (3) is equivalent to the problem

$$\Phi(t, \chi(\tau_0 t), D^{\nu} \chi(\tau_1 t), D^{2\nu} \chi(\tau_2 t), ..., D^{n\nu} \chi(\tau_n t), I^{\nu} \phi(t)) = 0.$$
(5)

An arbitrary function $\chi(t)$ can be formulated in terms of fractional Taylor series (see [21]) about the point t_0

$$\chi(t) = \sum_{\kappa=0}^{\infty} \frac{(t-t_0)^{\kappa\nu}}{(\nu\kappa)!} [D^{\kappa\nu}\chi(t)]_{t=t_0}$$

$$:= \sum_{\kappa=0}^{\infty} (t-t_0)^{\kappa\nu} \Lambda_{\nu}(\kappa),$$
(6)
where $(\nu\kappa)! = \Gamma(\nu\kappa + 1)$, $D^{\kappa\nu}\chi(t) = (D^{\nu}...D^{\nu})_{\kappa - times}\chi(t)$ and

$$\Lambda_{\nu}(\kappa) = \frac{\left[D^{\nu\kappa}\chi(t)\right]_{t=t_0}}{(\nu\kappa)!}, \nu \in (0,1],$$
(7)

is FDT of $\chi(t)$.

Equation (3) represents various types of fractional and integer control systems such as the Hamilton Jacobi Bellman equation, Euler equation, Fokker-Planck equation, and Hybrid systems. The major important advantage of FCCS is their nonlocal property. The CCS (integer case) is a local, linear system, while the FCCS is a nonlocal, nonlinear system. Therefore, the subsequent formal of a system is simulated by its current and previous positions (this asset is beneficial in CC). Thus, fractional formal has developed progressively into practical and manufacturing areas. A virtuous consequence is attained in FCCS by minimizing the time of utility function ($\chi(t)$) and the cost function (Φ) of the cloud. The change in the quality of the cloud may calculate the difference in the promised quality (Q_n) and the recent quality (Q_r)

$$\Delta Q = Q_p - Q_r. \tag{8}$$

We accept that cost, request, profit, and income functions are continuous and different with respect to time. The request is dwindling in the cost for the service and growing in quality delivered. The CCS is expected to determine its request for Information Technology (IT) services for any assumed value and quality. The base flat of quality essential to function the system is zero, and the base flat of request is at zero quality and cost. The request of the internal customer is such that at a satisfactorily high cost, there will be no feeding the services. Therefore, the quantity of the CCS can be expended as a function of the amount and quality.

We have the following properties:

Proposition 2.1. Suppose that $\theta(\kappa)$ and $\Lambda(\kappa)$ are FDTs of $\theta(t)$ and $\chi(t)$, respectively, then

1. If
$$\theta(t) = \chi(\rho t)$$
, then $\Theta_{\nu}(\kappa) = \rho^{\kappa} \Lambda_{\nu}(\kappa)$.

2. If
$$\theta(t) = D^{m\nu}\chi(\rho t)$$
, then $\theta_{\nu}(\kappa) = \frac{(\nu(m+\kappa))!}{(\nu\kappa)!}\rho^{m+\kappa}\Lambda_{\nu}(m+\kappa), m \in \mathbb{N}$

3. If
$$\theta(t) = I^{\nu} \chi(\rho t)$$
, then $\Theta_{\nu}(\kappa) = \frac{(\nu(\kappa-1))!}{(\nu\kappa)!} \rho^{\kappa-1} \Lambda_{\nu}(\kappa-1)$.

for $\rho \in (0,1)$

Proof: From Eq. (7), we have

$$D^{\nu\kappa}\theta(t) = D^{\nu\kappa}\chi(\rho t) = \rho^{\kappa}D^{\nu\kappa}\chi(\rho t);$$
(9)

therefore, we get

$$D^{\nu\kappa}\theta(t)|_{t=t_0} = \rho^{\kappa}D^{\nu\kappa}\chi(\rho t)|_{t=t_0} = \rho^{\kappa}(\nu\kappa)!\Lambda_{\nu}(\kappa);$$
(10)

thus, we obtain

$$\Theta_{\nu}(\kappa) = \frac{\left[D^{\nu\kappa}\theta(t)\right]_{t=t_0}}{(\nu\kappa)!} = \rho^{\kappa}\Lambda_{\nu}(\kappa).$$
(11)

This proves the first part. By using the first part, we conclude that

$$D^{\nu\kappa}\theta(t)|_{t=t_0} = D^{\nu\kappa}(D^{m\nu}\chi(\rho t))|_{t=t_0} = D^{\nu(m+\kappa)}\chi(\rho t))|_{t=t_0}, \text{ and}$$
(12)

consequently, we receive

$$D^{\nu\kappa}\theta(t)\big|_{t=t_0} = (\nu(m+\kappa))!\rho^{m+\kappa}\Lambda_{\nu}(m+\kappa).$$
(13)

Hence, we arrive at the part two,

$$\Theta_{\nu}(\kappa) = \frac{(\nu(m+\kappa))!}{(\nu\kappa)!} \rho^{m+\kappa} \Lambda_{\nu}(m+\kappa).$$
(14)

Finally, to prove the last part,

$$D^{\nu\kappa}\theta(t)|_{t=t_0} = D^{\nu\kappa}(I^{\nu}\chi(\rho t))|_{t=t_0} = D^{\nu(\kappa-1)}\chi(\rho t))|_{t=t_0}$$
$$= \frac{(\nu(\kappa-1))!}{(\nu\kappa)!}\rho^{\kappa-1}\Lambda_{\nu}(\kappa-1)$$
$$= \Theta_{\nu}(\kappa).$$
(15)

This completes the proof.

Proposition 2.2. Consider that $\theta_{\nu}(\kappa)$, $\Lambda_{\nu}(\kappa)$ and $\Omega_{\nu}(\kappa)$ are the FDT of the functions $\theta(t)$, $\chi(t)$ and $\omega(t)$, respectively, then

- **1.** If $\theta(t) = \chi(t) \pm \omega(t)$, then $\Theta_{\chi}(\kappa) = \Lambda_{\chi}(\kappa) \pm \Omega_{\chi}(\kappa)$.
- **2.** If $\theta(t) = \delta \chi(t)$, then $\Theta_{\nu}(\kappa) = \delta \Lambda_{\nu}(\kappa)$.

Proof: By applying Eq. (7) in the first part, we obtain

$$\Theta_{\nu}(\kappa) = \frac{\left[D^{\nu\kappa}\theta(t)\right]_{t=t_{0}}}{(\nu\kappa)!} = \frac{\left[D^{\nu\kappa}(\chi(t)\pm\omega(t))\right]_{t=t_{0}}}{(\nu\kappa)!}$$
$$= \frac{\left[D^{\nu\kappa}\chi(t)\right]_{t=t_{0}}}{(\nu\kappa)!} \pm \frac{\left[D^{\nu\kappa}\omega(t)\right]_{t=t_{0}}}{(\nu\kappa)!}$$
$$= \Lambda_{\nu}(\kappa) \pm \Omega_{\nu}(\kappa).$$
(16)

Similarly, for the second part, we have

$$\Theta_{\nu}(\kappa) = \frac{\left[D^{\nu\kappa}\theta(t)\right]_{t=t_0}}{(\nu\kappa)!} = \frac{\left[D^{\nu\kappa}\delta\chi(t)\right]_{t=t_0}}{(\nu\kappa)!} = \delta\Lambda_{\nu}(\kappa).$$
(17)

The procedure of CCS as a contribution to the Information Technology (IT) section can be observed as a resource chain. The Information Technology (IT) section improves the cost of the external part of CCS and deliveries such improved services to the inner consuming parts. Therefore, there are correspondences and there are parts of differences between the traditional resource chain techniques [22] and the suggested one that introduced in this effort (see **Figure 1**). The proposed CCS model shows dual relegation under the incomes because the Information Technology (IT) section scripts the cost of the incoming cloud system. Nevertheless, the cost reports double relegation by posing Information Technology (IT) services at no control. Moreover, apart from the differing reasons of the numerous objects, information irregularity is an additional problem in resource chain coordination, which causes the bullwhip result.



Figure 1. The proposed method of FCCS.

The proposed FCCS model theoretically looks like a resource chain construction. However, there are significant parts of the changes. We contain the stability and convergence of the system to the optimal solution by using the FDT, which implies the authority to inflict the operating organizational arrangement. The optimizing owns benefit discretely, or the models optimize the cost. The ICCS model, in the recent studies in Information Technology (IT) services, does not meet the resource problems for physical properties such as stock-outs or excess record. Moreover, we take into account the quality of Information Technology (IT) services under the ICCS model. Value development under CCS is expected to formulate a stationary price component and a quality-related minimal cost component. This is the new ICCS model that fixed the cost organization and increased the variable labor cost. Under this seeking, higher internal Information Technology (IT) value stages need a suitably high level of value from the ICCS because the price of a considerable value development is gross.

3. Applications

In this section, we illustrate some prototype applications to explain the accuracy of the FCCS.

3.1. System 1

Assume the following cost function:

$$\Phi(t,\chi) = D^{\nu}\chi(t) - \chi(t)$$

= 0 (18)
Initial condition : $\chi(0) = 1, \chi'(0) = 1.$

In view of Proposition 2.1, we obtain the fractional differential transform type of Eq. (18) as follows:

$$\frac{(\nu(1+\kappa))!}{(\nu\kappa)!}\Lambda_{\nu}(1+\kappa) - \Lambda_{\nu}(\kappa) = 0$$
(19)
Initial condition : $\Lambda_{\nu}(0) = \Lambda_{\nu}(1) = 1$,

where Λ_{γ} is the fractional differential transform of $\chi(t)$, which is read in Eq. (7). By utilizing Eq. (19), we obtain the following system:

Measurement of the Communication Possibility of Service Requests for Multiservers in Parallel Connection... 29 http://dx.doi.org/10.5772/65285

$$\frac{(2\nu)!}{\nu!} \Lambda_{\nu}(2) - 1 = 0,
\frac{(3\nu)!}{(2\nu)!} \Lambda_{\nu}(3) - \Lambda_{\nu}(2) = 0,
\frac{(4\nu)!}{(3\nu)!} \Lambda_{\nu}(4) - \Lambda_{\nu}(3) = 0,
\frac{(5\nu)!}{(4\nu)!} \Lambda_{\nu}(5) - \Lambda_{\nu}(4) = 0.$$
(20)

Solving the above system, we obtain the following outcome:

$$\Lambda_{\nu}(t) \approx 1 + t + \frac{1}{\Gamma(2\nu+1)}t^{2} + \frac{1}{\Gamma(3\nu+1)}t^{3} + \frac{1}{\Gamma(4\nu+1)}t^{4} + \frac{1}{\Gamma(5\nu+1)}t^{5}.$$
(21)

In general, we attain to the solution

$$\Lambda_{\nu}(t) \approx E_{\nu,1}(t), \tag{22}$$

where $E_{\nu,1}$ is the Mittag-Leffler function defined by the series:

$$E_{\nu,\mu}(t) = \sum_{\kappa=0}^{\infty} \frac{t^{\kappa}}{\Gamma(\nu\kappa + \mu)}.$$
(23)

3.2. System 2

Assume the following cost function:

$$\frac{(\nu(1+\kappa))!}{(\nu\kappa)!}\Lambda_{\nu}(1+\kappa) - \frac{1}{2}\Lambda_{\nu}(\kappa) - \frac{1}{2}\frac{(\nu(\kappa-1))!}{(\nu\kappa)!}\Lambda_{\nu}(\kappa-1) = 0$$
Initial condition : $\Lambda_{\nu}(0) = \Lambda_{\nu}(1) = \frac{1}{2}$,
(24)

In view of Proposition 2.1, we get the fractional differential transform of Eq. (24) as follows:

$$\frac{(\nu(1+\kappa))!}{(\nu\kappa)!}\Lambda_{\nu}(1+\kappa) - \frac{1}{2}\Lambda_{\nu}(\kappa) - \frac{1}{2}\frac{(\nu(\kappa-1))!}{(\nu\kappa)!}\Lambda_{\nu}(\kappa-1) = 0$$
Initial condition : $\Lambda_{\nu}(0) = \Lambda_{\nu}(1) = \frac{1}{2},$
(25)

where Λ_{γ} is the fractional differential transform of $\chi(t)$, which is given in Eq. (7). By utilizing Eq. (25), we obtain the following system:

$$\frac{(2\nu)!}{\nu!}\Lambda_{\nu}(2) - \frac{1}{2}\Lambda_{\nu}(1) - \frac{1}{2}\frac{1}{\nu!}\Lambda_{\nu}(0) = 0,$$

$$\frac{(3\nu)!}{(2\nu)!}\Lambda_{\nu}(3) - \frac{1}{2}\Lambda_{\nu}(2) - \frac{1}{2}\frac{\nu!}{(2\nu)!}\Lambda_{\nu}(1) = 0,$$

$$\frac{(4\nu)!}{(3\nu)!}\Lambda_{\nu}(4) - \frac{1}{2}\Lambda_{\nu}(3) - \frac{1}{2}\frac{(2\nu)!}{(3\nu)!}\Lambda_{\nu}(2) = 0,$$

$$\frac{(5\nu)!}{(4\nu)!}\Lambda_{\nu}(5) - \frac{1}{2}\Lambda_{\nu}(4) - \frac{1}{2}\frac{(3\nu)!}{(4\nu)!}\Lambda_{\nu}(3) = 0.$$
(26)

Solving the above system, we obtain the following outcome for $v \in (0,1]$:

$$\Lambda_{\nu}(t) \approx \frac{1}{2} \left[1 + t + \frac{1}{\Gamma(2\nu+1)} t^2 + \frac{1}{\Gamma(3\nu+1)} t^3 + \frac{1}{\Gamma(4\nu+1)} t^4 + \frac{1}{\Gamma(5\nu+1)} t^5 \right].$$
(27)

In general, we attain the solution

$$\Lambda_{\nu}(t) \approx \frac{1}{2} E_{\nu,1}(t).$$
 (28)

3.3. System 3

Suppose the following cost function:

$$\Phi(t,\chi) = D^{2\nu}\chi(t) + D^{\nu}\chi(t) - \chi(t) + I^{\nu}\chi(t)$$

= 0 (29)
Initial condition : $\chi(0) = 1, \chi'(0) = -1.$

In view of Proposition 2.1, we have the fractional differential transform of Eq. (29) as follows:

Measurement of the Communication Possibility of Service Requests for Multiservers in Parallel Connection... 31 http://dx.doi.org/10.5772/65285

$$\frac{(\nu(2+\kappa))!}{(\nu\kappa)!}\Lambda_{\nu}(2+\kappa) + \frac{(\nu(1+\kappa))!}{(\nu\kappa)!}\Lambda_{\nu}(1+\kappa) - \Lambda_{\nu}(\kappa) + \frac{(\nu(\kappa-1))!}{(\nu\kappa)!}\Lambda_{\nu}(\kappa-1) = 0$$
Initial condition : $\Lambda_{\nu}(0) = 1, \Lambda_{\nu}(1) = -1,$
(30)

where Λ_{ν} is the fractional differential transform of $\chi(t)$, which is given in Eq. (7). By utilizing Eq. (25), we obtain the following system:

$$\frac{(3\nu)!}{\nu!}\Lambda_{\nu}(3) + \frac{(2\nu)!}{\nu!}\Lambda_{\nu}(2) - \Lambda_{\nu}(1) + \frac{1}{\nu!}\Lambda_{\nu}(0) = 0,$$

$$\frac{(4\nu)!}{(2\nu)!}\Lambda_{\nu}(4) + \frac{(3\nu)!}{(2\nu)!}\Lambda_{\nu}(3) - \Lambda_{\nu}(2) + \frac{\nu!}{(2\nu)!}\Lambda_{\nu}(1) = 0,$$

$$\frac{(5\nu)!}{(3\nu)!}\Lambda_{\nu}(5) + \frac{(4\nu)!}{(3\nu)!}\Lambda_{\nu}(4) - \Lambda_{\nu}(3) + \frac{(2\nu)!}{(3\nu)!}\Lambda_{\nu}(2) = 0,$$

$$\frac{(6\nu)!}{(4\nu)!}\Lambda_{\nu}(6) + \frac{(5\nu)!}{(4\nu)!}\Lambda_{\nu}(5) - \Lambda_{\nu}(4) + \frac{(3\nu)!}{(4\nu)!}\Lambda_{\nu}(3) = 0.$$
(31)

Solving the above system, we get the following outcome for $v \in (0,1)$:

$$\Lambda_{\nu}(t) \approx \left[1 - t + \frac{1}{\Gamma(2\nu+1)}t^2 - \frac{1}{\Gamma(3\nu+1)}t^3 + \frac{1}{\Gamma(4\nu+1)}t^4 - \frac{1}{\Gamma(5\nu+1)}t^5\right].$$
(32)

In general, we attain the solution

$$\Lambda_{\nu}(t) \approx E_{\nu,1}(-t). \tag{33}$$

From above examples, we conclude that the solution of a fractional system, taking the formal (3), has the following general formula:

$$\chi(t) = (\chi(0))E_{\nu}(\lambda t^{\nu}). \tag{34}$$

4. Results and discussion

For the extra model improvement, it is essential to form some model considerations that come across the demand of applicability and to study major information and price styles.

- Every worker has a computer linked to the internet. Therefore, the price of acquiring and the price of the network (switches, routers, and network security) are not recognized to the CCS.
- The effort of the analysis deceits in the price evaluation of different CCS and service structures. As a consequence, the price for servers is reflected in the mathematical model.
- The price (cost) construction and empathy of cost styles have been initially formed in the source of the CCS.

The cost function is suggested and studied by many researchers in various methods [23–25]. For example, the authors used the form

$$Total Cost = Expenditure of time = Total time* The cost per hour.$$
(35)

Most of these methods do not request the quality of the CCS. Our technique requested the quality and integrated it. **Table 1** shows the experimental results, the proposed method, and comparison with the usual method for calculating the cost of CCS. It is well known that E_{ν} satisfies the following asymptotic behavior (see [26]; Theorem 1):

$$E_{\nu}(t) \sim \frac{1}{\nu} e^{t^{1/\nu}}, \quad \nu \neq 0.$$
 (36)

Hence, we have the solution

$$\chi(t) \approx \frac{\lambda}{\nu} e^{t(\Delta Q)^{1/\nu}},\tag{37}$$

where ΔQ is the difference in the quality of the CCS, which is suggested in [0, 1] and $\lambda = \chi(0)$ is the initial cost of the cloud. In this effort, based on the above discussion, we impose the following formal of the total cost

Total Cost = Total time* The cost per hour *
$$\chi(t)$$
, (38)

where $\chi(t)$ is given in Eq. (37).

Obviously, the quality of the cloud plays an important role in the cost. Increasing the quality leads to higher cost. At the level $\Delta Q = v = 0.1$ (high quality, low fractional value), we see that the cost is minimized to the quarter, while the level $\Delta Q = v = 0.5$ (the quality is less and higher fractional power), we obtain no much minimization for the cost. Moreover, in the case $\Delta Q = 0.1$, = v = 0.75, we have a minimum up to the half. Hence, we conclude that the fractional

Time	[23-25]	FCCS	[Eq. (38)]			
		$\Delta Q = \nu = 0.1$	$\Delta Q = \nu = 0.25$	$\Delta Q = \nu = 0.4$	$\Delta Q = \nu = 0.5$	$\Delta Q = 0.1, \nu = 0.75$
10	\$112.0	\$246.6	\$560	\$896	\$1131.1	\$896
12	\$134.4	\$295.68	\$672	\$1075.2	\$1357.44	\$1075.2
14	\$156.8	\$344.96	\$784	\$1254.4	\$1583.66	\$1254.4
16	\$179.2	\$394.24	\$896	\$1433.6	\$1809.92	\$1433.6
18	\$201.6	\$433.5	\$1008	\$1612.8	\$2036.16	\$1612.8
20	\$224.0	\$492.8	\$1120	\$1792	\$2262.4	\$1792

power of the cloud system also plays an important role to minimize the cost. The fractional power of the system may be realized as the rate of the gain of the FCCS.

Table 1. The cost of FCCS with initial cost = \$112.

5. Conclusion

The cloud computing system moves the manner activities (business, economy, and industries) and deals with their plans. The cloud computing system uses a new method to improve the economy, industries, and businesses, that is, everything is supposed as a service, be a service you demand or a service you deliver. We employed the concept of fractional formal calculus to impose a mathematical model of the cloud computing system. This system is obligated to various factors such as the controller, the utility of the cloud, structure for the Information Technology (IT) unit, and the cost taking into account the gain of the producers. Although the model has created simplifying conditions, it yielded a selection of outcomes that enhance the literature and have management suggestions. Obviously, any concept that is essential for a dynamic system is restricted in its ability to internment the complexity of Information Technology (IT) control. The proposed system may be viewed as a hybrid system, integrated system, as well as a perturbed system. It included various types of well-known systems (see [22, 27]). Also, the suggested system is addressed with huge number of data (time and space). Our goal was to introduce a method to minimize the cost function of the cloud. The anticipated framework is comprehensive in that the coverage contains the conceptual model, quantitative computation, logical representation, straddling both the theory and the applications of the problem. The results in **Table 1** show the performance of the method, by using fractional formal calculus. The obtainable method raises the quality of the system as well as minimizes the cost in the cloud computing system.

6. Competing interests

The author declares that there are no competing interests.

7. Author's contributions

There is no conflict of interests regarding the publication of this article.

Acknowledgements

The author would like to thank the referees for giving useful suggestions for improving the work.

Author details

Rabha W. Ibrahim

Address all correspondence to: rabhaibrahim@yahoo.com

Kampung Kerinch, Wilayah Persekutuan, Kuala Lumpur, Malaysia

References

- [1] Moschakis I.A., Karatza H.D. Evaluation of gang scheduling performance and cost in a cloud computing system. The Journal of Supercomputing 59.2 (2012): 975–992.
- [2] Chang B.R., Tsai H.F., Chen C.M. Evaluation of virtual machine performance and virtualized consolidation ratio in cloud computing system. Journal of Information Hiding and Multimedia Signal Processing 4.3 (2013): 192–200.
- [3] Ali J., Yupapin P.P. Micro-cloud computing system by human quantum computer. Journal of Biosensors and Bioelectronics 5.1 (2014): e127.
- [4] Da Silva D.M., et al. Resource management using reliable and efficient delivery of application performance information in a cloud computing system. U.S. Patent No. 8,935,704. 2015.
- [5] Chen T., Wang Y.C. A nonlinearly normalized back propagation network and cloud computing approach for determining cycle time allowance during wafer fabrication. Robotics and Computer-Integrated Manufacturing In Press (2016).
- [6] Thackston R., Fortenberry R.C. The performance of low-cost commercial cloud computing as an alternative in computational chemistry. Journal of Computational Chemistry 36.12 (2015): 926–933.

- [7] Sookhak M., et al. Dynamic remote data auditing for securing big data storage in cloud computing. Information Sciences (2015): 9 1–16.
- [8] Sabi H.M., et al. Conceptualizing a model for adoption of cloud computing in education. International Journal of Information Management 36.2 (2016): 183–191.
- [9] Ibrahim R.W., Jalab H.A., and Gani A. Perturbation of fractional multi-agent systems in cloud entropy computing. Entropy 18.1 (2016): 31.
- [10] Jeong D.H. An integrated framework for managing sensor data uncertainty using cloud computing. Information Systems 38 (2013): 1252–1268.
- [11] Cao N., et al. Privacy-preserving multi-keyword ranked search over encrypted cloud data. IEEE Transactions on Parallel and Distributed Systems 25.1 (2014): 222–233.
- [12] Buyya R., et al. Big data analytics-enhanced cloud computing: challenges, architectural elements, and future directions. arXiv:1510.06486 (2015):21 (pp. 75–84).
- [13] Hashem I.A.T., et al. The rise of big data on cloud computing: review and open research issues. Information Systems 47 (2015): 98–115.
- [14] Khulal U., et al. Nondestructive quantifying total volatile basic nitrogen (TVB-N) content in chicken using hyperspectral imaging (HSI) technique combined with different data dimension reduction algorithms. Food Chemistry 197 (2016): 1191–1199.
- [15] Podlubny I. Fractional differential equations, Academic Press, San Diego, Boston, NewYork, London, Tokyo, Toronto, 1999.
- [16] Kilbas A.A., Srivastava H.M., Trujillo J.J. Theory and applications of fractional differential equations, vol. 204. Elsevier Science Limited, Amsterdam 2006.
- [17] Ibrahim R.W., Jalab H.A., Gani A. Cloud entropy management system involving a fractional power. Entropy 18 (2016): 1–11.
- [18] Ibrahim R.W. Fractional complex transforms for fractional differential equations. Advances in Difference Equations 2012.1 (2012): 1–12.
- [19] Ibrahim R.W. Numerical solution for complex systems of fractional order. Journal of Applied Mathematics 2012 (2012) 1–12.
- [20] Ibrahim R.W. Complex transforms for systems of fractional differential equations. Abstract and Applied Analysis, vol. 2012. Hindawi Publishing Corporation, 2012 pp. 1–16.
- [21] Jumarie G. Fractional differential calculus for non-differentiable functions. Mechanics, Geometry, Stochastics, Information Theory, Lambert Academic Publishing, Saarbrucken, 2013.
- [22] Saber R., et al., Consensus and cooperation in networked multi-agent systems, Proceedings of the IEEE, 95.1 (2007): 215–233.

- [23] Li X., Li Y., Liu T., Qiu J., Wang F. The method and tool of cost analysis for cloud computing. 2009 IEEE International Conference on Cloud Computing (2009): 93–100.
- [24] Becker J., Beverungen D., Matzner M., Mller O. Total costs of service life: the need of decision support in selecting, comparing and orchestrating services. First International Conference on Exploring Services Sciences, Berlin (2010).
- [25] Benedikt M., Marc W., Frank T., Costing of cloud computing services: a total cost of ownership approach, 45th Hawaii International Conference on System Sciences, Hawaii (2012).
- [26] Gerhold S. Asymptotics for a variant of the Mittag-Leffler function. Integral Transforms and Special Functions, 23.6 (2012): 397–403.
- [27] Salih Y.K., et al., A user-centric game selection model based on user preferences for the selection of the best heterogeneous wireless network. Annals of Telecommunications 70 (2015): 239–248.

System Control for Sustainability: Application to Building Design

Kazutoshi Fujihira

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/65875

Abstract

In order to meet environmental, social and economic sustainability objectives, an immediate transformation needs to occur in housing design. We have been investigating a methodology for sustainable home design by applying control science because it can be applied to all of goal-oriented tasks. Our first study has shown the basic control system for sustainability. After that, regarding home as the complex of material and spatial elements, we have identified each element's variables and their desired values that lead to sustainability. Utilizing these schemes, we have composed the "control system for promoting sustainable home design" in which the "sustainable design guidelines" and "sustainability checklist" are incorporated. Following this control system, we have actually designed a home and constructed it. The results of this study indicate that if system users closely follow the methodology, they can design comprehensively sustainable and sufficiently energy-saving homes. Furthermore, the studies imply that this methodology is superior in user-friendliness as well as adaptable to regional differences and changes over time.

Keywords: control system, sustainable housing, climate change, adaptation, ageing population, material elements, spatial elements

1. Introduction

Buildings including homes have been causing various environmental problems such as climate change, environmental destruction, pollution, and depletion of natural resources. According to the IPCC, in 2010, buildings accounted for 32% of total global final energy use (24% for residential and 8% for commercial) and 19% of all global greenhouse gas emissions [1]. In the twenty-first century, continued emission of greenhouse gases will cause further warming, increasing the likelihood of severe, pervasive, and irreversible impacts on people and ecosystems [2].



© 2016 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. Reducing climate change risks would require both "mitigation" and "adaptation" measures [2, 3], typically shown in **Table 1**. The chapter of "Climate-Resilient Pathways: Adaptation, Mitigation and Sustainable Development" in *Climate Change 2014* explains, "It takes sustainable development as the ultimate goal, and considers mitigation as a way to keep climate change moderate rather than extreme. Adaptation is considered a response strategy to anticipate and cope with impacts that cannot be (or are not) avoided under different scenarios of climate change" [4].

Mitigation measures		Adaptation measures	Adaptation measures			
•	Energy conservation	Measures against inland and coastal floods				
•	Harnessing renewable energy	Measures for dealing with water shortages				
•	Conserving timber resources	Heatstroke warning systems				
•	Carbon dioxide capture and storage	Building resilient infrastructure				

Table 1. Typical examples of mitigation and adaptation measures.

While growing human activities are aggravating climate change and seriously harming environmental sustainability, "ageing population" is becoming a fierce challenge to social and economic sustainability. According to the book titled *Agequake*, in the twenty-first century there will be more people older than younger ones. The "agequake" will turn the age pyramids upside down, first in Japan and Europe, but ultimately throughout the world [5]. Demographic pressures will continue to mount and add to concerns about fiscal sustainability [6, 7]. Ed Harding – International Longevity Centre UK clearly states, "Suitable housing is central to the challenge of population ageing. Appropriate housing offers the potential to reduce expenditure on public services and promote older people's independence and wellbeing" [8].

It is obvious that an immediate transformation needs to occur in housing design, in order to meet sustainability objectives. Achieving sustainable development or sustainability is the ultimate goal-oriented task of humankind. It is a rational and reliable approach to apply control science to this challenge. Control science can be applied to all of goal-oriented tasks; it has already produced remarkable results in various fields, including engineering [9].

Based on the above recognition, we have been investigating a methodology for sustainable housing design by applying control science. Our first study has demonstrated the basic control system for sustainability. After that, we have systematically determined the relationships between the elements of home and sustainability. Utilizing these schemes, we have shown the control system for promoting sustainable home design and started putting it into practice. This paper reviews these studies and discusses the characteristics of the methodology and future prospects of our research.

2. Basic control system for sustainability

In the basic control system for sustainability (**Figure 1**), "controlled objects" are human activities which need to be controlled [10, 11]. "Disturbances" are harmful influences on controlled objects, which are caused by environmental, social, or economic problems. Examples of the disturbances are damaging influences caused by environmental pollution, floods or landslides resulting from environmental destruction, and various impacts caused by climate change. Moreover, "adaptation" has been added as the route from "disturbances" to "sustainability," based on the recent IPCC's recognition. "Controlled variables" are the variables that are related to the human activities and need to be controlled for basically solving or preventing the problems or adapting to disturbances.



Figure 1. Basic control system for sustainability.



Figure 2. Model of sustainability.

"Desired values" are derived from the purpose of control, that is to say, sustainability. The model of sustainability (**Figure 2**) demonstrates that sustainability requires both "fundamental stability" and "internal stability," in order to accomplish the long-term

well-being of humankind or ultimate goal, within the finite global environment and natural resources, or absolute limitations [10, 11]. Fundamental stability means environmental stability and a stable supply of necessary goods; the conditions for fundamental stability are environmental preservation and the sustainable use of natural resources. On the other hand, internal stability means social and economic stability; the conditions for internal stability are health, safety, mutual help, and self-realization, which are essential for well-being of humans.

The control objective is to adjust the controlled variables to their desired values. Furthermore, the control system requires designing and implementing "control measures," or measures for achieving the control objective.

3. Methodology for sustainable housing design

3.1. Two-step preparatory work

Our previous study has provided the two-step preparatory work for sustainable home design. The two steps are (1) "determining the relationships between the standard home and sustainability" and (2) "sustainability checkup on a home as an object" [12, 13]. The following sections show the essence of these two steps.

3.1.1. Determining the relationships between the standard home and sustainability

In the first step, system designers select important elements of the standard home and determine the relationships between such elements and sustainability.

When selecting important elements, we have analyzed two main factors: "material" and "space" [12]. "Material" regards the home as the aggregate of material elements such as framework, exterior, interior, and piping; "space" considers it as the aggregate of spatial elements, such as rooms and areas. Based on these two factors, we have selected important elements, from which extracts are shown in the central column of **Table 2**. "Material elements" are from "framework" to "equipment for harnessing natural energy;" "spatial elements" are from "specified bedroom" to "garden area."

After that, we have determined the relationships between these elements and sustainability. The left side of **Table 2** shows the relationships between the elements and fundamental stability; the right side demonstrates the relationships between the elements and internal stability. Considering the relationships between each element and the stability conditions, we have identified variables that indicate the degree of stability. Moreover, we have set the desired values of these variables that can meet relevant stability conditions.

Choosing only one element from **Table 2**, that is, "framework," I concretely describe the relationships. Considering the relationship between "framework" and "sustainable use of natural resources," a condition for fundamental stability, we have identified "durability" and "materials" as variables. The desired value of "durability" is set at "Grade 3" in the deterioration resistance grades (building frames, etc.) of JHPIS, that is, the Japan housing performance

Relationships with fundamental stability		T1 (Relationships with internal stability				
Condition	Desired value	Variable	Element	Variable	Desired value	Condition	
Sustainable use of resources	JHPIS 3.1: Grade 3	Durability		Resistance to	JHPIS 1.1:	Safety	
Sustainable use of resources	CASBEE LR _H 2 1.1: Level 4 or over	Materials	Framework	earthquakes	Grade 2 or over		
Sustainable use of resources	CASBEE LR _H 2 1.3: Level 4 or over	Durability	Exterior (outer	Fire resistance (outer wall)	JHPIS 2.6: Grade 3 or over	Safety	
Sustainable use of resources	CASBEE Q _H 2 1.2 & 3: Level 4 or over	Materials	wall, roof, etc.)	Shape & color	Consideration for the landscape	Health	
Enviro-preserve Sustainable use of resources	JHPIS 5.1: Grade 4	Thermal insulation performance	Windows &	Sound insulation performance	JHPIS 8.4: Grade 2 or over	Health	
Enviro-preserve Sustainable use of resources	CASBEE Q _H 1 1.1.2: Level 4 or over	Sunlight adjustment capability	doors	Protection of glass against impacts	With shutters	Safety	
Sustainable use of resources	CASBEE LR _H 2 1.4: Level 4 or over	Materials	Interior	Measures against formaldehyde	JHPIS 6.1: Grade 3	Health	
Enviro-preserve Sustainable use of resources	LED	Type of light	Lighting fixtures				
Enviro-preserve Sustainable use of resources	CASBEE LR _H 1 3.2: Level 4 or over	Rainwater equipment	Equipment for rainwater use	Rainwater CASBEE LR _H 1 3.2 equipment Level 4 or over		Health Safety (in crisis)	
Enviro-preserve Sustainable use of resources	100% or more of the total energy usage	Harnessed natural energy	Equipment for harnessing natural energy	Harnessed natural energy	100% or more of the total energy usage	Health Safety (in crisis)	
			Specified bedroom	Routes to toilet & bath area, dining room, kitchen, and entrance	Accessible without steps	Health safety	
Enviro-preserve Sustainable use of resources	Placing them closer	Areas in the home	Areas relating to water use & hot water supply				
			Doorways	Differences in level	No differences	Health safety	
				Width	75 cm or more		
Enviro-preserve	40% or more	Ratio to the exterior area	Garden area				

Note: (1) JHPIS means the Japan housing performance indication standards (for new homes).

(2) CASBEE means CASBEE for detached houses (for new construction)-technical manual 2010 edition.

Table 2. Relationships between the standard home and sustainability [Essence].

Relationships with fundamental stability					Relationships with internal stability			
Desired value	Assess.	Measured or estimated value	Variable	Element	Variable	Measured or estimated value	Assess.	Desired value
JHPIS 3.1: Grade 3	No	JHPIS 3.1: Grade 1	Durability		Resistance to earthquakes	JHPIS 1.1 & 1.2: Grade 1	No	JHPIS 1.1: Grade 2 or over
CASBEE LR _H 2 1.1: Level 4 or over	ОК	CASBEE LR _H 2 1.1: Level 4	Materials	Framework				
CASBEE Q _H 2 1.2 & 3: Level 4 or over	No	CASBEE Q _H 2 1.2 & 3: Level 2	Durability	Exterior (Outer	Fire resistance (Outer wall)	JHPIS 2.6: Grade 3	ОК	JHPIS 2.6: Grade 3 or over
CASBEE LR _H 2 1.3: Level 4 or over	No	CASBEE LR _H 2 1.3: Level 3	Materials	wall, roof, etc.)	Shape & color	Consideration for the landscape	ОК	Consideration for the landscape
JHPIS 5.1: Grade 4	No	JHPIS 5.1: Grade 1	Thermal insulation performance	Windows &	Sound insulation performance	JHPIS 8.4: Grade 1	No	JHPIS 8.4: Grade 2 or over
CASBEE Q _H 1 1.1.2: Level 4 or over	No	CASBEE Q _H 1 1.1.2: Level 3	Sunlight adjustment capability	doors	Protection of glass against impacts	With shutters	ОК	With shutters
CASBEE LR _H 2 1.4: Level 4 or over	ОК	CASBEE LR _H 2 1.4: Level 4	Materials	Interior	Measures against formaldehyde	JHPIS 6.1: Grade 3	ОК	JHPIS 6.1: Grade 3
LED	No	Fluorescent	Type of light	Lighting fixtures				
CASBEE LR _H 1 3.2: Level 4 or over	No	No equipment	Rainwater equipment	Equipment for rainwater use	Rainwater equipment	No equipment	No	CASBEE LR _H 1 3.2: Level 4 or over
100% or more of the total energy	No	0 (zero)	Harnessed natural energy	Equipment harnessing natural energy	Harnessed natural energy	0 (zero)	No	100% or more of the total energy
				Specified bedroom	Routes to toilet & bath area, dining, kitchen, and entrance	With steps	No	Accessible without steps
Placing them closer	ОК	Placing them closer	Areas in the home	Areas water use & hot-water supply				
				Doorways	Differences in level	With differences	No	No differences
					Width	60-70 cm	No	75 cm or more
40% or more	ОК	45%	Ratio to the exterior area	Garden area				

Note: (1) JHPIS means the Japan housing performance indication standards (for new homes).

(2) CASBEE means CASBEE for detached houses (for new construction)-technical manual 2010 edition.

(3) When this checklist is used for the inspection or evaluation of existing homes, JHPIS (for existing homes) and CASBEE for detached houses (for existing building)—technical manual 2011 edition need to be referred to, instead of the "for new homes" version and "for new construction" version, respectively.

Table 3. An example of sustainability checkup on a home as an object [Essence].

indication standards (for new homes) [14]. Meanwhile, the desired value of "materials" is set at "Level 4 or over" in the assessment levels of the "use of materials effective for resource saving and waste prevention" of CASBEE, that is, CASBEE for detached houses (for new construction)—technical manual 2010 edition [15].

On the other hand, considering the relationship with "safety," a condition for internal stability, we have selected "resistance to earthquakes" as a variable and set its desired value at "Grade 2 or over" in the "seismic resistance grades (prevention of collapse of building structures)" of JHPIS [14].

In addition, **Table 2** is the first updated version, which has been revised due to several reasons. First, we have revised the table so that following it leads to obtain long-life quality housing (*Choki Yuryo Jutaku*) certification. The long-life quality housing certification started in 2009 and has rapidly spread in Japan because of various incentives, such as tax reduction [16]. The second reason is the addition of adaptation measures against impacts caused by climate change. Other reasons are rapid spread of new technology, namely LED light, and minor changes in expressions and desired values.

3.1.2. Sustainability checkup on a home as an object

In the second step, "sustainability checkup on a home as an object" is conducted. To be concrete, first, the variables of a home as an object are measured or estimated. Next, the measured or estimated values are compared with the desired values and the comparison results are assessed. **Table 3** demonstrates the essence of "sustainability checkup," which includes an example of the results.

In this case, the comparison results have simply been assessed, whether the variable reaches the desired value or not, that is, "OK" or "No." The variables that have been assessed as "No" need to be identified as "controlled variables."

3.2. Control system for promoting sustainable home design

Based on the basic control system and the two-step preparatory work, we have composed the control system for promoting sustainable home design.

First, as shown in **Figure 3**, we have derived two functions from the two-step preparatory work, that is, the "sustainable design guidelines" from step 1 and the "sustainability check-list" from step 2, respectively.



Figure 3. Two functions derived from the two-step preparatory work.



Figure 4. Control system for promoting sustainable home design.

Next, we have composed the control system for promoting sustainable home design in which these two functions are incorporated. **Figure 4** demonstrates the block diagram of the control system. In this control system, "controlled objects" include both "new homes" and "existing homes." The following describes how to use the two functions in the process of sustainable home design, on the order of "new homes" and "existing homes."

3.2.1. New homes

In the case of new homes, first, information on the desired values reaches "people involved in design" through the "sustainable design guidelines" [17]. "People involved in design" include homeowners, architects, designers, and homebuilders. Such people make a design so that the variables of home's elements can attain their desired values as close as possible. At important steps in the design process, they check the drawings and specifications by referring to the "sustainability checklist." After the construction is finished, the new home can be also evaluated against the "sustainability checklist."

3.2.2. Existing homes

When existing homes are the objects, the design process starts with "inspection" on the home as an object. The "people involved in design" measure or estimate each element's variables of that home by using the "sustainability checklist." Next, they compare the measured or estimated variables with their desired values and assess the comparison results.

Table 3 in the last section is equivalent to an example of such inspection results. In addition, when inspecting an existing home and measuring or estimating variables by referring to the JHPIS or CASBEE for detached houses, "people involved" use the "for existing homes" version or "for existing building" version, instead of the "for new homes" version or "new construction" version, respectively. The "JHPIS (for existing homes)" and "CASBEE

for detached houses (for existing building)" are almost the same as their new home versions; however, these existing home versions include suitable assessment criteria for existing homes [14, 18].

After the inspection, the "people involved" usually make a design for improvement so that controlled variables meet their desired values as close as possible. When improvement is assumed to be technically difficult or costly, reconstruction can be chosen. Similar to the cases of new homes, they check the drawings and specifications for improvement or reconstruction against the "sustainability checklist." Moreover, sustainability of actually improved or reconstructed homes can be evaluated against the "checklist."

4. Case study

Following the above methodology, we designed a new home and constructed it. After this home began to be used, we objectively assessed its environmental performances.

4.1. Design process

In this case, the homeowner, who is the author of this paper, made the basic design by himself. After that, he made a contract with a housing manufacturer whose skills seemed sufficient to achieve the desired values as shown in **Table 2**.

In the basic design stage, the homeowner used the "sustainable design guidelines" and "sustainability checklist," in addition to following related laws and regulations. First, when making the site plan and floor plans, he considered spatial elements, such as "specified bedroom," "areas relating to water use and hot water supply," and "garden area." In addition, considering natural ventilation and day lighting, he planned "position and area of windows," which is omitted from the tables due to space limitations. Moreover, he examined how to install "equipment for harnessing natural energy" and "equipment for rainwater use" so as to satisfy the desired values of these material elements' variables.

At the beginning of the detailed design stage, the homeowner requested the architects of the housing manufacturer to refer to the "guidelines" and "checklist." They readily accepted them because the material and spatial elements are equivalent to real parts of the home. Next, we determined the site plan, floor plans, elevation, and fundamental specifications. After that, the architects designed the home's elements such as framework, exterior, windows and doors, interior, and lighting fixtures, so that as much as possible, the elements' variables meet their desired values.

After the design stage, we obtained long-life quality housing certification as well as building confirmation. Subsequently, the home, a wooden structure with two stories, was constructed in a residential area of suburban Tokyo.

4.2. Results

This section shows the results of the design and construction, focusing on main elements' variables.



Figure 5. Site and floor plans.

First, **Figure 5** demonstrates the site plan and floor plans in which three spatial elements and one material element are referred to:

• Specified bedroom

A "specified bedroom" is a bedroom which is used or expected to be used by elderly or wheelchair users. The owner couple was in their late fifties and not handicapped when they moved in. However, in order to prepare for their future, both their bedroom and other essential areas for their daily living, all have been placed on the same first floor. This arrangement enables them to have access to such areas without any steps.

• Areas relating to water use and hot water supply

"Areas relating to water use and hot water supply," that is, kitchen, bath, washing room, toilet, and a place to put the water heater, have been placed closer. This planning reduces the total length of water piping and drainage piping. Moreover, this consideration helps to reduce heat loss from hot water piping.

• Garden area

A "garden area" is an area with plants such as trees, shrubs, herbs, grasses and vegetables. The ratio of this home's garden area to the total exterior area is 67%, which surpasses its desired value, 40%. A larger garden area is favorable for environmental preservation, including mitigation of heat island phenomenon, and a higher level of biodiversity.

• Framework

This home has been designed so that the "resistance to earthquakes" of the framework meets "Grade 3" in the seismic resistance grades of JHPIS, which surpasses the required level, Grade 2.

The "durability" of the framework satisfies the desired value, or "Grade 3" in the deterioration resistance grades of JHPIS. Grade 3, the highest grade, requires measures to extend the period up to when large-scale renovation becomes necessary after three generations, or about 75–90 years, under normally assumed conditions and maintenance [14].

As for "materials," more than half of the wood is domestically produced timber, legality and sustainability of which are authorized. This consideration has sufficiently satisfied the desired value.

Figure 6 shows external appearance, including a panoramic view, with the descriptions of four material elements: exterior, windows and doors, equipment for harnessing natural energy, and equipment for rainwater use.

• Exterior

The "fire resistance" capacity of the outer walls fulfills the desired value, namely "Grade 3" in the fire resistance grades of JHPIS [14]. Meanwhile, the "shape" is simple but accented by

Equipment for harnessing natural energy:

Harnessed natural energy –
 516% of the total energy usage > 100%

Equipment for rainwater use:

- Rainwater equipment -
- CASBEE LR_H1 3.2: Level 4 [OK]



Exterior (outer wall, roof, etc.): • Fire resistance (outer wall) – JHPIS 2-6: Grade 3 [OK]

 Shape & color – Consideration for the landscape [OK] Windows and doors: •Thermal insulation performance – JHPIS 5-1: Grade 4 [OK] •Sunlight adjustment capability – CASBEE Q_H1 1.1.2: Level 5 > Level 4 •Protection of glass against impacts – With shutters (large windows) [OK]

Figure 6. External appearance.

the balcony; the "color" of the outer walls, light beige, is coordinated with the colors of other parts, such as the windows and doors, entrance porch, and rainwater pipes. We are expecting that these considerations bring harmony and stability to the landscape.

• Windows and doors

Higher "thermal insulation performance" and "sunlight adjustment capability" of the openings contribute to reducing energy for heating and air conditioning as well as the occupants' health and comfort [15, 19, 20]. However, both of these variables appear on only the left side of **Tables 2** and **3** as the space is limited. The "thermal insulation performance" of this house's openings meets the desired value, or the highest grade of the relevant item of JHPIS. Meanwhile, the "sunlight adjustment capability" reaches "Level 5" in the assessment levels of the CASBEE's relevant item. "Level 5" means that the sunlight penetration ratio of the subject windows can be reduced to 0.3 or less in the summer and 0.6 or over in the winter [15].

We have installed "shutters" on the large windows facing south, in order to "protect the glass against impacts" such as fire, hurricane, and flying objects. This installment is also expected to decrease the risk of being damaged by tornados or huge typhoons, which can be caused by serious climate change.

• Equipment for harnessing natural energy

The surface of the single-pitch roof is almost entirely covered with solar panels. The total generating capacity of the 49 solar panels amounts to 11.4 kW. In the first year, after moving-in, this solar system produced the electricity of 15,911 kWh. In the same period, the energy used in this home has been 3085 kWh. Accordingly, the energy self-sufficiency reached an amazing 516%. This value substantially exceeded the desired value, 100%.

In addition, equipment for harnessing natural energy secures alternative energy sources and increases resilience or passive survivability in crisis. Therefore, it is recognized as an "adaptation" measure as well as "mitigation" [3, 21].

• Equipment for rainwater use

"Equipment for rainwater use," in this case, a 200-1 rainwater tank, meets the desired value. The rainwater in this tank is used for watering the garden, which can reduce the quantity of



Interior: • Measures against formaldehyde – JHPIS 6.1: Grade 3 [OK] (Washing room and bath)

Doorways:
 Differences in level – No differences [OK]
 Width (except bath) – 75cm [OK]

Figure 7. Interior, lighting fixtures and doorways.

water supply. At the same time, storing rainwater is also considered one of "adaptation" measures because it leads to securing emergency water supply [3, 21].

Figure 7 illustrates two material elements, namely "interior" and "lighting fixtures," and one spatial element, namely "doorways."

Interior

Interior, which includes floors, walls and ceilings, requires "measures against formaldehyde" as an important variable. Formaldehyde is a harmful pollutant; therefore, its desired value is set at the highest level of the countermeasures against formaldehyde in JHPIS. All of the interior finish and base materials used in this house are certified materials that meet the desired value.

• Lighting fixtures

All of the "lighting fixtures" of this house use ""LED" lights, which meets the desired value.

• Doorways

We have taken barrier-free and universal design into maximum consideration. All of the "doorways" in this home have no "difference in level," and "width" of the doorways on the first floor except for the bath is 75 cm.

4.3. Performance evaluation

4.3.1. Sustainability performance

After the home started to be used, we had its sustainability evaluated by the "CASBEE for detached houses (for new construction, 2010 edition)." The evaluation results show that the "environmental quality (Q)" and "environmental load (L)" have been "88" and "13" respectively. Consequently, as demonstrated in **Figure 8(A)**, the score of Building Environmental



Figure 8. Sustainability performance and energy-saving performance of this home.

Efficiency (BEE) has reached 6.7, which has rated and certified the home with the highest "S," or "five stars" [22].

4.3.2. Energy-saving performance.

We have analyzed the energy-saving performance of this home, by comparing its annual energy usage with that of the average home. The actual total energy usage in this home in a year was 3085 kWh. On the other hand, the average annual energy usage of the same two-person-household detached houses amounts to 41,325 MJ [23], which is equivalent to 11,479 kWh. The usage of 3085 kWh is equal to 27% of 11,479 kWh. Therefore, as shown in **Figure 8(B)**, this house has reduced energy consumption by over 70%, as compared to the average house under the same conditions.

5. Discussion

This chapter discusses the results of this study from the following perspectives: (1) effects of the methodology on achieving sustainable homes, (2) characteristics of the methodology, (3) future work, and (4) applicability of the methodology.

5.1. Effects of the methodology on achieving sustainable homes

As shown in the case study, we have designed a home, following the methodology. To be concrete, we have designed the home's parts, or elements, so that as much as possible the elements' variables meet their desired values. After the home started to be used, we have objectively assessed its environmental performances. First, CASBEE for detached houses, a comprehensive assessment system, has readily ranked the home in the highest "S," with an extremely high score of BEE. Meanwhile, the energy-saving performance assessment has shown that the total energy usage of this home is equal to about 27% of the average home under the same conditions. These assessment results indicate that if system users closely follow the methodology, they can achieve comprehensively sustainable and sufficiently energy-saving homes.

5.2. Characteristics of the methodology

The characteristics of the methodology includes (1) visualization of the processes for promoting sustainable home design, (2) user-friendliness, (3) adaptability to regional differences and changes over time. In addition, the third point is beneficial for the system designers, whereas the second point is convenient for the system users.

5.2.1. Visualization of the processes

Figure 4 has demonstrated the control system in which the "sustainable design guidelines" and "sustainability checklist" are incorporated. It concisely shows processes for promoting sustainable design on both new and existing homes. Moreover, this schematic diagram contains "disturbances" caused by problems, such as climate change. Accordingly, we consider

that this diagram itself explains the whole picture of the sustainable design processes with the guidelines and checklist and helps people concerned to recognize them.

5.2.2. User-friendliness

The "sustainable design guidelines" and "sustainability checklist" are simple tables. The material and spatial elements in these functions are equivalent to real parts of homes. Therefore, the system users smoothly design, check, evaluate, and inspect the home, by easily comparing the drawings or real home with the guidelines or checklist. In fact, the design process in the case study has supported the user-friendliness of the functions; the designers of the home easily accepted them and efficiently made drawings.

5.2.3. Adaptability to regional differences and changes over time

This methodology originally has a mechanism of easily adapting to regional differences. As shown in Section 3.1., system designers determine the elements' variables and their desired values, considering the relationships between the elements and sustainability conditions. This determination process has a mechanism of reflecting various regional features, such as natural, geographical, social, and cultural features. For example, a variable of the framework, "resistance to earthquakes," reflects a geological feature of earthquake-prone Japan. This mechanism also enables the system designers to easily vary the guidelines according to the characteristics of the region. For instance, if the region is in a snowy area or strong wind area, they can adjust the guidelines to the region, by adding "resistance to snowfall" or "resistance to wind" as a variable of the framework.

This characteristic also leads to the flexibility of the functions toward changes over time. In fact, in order to adjust to ageing population, we could smoothly take basic barrier-free design in the guidelines and checklist. To be concrete, we have added necessary spatial elements, such as "specified bedroom" and "doorways." Similarly, in order to adapt housing to climate change, we have taken adaptation measures in the functions. That is to say, we have added relevant variables including "protection of glass against impacts."

5.3. Future work

5.3.1. Further case studies

The case study has successfully demonstrated the effects of the methodology on achieving sustainable homes. However, in order to confirm the effects, we need to conduct more case studies, applying it to both new and existing homes. We expect that an increase in the number of case studies will also strengthen the credibility of the methodology and help it to be widely used.

5.3.2. Update of the guidelines and checklist

It is necessary to update the "guidelines" and "checklist," as the occasion arises. Probably, such revision will have to be made due to several reasons, for example, in response to the results of case studies, changes in the natural and social environment, developments in related sciences, and innovations in related technologies. Furthermore, through such processes, we are planning to examine how to revise them efficiently.

5.4. Applicability of the methodology

This section discusses the applicability of the methodology from two viewpoints: (1) application to other regions, (2) application possibility to other kinds of objects.

5.4.1. Application to other regions

As shown in Section 2, this methodology, specifically the "guidelines" and "checklist," has a characteristic of being adaptable to regional differences. Therefore, it will probably be easy for system designers in another region to adapt the functions for that region. To be concrete, the system designers can compile its regional version, by examining the elements and adapting the elements' variables and their desired values to the region's features.

5.4.2. Application possibility to other kinds of objects

Theoretically, the methodology can be applied to various kinds of objects, or human activities. That is to say, in the control system of **Figure 4**, "homes" in the block of "controlled objects" can be replaced with other kinds of objects. Possibility of such replacement depends on if the table of relationships, or the "guidelines," can be compiled or not.

It will not be difficult to apply it to other types of buildings besides the home, since the structure is similar to one another. It is also possible in theory to apply it to other sorts of infrastructure, such as roads and parks. Furthermore, we consider it possible to apply the methodology to more complex and large-scale objects, such as the city and town, for they are also regarded as the complex of material and spatial elements.

Acknowledgements

The author would like to thank Mr. Vance Carothers for his valuable advice and suggestions on the English expression of this paper.

Author details

Kazutoshi Fujihira

Address all correspondence to: fujihira@kankyogaku.com

Institute of Environmentology, Inagi, Tokyo, Japan

References

 Lucon O, et al. Buildings. In: Edenhofer O, et al, editors. Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press; 2014. pp. 671–738.

- [2] IPCC. Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Geneva, Switzerland: IPCC; 2014. 151 p.
- [3] IPCC. Climate Change 2007: Synthesis Report. Contribution of Working Groups I, II and III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Geneva, Switzerland: IPCC; 2007. 104 p.
- [4] Denton F, Wilbanks T, Abeysinghe A, Burton I, Gao Q, Lemos M, Masui T, O'Brien K, Warner K. Climate-resilient pathways: adaptation, mitigation, and sustainable development. In: Field C, et al, editors. Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press; 2014. pp. 1101–1131
- [5] Wallace P. Agequake: Riding the Demographic Rollercoaster Shaking Business, Finance and our World. Naperville, USA: Nicholas Brealey Publishing; 2001. 266 p.
- [6] European Union. Working Group on Ageing Populations and Sustainability [Internet]. Available from: http://europa.eu/epc/working_groups/ageing_en.htm [Accessed: 2016-06-10]
- [7] Oshio T, Oguro K. Fiscal sustainability under an aging population in Japan: a financial market perspective. Public Policy Review. 2013;9(4):735–750.
- [8] Ed Harding—International Longevity Centre UK. Sustainable Planning for Housing in an Ageing Population: A Guide for Regional-Level Strategies. International Longevity Centre UK; 2008. 29 p.
- [9] Osuka K, Adachi S. Approach to Systems Control (in Japanese). Tokyo, Japan: Corona Publishing; 1999. 177 p.
- [10] Fujihira K, Osuka K, Yoshioka T, Hayashi N. An educational methodology for sustainable development applying control theory and confirming its validity (in Japanese). Environmental Education. 2008;18(1):17–28.
- [11] Fujihira K, Osuka K. An educational methodology for sustainable development. In: ICCAS-SICE 2009 Final Program and Papers; 18–21 August, 2009; Fukuoka, Japan. Tokyo, Japan: SICE; 2009.
- [12] Fujihira K. An approach to sustainable homes by applying control science. Advanced Materials Research. 2012;403–408:2087–2092. DOI: 10.4028/www.scientific.net/ AMR.403-408.2087
- [13] Fujihira K. An approach to sustainable development by applying control science. In: Ghenai C, editor. Sustainable Development – Policy and Urban Development – Tourism, Life Science, Management and Environment. Rijeka, Croatia: InTech; 2012. pp. 299–318.

- [14] Ministry of Land, Infrastructure, Transport and Tourism. Japan Housing Performance Indication Standards (in Japanese) [Internet]. 2001. Available from: http://www.mlit. go.jp/jutakukentiku/house/torikumi/hinkaku/061001hyouji.pdf [Accessed: 2016-06-10]
- [15] JaGBC, JSBC. CASBEE for Detached Houses (for New Construction) Technical Manual 2010 Edition (in Japanese). Tokyo, Japan: IBEC; 2010.
- [16] Ministry of Land, Infrastructure, Transport and Tourism. Concerning Long-life Quality Housing Certification System (in Japanese) [Internet]. Available from: http://www.mlit. go.jp/common/001008009.pdf [Accessed: 2016-06-10]
- [17] Fujihira K. Inagi EcoHouse: 500% energy self-sufficient home (in Japanese). IBEC. 2015;36(1):48–51.
- [18] JaGBC, JSBC. CASBEE for Detached Houses (for Existing Building) Technical Manual 2011 Edition (in Japanese). Tokyo, Japan: IBEC; 2011.
- [19] Howden-Chapman P, et al. Effect of insulating existing houses on health inequality: cluster randomised study in the community. BMJ. 2007;334. DOI: http://dx.doi.org/10.1136/ bmj.39070.573032.80
- [20] Ikaga T, Eguchi R, Murakami S, Iwamae A, Hoshi T, Mizuishi T, Kawakubo S, Okumura K. Evaluation of investment in residential thermal insulation considering non-energy benefits delivered by health (in Japanese). Journal of Environmental Engineering (Transactions of AIJ). 2011;76:735–740. DOI: http://doi.org/10.3130/aije.76.735
- [21] Wilson A. Passive Survivability: A New Design Criterion for Buildings [Internet]. 2006. Available from: https://www2.buildinggreen.com/article/passive-survivability-newdesign-criterion-buildings [Accessed: 2016-06-10]
- [22] IBEC. CASBEE for Detached Houses (for New Construction) Evaluation results: Inagi EcoHouse (in Japanese) [Internet]. Available from: http://www.ibec.or.jp/CASBEE/DH_ certification/kohyo/IBEC-L0040H.pdf [Accessed: 2016-06-10]
- [23] Mitsubishi Research Institute, Research Division of Environment and Energy. Survey of Energy Consumption in Fiscal 2012: Fact-Finding Survey on Energy Consumption in the Consumer Sector (in Japanese) [Internet]. 2013. Available from: http://www.meti.go.jp/ meti_lib/report/2013fy/E003078.pdf [Accessed: 2016-06-10]

Chapter 4

Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic Robot Using Delayed Feedback Control

Dasheng Liu

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/65831

Abstract

Open-loop dynamic characteristics of an underactuated system with nonholonomic constraints, such as a horizontal bar gymnastic robot, show the chaotic nature due to its nonlinearity. This chapter deals with the stabilization problems of periodic motions for the giant swing motion of gymnastic robot using chaos control methods. In order to make an extension of the chaos control method and apply it to a new practical use, some stabilization control strategies were proposed, which were, based on the idea of delayed feedback control (DFC), devised to stabilize the periodic motions embedded in the movements of the gymnastic robot. Moreover, its validity has been investigated by numerical simulations. First, a method named as prediction-based DFC was proposed for a two-link gymnastic robot using a Poincaré section. Meanwhile, a way to calculate analytically the error transfer matrix and the input matrix that are necessary for discretization was investigated. Second, an improved DFC method, multiprediction delayed feedback control, using a periodic gain, was extended to a four-link gymnastic robot. A set of plural Poincare maps were defined with regard to the original continuous-time system as a T-periodic discrete-time system. Finally, some simulation results showed the effectiveness of the proposed methods.

Keywords: gymnastic robot, nonholonomic system, giant swing motion, delayed feedback control, chaos control, stabilization



© 2016 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

1. Introduction

In the past decades, scientists have found that unusual and unexpected evolution patterns arise frequently in numerous natural and model nonlinear systems in physics, chemistry, biology, engineering, medicine, economics, and so forth, since Lorenz [1] in 1963, May [2] in 1976, and others reported chaotic behavior in very simple dynamical models. The most peculiar aspect of these patterns is their random-like behavior, although the systems are deterministic, in other words, the deterministic nature of these systems does not make them predictable [3]. This behavior is known as deterministic chaos, or simply chaos. It is considered that chaos is due to sensitive dependence on the initial conditions. This property implies that two trajectories emerging from two different close-by initial conditions separate exponentially in the course of time. Due to this property, and due to the fact that, in general, experimental initial conditions are never known perfectly, chaotic systems cannot be predicted by their long-term behaviors.

This feature of their critical sensitivity to initial conditions, often regarded as a troublesome property, made chaos undesirable in engineering control practice, and most experimentalists considered such characteristics as something to be strongly avoided since it restricts the operating range of many electronic and mechanic devices. Besides that, chaotic systems exhibit two other important properties. One is that there is an infinite number of unstable periodic orbits embedded in the underlying chaotic set. The other is that the dynamics in the chaotic attractor is ergodic; that is to say, the system visits ergodically small neighborhood of every unstable fixed point during its temporal evolution [4].

Although the existence of steady states and an infinity of different unstable periodic orbits embedded in chaotic motion is not usually obvious in free-running chaotic evolution, these orbits offer a great potential advantage if one wants to control a chaotic system. The presence of chaos may be a great advantage for control in a variety of situations. In a nonchaotic system, small controls typically can only change the system dynamics slightly. However, in a chaotic system, one can choose between a rich variety of dynamical behaviors. It is then not surprising that the matter of controlling chaotic systems has come under detailed investigation by several different scientific communities. Chaotic dynamics consists of a motion where the system moves in the neighborhood of one of the unstable periodic orbits (UPO) for a while, then falls close to a different UPO and remains for a limited time. Thus, it allows to exploit a single dynamical system for the production of a large number of different yields [5]. The result is to render an otherwise chaotic motion more stable and predictable, which is often an advantage. The perturbation must be tiny, to avoid significant modification of the system's natural dynamics.

Since Ott, Grebogi, and Yorke [6], in 1990, pointed out the existence of many unstable periodic orbits (UPOs) embedded in chaotic attractors that raise the possibility of using very small external forces to obtain various types of regular behavior, Pyragas [7] in 1992, proposed a

so-called "Delayed Feedback Control (DFC)" idea that an appropriate continuous controlling signal formed from the difference between the current state and the delayed state is injected into the system, whose intensity is practically zero as the system evolves close to the desired periodic orbit but increases when it drifts away from the desired orbit, several techniques were devised for controlling chaos [8–21] during the past years and applied to various systems [22–31]. It is worth noting that in spite of the enormous number of applications among the chaos control, very few rigorous results are so far available. Most results are justified by computer simulations rather than by analytical tools. Therefore, many problems remain unsolved.

On the other hand, recently in the field of mechanical and control engineering, the open-loop link mechanism, in terms of diversity and ease of movement, has been used in industrial robots, although a large control input is needed due to speeding up, and the opposing force to the joint and the base increases. Therefore, the actuator with a large driving performance is needed, and the mass of the entire mechanism is increased. It is thought that the control method using the opposing force and gravity is important to achieve the desired movement with high speed and high efficiency by the limited driving performance. The joint type movement robot is actively researched in recent years, and the achievement of dynamic walking and running with high speed and efficiency becomes a problem. Although living creatures, such as humans or animals, have achieved the high-speed movement by using the open-loop link mechanism, it is thought that this is because that a good movement possessing the energy efficiency and small joint drive power and small impact power, and so forth in the gravitational filed had been acquired adaptively during the process of evolution. For example, the human biped locomotion is performed by dynamic link mechanism coupled to the pendulum motion [32, 33], and brachiation (movement across branches) used by the monkeys is a specialized form of arboreal locomotion that can be seen as a continuous pendulum motion [34, 35]. In addition, humans have competed for the manipulation of their own multi-degree of freedom link mechanism through sports or physical exercise, and so forth, and the achievement of a complex movement meeting a specific initial terminal requirement as a technique.

The systems, such as humans and robots, which control the internal force to achieve motions by using reaction force from the outside world, can be modeled as an underactuated link mechanism. These systems usually have nonholonomic second-order constraints due to the presence of passive joints, which makes development of control methods a difficult task. In recent years, studies on underactuated mechanical systems that possess fewer actuators than degrees of freedom have received increased interest [36–40].

Open-loop dynamic characteristics of an underactuated system with nonholonomic constraints show the chaotic nature that even though there are small differences in the initial conditions, the amplitude grows to be a completely different movement for its nonlinearity due to centrifugal force, Coriolis force, and gravity. Motions control of underactuated systems known also as nonholonomic control, such as wheeled robot positioning control [41], aerial posture control of space robots [42], and positioning control of underactuated link mechanism robot [43, 44], have been studied. However, a generalized control method has not yet been established to this kind of system due to the difficulties in analysis, and the fact that for an underactuated system, it cannot be directly controlled because of the generalized coordinates of its passive parts. Control methods, which utilize empirical skills catching the motion characteristics and proper motion of the system, are desirable. From the above-mentioned view-point, in order to achieve a robot with a high-speed, highly effective optimal motion, and the skill in movement seen in living creatures, fundamental researches on the control method and the trajectory planning method using periodic free movement for the linkage with a passive joint have been done [45–51].

The remaining parts of this chapter are organized as follows: Section 2 discusses a kind of DFC method, PDFC, for a two-link gymnastic robot by using of a Poincaré section. In the Section 3, an improved DFC method, Multiprediction Delayed Feedback Control, is extended to a four-link gymnastic robot. Section 4 gives some numerical simulations to show the effectiveness of the proposed method. Section 5 summarizes the chapter.

2. Delayed feedback control for a two-link horizontal bar gymnastic robot

The two-link horizontal bar gymnastic robot is a highly simplified model of a human gymnast on a high bar, where the underactuated first joint models the gymnast's hands on the bar and the actuated second joint models the gymnast's waist [52]. Studies on such underactuated mechanical systems that possess fewer actuators than degrees of freedom, as gymnastic robot with the underactuated first joint, have received considerable interest in recent years. Openloop dynamic characteristics of such a linkage as this kind of system, which is classified as an underactuated systems with nonholonomic constraints, show the chaotic nature that even though there are small differences in initial conditions, the amplitude grows to be a completely different movement for its nonlinearity due to centrifugal force, Coriolis force, and gravity. Since a generalized control method has not yet been established to this kind of system for the difficulties in analysis, their control problems are challenging.

While applying the original DFC to the continuous system, such as gymnastic robot, stability analysis of the closed-loop system becomes a very difficult task because of the time-delay dynamics described by a difference-differential equation. A modified delayed feedback control method, called Prediction-based DFC (PDFC), is proposed to stabilize the giant swing motion of two-link horizontal bar gymnastic robot.

2.1. Two-link gymnastic robot model

The two-link horizontal bar gymnastic robot is a simplified model of a human gymnast on a high bar, where the first joint being passive models the gymnast's hands on the bar and the second joint comprising of an actuator models the gymnast's hips. The formula m_i , l_i , a_i , I_i (i = 1, 2) in **Figure 1** denotes the *i*th link's mass, length, distance from joint to its center of mass, inertia moment around its center of mass, respectively.
Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic... 61 http://dx.doi.org/10.5772/65831



Figure 1. Two-link gymnastic robot model.

The equation of motion of the gymnastic robot is as follows:

$$\begin{bmatrix} u_1 \\ u_2 \end{bmatrix} = \underbrace{\begin{bmatrix} M_{11} & M_{12} \\ M_{21} & M_{22} \end{bmatrix}}_{M} \begin{bmatrix} \ddot{x}_1 \\ \ddot{x}_2 \end{bmatrix} + \underbrace{\begin{bmatrix} C_1 \\ C_2 \end{bmatrix}}_{C} + \underbrace{\begin{bmatrix} G_1 \\ G_2 \end{bmatrix}}_{G}, \tag{1}$$

where $\mathbf{x} = (x_1, x_2)^T \in \mathbb{R}^2$ is the generalized coordinate vector, $\mathbf{u} = (u_1, u_2)^T \in \mathbb{R}^2$ is the joint torque vector, and $M \in \mathbb{R}^{2 \times 2}$, $C \in \mathbb{R}^{2 \times 1}$, $G \in \mathbb{R}^{2 \times 1}$ denote the inertia matrix, Coriolis matrix, and gravitational matrix, respectively.

$$M_{11} = I_1 + I_2 + m_1 a_1^2 + m_2 (l_1^2 + a_2^2 + 2l_1 a_2 \cos x_2)$$

$$M_{12} = I_2 + m_2 (a_2^2 + l_1 a_2 \cos x_2)$$

$$M_{21} = I_2 + m_2 (a_2^2 + l_1 a_2 \cos x_2)$$

$$M_{22} = I_2 + m_2 a_2^2$$

$$C_1 = -(\dot{x}_2^2 + 2\dot{x}_1 \dot{x}_2) m_2 l_1 a_2 \sin x_2$$

$$C_2 = \dot{x}_1^2 m_2 l_1 a_2 \sin x_2$$

$$G_1 = g\{(m_1a_1 + m_2l_1)\sin x_1 + m_2a_2\sin (x_1 + x_2)\}$$
$$G_2 = gm_2a_2\sin (x_1 + x_2)$$

In addition, since the first joint cannot generate active torque, the following constraint must be satisfied.

$$u_1 \equiv h(\ddot{\mathbf{x}}, \dot{\mathbf{x}}, \mathbf{x}) = 0 \tag{2}$$

In order to facilitate the analysis, we rewrite the motion Eq. (1) into the following equation of state:

$$\dot{\mathbf{x}} = \begin{bmatrix} \dot{\mathbf{x}} \\ M^{-1}(\mathbf{x})(-C(\mathbf{x},\dot{\mathbf{x}})-G(\mathbf{x})) \end{bmatrix} + \begin{bmatrix} 0 \\ M^{-1}(\mathbf{x})E \end{bmatrix} u \equiv f(x,u), \tag{3}$$

where

$$M(\mathbf{x}) = \begin{bmatrix} c_1 + c_2 + 2d\cos x_2 & c_2 + d\cos x_2 \\ c_2 + d\cos x_2 & c_2 \end{bmatrix},$$

$$C(\mathbf{x}, \dot{\mathbf{x}}) = \begin{bmatrix} -(\dot{x}_2^2 + 2\dot{x}_1 \dot{x}_2)d\sin x_2 \\ \dot{x}_1^2 d\sin x_2 \end{bmatrix},$$

$$G(\mathbf{x}) = \begin{bmatrix} g_1 \sin x_1 + g_2 \sin (x_1 + x_2) \\ g_2 \sin (x_1 + x_2) \end{bmatrix},$$

$$E = \begin{bmatrix} 0 \\ 1 \end{bmatrix}.$$
(4)

Here, $x = [\mathbf{x}, \dot{\mathbf{x}}]^T \in \mathbb{R}^4$ denotes the state vector, and $u \in \mathbb{R}^1$ represents the control input. Meanwhile, the variables among the above equations are defined as follows:

$$c_{1} = I_{1} + m_{1}a_{1}^{2} + m_{2}l_{1}^{2}C$$

$$c_{2} = I_{2} + m_{2}a_{2}^{2}C$$

$$d = m_{2}l_{1}a_{2}C$$

$$g_{1} = g(m_{1}a_{1} + m_{2}l_{1})C$$

$$g_{2} = gm_{2}a_{2}.$$

2.2. Prediction-based delayed feedback control

As is known, the stability of a periodic orbit of the original continuous-time system is closely related to the stability of the fixed point of the corresponding Poincaré map. Since the stability of the periodic orbit means that the sequence of points converges to a fixed point in the phase plane, the objective system can be expressed as the following difference equation for the discrete-time systems.

Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic... 63 http://dx.doi.org/10.5772/65831

$$\tilde{x}(k+1) = A\tilde{x}(k) + Bu(k) \equiv f(\tilde{x}(k), u(k))$$
(5)

where $k, \tilde{x} \in \mathbb{R}^4, u \in \mathbb{R}^1$ denote the discrete time, the state error, and the control input, respectively. And $A \in \mathbb{R}^{4 \times 4}$ is named as the error transfer matrix (ETM), $B \in \mathbb{R}^{4 \times 1}$ is called the Input Matrix.



Figure 2. Poincaré map of a closed periodic trajectory.

2.2.1. Solution for obtaining the error transfer matrix A

Here, let $y_1 = \dot{x}_1$, $y_2 = \dot{x}_2$. Then the state vector of two-link gymnastic robot can be written in the following form:

$$x = (x_1, x_2, y_1, y_2) \tag{6}$$

As shown in **Figure 2**, the (k + 1)th state vector $x_{p,k+1}$ can be calculated by one-cycle integration in accordance with the control law by defining the *k*th state vector passing through the section P as $x_{p,k}$.

This relationship representing this state transfer map is defined as ϕ as follows:

$$x_{p,k+1} = \phi(x_{p,k}) \tag{7}$$

If given an initial condition from which a periodic orbit can be formed, it will return to the same point after one period. Hence, the following equation holds:

$$x_{p,ref} = \phi(x_{p,ref}) \tag{8}$$

While a difference occurs in the initial state, the motion will stray away from the periodic orbit. Suppose that there was an error corresponding to $e_{p,k}$ in *k*th state vector, the (k + 1)th state $x_{p,k+1}$ can be described as

$$x_{p,k+1} = \phi(x_{p,ref} + e_{p,k}).$$
 (9)

Using the above equation's Taylor series at $e_{v,k}$ and neglecting the higher-order terms can yield

$$e_{p,k+1} = \frac{\partial \phi(x_{p,ref})}{\partial x_p} e_{p,k} \equiv A e_{p,k}.$$
 (10)

It is obvious that $\partial \phi(x_{p,ref})/\partial x_p$ is equivalent to the ETM *A* defined in Eq. (5). It can be judged from this equation that if the absolute value of the ETM's eigenvalue >1, then the error increases with each cycle, and if <1, the motion will approach asymptotically the periodic orbit. However, because the state transfer mapping cannot be stated explicitly as a function, in previous studies, this ETM had been computed based on numerical differentiation. In this work, we obtained ETM with an analytical method by using variational equation.

Now consider the objective system given by the equation of motion in Eq. (3). First, given the case of u = 0, the objective system becomes

$$\dot{x} = f(x(t)) \equiv [f_1, f_2, f_3, f_4]^T$$
 (11)

Assume the state *x* to be the following vector function

$$\mathbf{x}(t) = \boldsymbol{\varphi}(t, \mathbf{x}) \equiv \left[\boldsymbol{\varphi}_1, \boldsymbol{\varphi}_2, \boldsymbol{\varphi}_3, \boldsymbol{\varphi}_4\right]^T,$$
(12)

where $f_{ii} \varphi_i$ (*i* = 1 ~ 4) are the functions of *x*. The solution of equation passing through *x*(0) at *t* = 0 is defined as follows:

$$x(0) = x_0.$$
 (13)

Moreover, the periodic solution satisfies the following equation.

$$x(T) = x(0) = \varphi(T, x_0)$$
(14)

Here, a Poincaré map is defined as follows:

$$P: \frac{\mathbf{R}^n \to \mathbf{R}^n \quad (n=4)}{x_0 \mapsto x_1 = P(x_0) = \varphi(T, x_0)}$$

Then, the T-period sequence of points obtained by sampling the trajectory from the initial value x_0 is

$$\{x_0, x_1, .., x_k, ...\} = \{x_0, P(x_0), P^2(x_0), .., P^k(x_0), ...\}.$$
(15)

Thus, it is obvious that the periodic solution is shown as a point passing through the same position. Substituting $x = \varphi(t, x_0)$ into Eq. (11), we obtain an equation for the variable $\varphi(t, x_0)$, which can be written as

$$\frac{d\varphi(t,x_0)}{dt} = f(\varphi(t,x_0)). \tag{16}$$

Next, taking differentiation for Eq. (16) at x_0 , we can obtain

$$\frac{\partial}{\partial x_0} \left(\frac{d\varphi}{dt}(t, x_0) \right) = \frac{\partial}{\partial x_0} (f(\varphi(t, x_0))).$$
(17)

The differential order of the left-hand side in the above equation can be modified as

$$\frac{d}{dt}\left(\frac{\partial\varphi}{\partial x_0}(t,x_0)\right) = \frac{\partial f}{\partial x}(\varphi(t,x_0))\frac{\partial\varphi}{\partial x_0}(t,x_0) \equiv \tilde{A}(t)\frac{\partial\varphi}{\partial x_0}(t,x_0).$$
(18)

Thus, it becomes a time-varying linear matrix differential equation with the form of

$$\frac{dX(t,x_0)}{dt} = \tilde{A}(t)X(t,x_0),\tag{19}$$

which is called the Variational Equation if defining $X(t, x_0) = \frac{\partial \varphi}{\partial x_0}(t, x_0)$. From Eqs. (11) and (12), the above equation is equivalent to

$$\frac{d}{dt} \begin{bmatrix} \partial \varphi_1 / \partial x_{i,0} \\ \partial \varphi_2 / \partial x_{i,0} \\ \partial \varphi_3 / \partial x_{i,0} \\ \partial \varphi_4 / \partial x_{i,0} \end{bmatrix} = \begin{bmatrix} \partial f_1 / \partial x_1 & \partial f_1 / \partial x_2 & \partial f_1 / \partial y_1 & \partial f_1 / \partial y_2 \\ \partial f_2 / \partial x_1 & \partial f_2 / \partial x_2 & \partial f_2 / \partial y_1 & \partial f_2 / \partial y_2 \\ \partial f_3 / \partial x_1 & \partial f_3 / \partial x_2 & \partial f_3 / \partial y_1 & \partial f_3 / \partial y_2 \\ \partial f_4 / \partial x_1 & \partial f_4 / \partial x_2 & \partial f_4 / \partial y_1 & \partial f_4 / \partial y_2 \end{bmatrix} \begin{bmatrix} \partial \varphi_1 / \partial x_{i,0} \\ \partial \varphi_2 / \partial x_{i,0} \\ \partial \varphi_3 / \partial x_{i,0} \\ \partial \varphi_4 / \partial x_{i,0} \end{bmatrix},$$
(20)

where $x_{i,0}$ refers to the solution at t = 0 of the *i*th state corresponding to the state vector x defined in Eq. (12).

Note that $X(0, x_0) = I$. Carrying out integrals in numerical integration to the above variational equations over the interval $t \in [0, T]$, one column of the matrix $X(T, x_0)$ can be obtained. Hence, repeating it four times, all of its values can be calculated. It should be emphasized here that to solve these equations since the value of $\tilde{A}(t)$ is required, however, $\tilde{A}(t)$ contains x(t) in terms of the relevant time, they have to be solved together with Eq. (11) simultaneously. From Eqs. (10) and (15), it is not difficult to be verified that $X(T, x_0)$ can be regarded as ETM.

2.2.2. Solution for obtaining the input matrix B

Note that Eq. (3) can be rewritten into

$$\dot{x} = \tilde{f}(x, u) = f(x) + g(x)u, \tag{21}$$

where

$$g(x) = \begin{bmatrix} 0\\ M^{-1}(x)E \end{bmatrix} = \begin{bmatrix} 0\\ 0\\ g_3\\ g_4 \end{bmatrix}.$$
 (22)

_ _

The input torque *u* can be described as

$$u = \operatorname{rect}(t)u_0,\tag{23}$$

where

$$\operatorname{rect}(t) = \begin{cases} 1 & \text{if } kT < t \le kT + \tau \quad (k = 0, 1, 2, ...) \\ 0 & \text{otherwise} \end{cases},$$
(24)

Here, u_0 is the control input during (t_0, τ) , while $x = x_0$ and τ are control parameters.

Since the solution of equation at x = x(0) can be stated as

$$x(t) = \varphi(t, x_0, u_0).$$
 (25)

Define a Poincaré map as follows:

$$\tilde{P}: \begin{array}{l} \mathbb{R}^n \to \mathbb{R}^n \quad (n=4) \\ x_0 \mapsto x_1 = P(x_0, u_0) = \varphi(T, x_0, u_0) \end{array}$$

The continuous system described by Eq. (21) will change to be a discrete system by mapping of \tilde{P} . Substituting $x = \varphi(t, x_0, u_0)$ into Eq. (21), the following equation for the variable $\varphi(t, x_0)$ can be obtained.

$$\frac{d\varphi(t, x_0, u_0)}{dt} = f(\varphi(t, x_0, u_0)) + g(x) \operatorname{rect}(t) u_0$$
(26)

It is obvious here that by taking differentiation at x_0 , its result will be equivalent to Eq. (18). Consider here the case of differentiation at u_0 , which can yield

$$\frac{\partial}{\partial u_0} \left(\frac{d\varphi}{dt}(t, x_0, u_0) \right) = \frac{\partial}{\partial u_0} \left(f(\varphi(t, x_0, u_0)) + g(x) \operatorname{rect}(t) \right).$$
(27)

The differential order of left-hand side in the above equation can also be changed.

$$\frac{d}{dt}\left(\frac{\partial\varphi}{\partial u_0}(t,x_0,u_0)\right) = \frac{\partial f}{\partial x}(\varphi(t,x_0,u_0))\frac{\partial\varphi}{\partial u_0}(t,x_0,u_0) + g(x)\operatorname{rect}(t)$$
(28)

Let $X(t, x_0, u_0) = \partial \varphi / \partial x_0(t, x_0, u_0)$. Similar to previous subsection, the above equation is equivalent to the following equation.

$$\frac{d}{dt} \begin{bmatrix} \partial \varphi_1 / \partial u_0 \\ \partial \varphi_2 / \partial u_0 \\ \partial \varphi_3 / \partial u_0 \\ \partial \varphi_4 / \partial u_0 \end{bmatrix} = \begin{bmatrix} \partial f_1 / \partial x_1 & \partial f_1 / \partial x_2 & \partial f_1 / \partial y_1 & \partial f_1 / \partial y_2 \\ \partial f_2 / \partial x_1 & \partial f_2 / \partial x_2 & \partial f_2 / \partial y_1 & \partial f_2 / \partial y_2 \\ \partial f_3 / \partial x_1 & \partial f_3 / \partial x_2 & \partial f_3 / \partial y_1 & \partial f_3 / \partial y_2 \\ \partial f_4 / \partial x_1 & \partial f_4 / \partial x_2 & \partial f_4 / \partial y_1 & \partial f_4 / \partial y_2 \end{bmatrix} \begin{bmatrix} \partial \varphi_1 / \partial u_0 \\ \partial \varphi_2 / \partial u_0 \\ \partial \varphi_3 / \partial u_0 \\ \partial \varphi_4 / \partial u_0 \end{bmatrix} + \operatorname{rect}(t) \begin{bmatrix} 0 & 0 & g_3 & g_4 \end{bmatrix}^T$$
(29)

Note that $X(0, x_0, u_0) = 0$. Carrying out integrals in numerical integration to Eq. (29) over the interval $t \in [0, T]$, it is not difficult to obtain the value of the matrix $X(T, x_0, u_0)$ which can be regarded as the input matrix *B* defined in Eq. (5).

2.2.3. Stability of periodic orbits

Since the value of both the matrices *A* and *B* can be calculated analytically, i.e., the continuous system of two-link gymnastic robot can be expressed in a discrete model described in Eq. (5). In order to study the stabilization of gymnastic robot's continuous system by delayed feedback control, it is sufficient to consider the discrete model.

Consider the following Prediction-based feedback control:

$$u(k) = K\{x(k) - x_P(k)\},$$
(30)

where $K \in \mathbb{R}^{1 \times 4}$ is a feedback gain, x(k) is the state vector at *k*-step, $x_P(k)$ denotes one period future states of uncontrolled system which can be obtained from Eq. (3). Moreover, notice that the following almost equality holds.

$$x(k) - x_P(k) \approx \tilde{x}(k) - f(\tilde{x}(k), 0)$$
(31)

Here $f(\tilde{x}(k), 0)$ stands for the state error at (k + 1)th step. From Eq. (5), it is obvious that $f(\tilde{x}(k), 0) = A\tilde{x}(k)$ holds.

Therefore, the closed-loop system can be described by

$$\tilde{x}(k+1) = A\tilde{x}(k) + BK(I-A)\tilde{x}(k)$$

$$= \{A + BK(I-A)\}\tilde{x}(k).$$
(32)

Let

$$\widehat{K} = K(I - A), \tag{33}$$

then the stabilization problem is reduced as follows:

Given a system Eq. (5), find a feedback gain \hat{K} that places the closed-loop poles of the system in the set

$$\Lambda = \{ z \in \mathbf{C} : |z| < 1 \}.$$
(34)

By using the pole placement technique, it is not difficult to obtain the value of \hat{K} , and if *det* $(I - A) \neq 0$, then the feedback gain *K* is given by

$$K = K(I - A)^{-1}.$$
 (35)

Thus, the above design procedure can be applied to analyze the stability of two-link gymnastic robot system.

3. Delayed feedback control for a four-link horizontal bar gymnastic robot

As is known, the simplified two-link robot shown in the previous section is not an ideal physical model of a human gymnast on a high bar. In order to mimic gymnastic routine more realistically, the complicated robot model with higher degrees of freedom (DOF) needs to be considered. An improved method based on PDFC which control a three-link gymnastic robot via a periodic gain has been proposed [51]. In this section, the Multiprediction Delayed Feedback Control (MDFC), is extended to a more complicated gymnastic robot with four DOF.

3.1. Four-link gymnastic robot model

Figure 3 shows a four-link horizontal bar gymnastic robot model, which consists of four links and four joints. Shoulder, hips, and knees are active, while the pivot connecting the hand and bar is a passive joint. Assume m_i , l_i , a_i , I_i (i = 1, 2, 3, 4) to be the link mass, length, distance from joint to its center of mass, and inertia moment around its center of mass, respectively.

The motion equation of the robot is

$$\begin{bmatrix} u_1 \\ u_2 \\ u_3 \\ u_4 \end{bmatrix} = \begin{bmatrix} M_{11} & M_{12} & M_{13} & M_{14} \\ M_{21} & M_{22} & M_{23} & M_{24} \\ M_{31} & M_{32} & M_{33} & M_{34} \\ M_{41} & M_{42} & M_{43} & M_{44} \end{bmatrix} \begin{bmatrix} \ddot{x}_1 \\ \ddot{x}_2 \\ \ddot{x}_3 \\ \ddot{x}_4 \end{bmatrix} + \begin{bmatrix} C_1 \\ C_2 \\ C_3 \\ C_4 \end{bmatrix} + \begin{bmatrix} G_1 \\ G_2 \\ G_3 \\ G_4 \end{bmatrix},$$
(36)

where $\mathbf{x} = (x_1, x_2, x_3, x_4)^T \in \mathbb{R}^4$ is the generalized coordinate vector, $(u_1, u_2, u_3, u_4)^T \in \mathbb{R}^4$ is the joint torque vector, and M_{ij} , C_i , G_i (i, j = 1, 2, 3, 4) are, respectively, the terms of inertia matrix, Coriolis matrix, and gravitational matrix. Each item of M_{ij} , C_i , G_i is as follows. Note that $M_{ij} = M_{ji}$, $(i \neq j)$.

$$\begin{split} \mathrm{M11} &= m_2 \big(2a_2 l_1 \cos \left(x_2 \right) + a_2^2 + l_1^2 \big) + m_3 \{ 2a_3 l_1 \cos \left(x_2 + x_3 \right) + 2a_3 l_2 \cos \left(x_3 \right) \\ &+ a_3^2 + 2l_1 l_2 \cos \left(x_2 \right) + l_1^2 + l_2^2 \} + m_4 \{ 2a_4 l_1 \cos \left(x_2 + x_3 + x_4 \right) + 2a_4 l_2 \cos \left(x_3 + x_4 \right) \\ &+ 2a_4 l_3 \cos \left(x_4 \right) + a_4^2 + 2l_1 l_2 \cos \left(x_2 \right) + 2l_1 l_3 \cos \left(x_2 + x_3 \right) + 2l_2 l_3 \cos \left(x_3 \right) + l_1^2 + l_2^2 + l_3^2 \} \\ &+ a_1^2 m_1 + I_1 + I_2 + I_3 + I_4; \end{split}$$

$$M12 = m_2 (a_2 l_1 \cos (x_2) + a_2^2) + m_3 \{a_3 l_1 \cos (x_2 + x_3) + 2a_3 l_2 \cos (x_3) + a_3^2 + l_1 l_2 \cos (x_2) + l_2^2\} + m_4 \{a_4 l_1 \cos (x_2 + x_3 + x_4) + 2a_4 l_2 \cos (x_3 + x_4) + 2a_4 l_3 \cos (x_4) + a_4^2 + l_1 l_2 \cos (x_2) + l_1 l_3 \cos (x_2 + x_3) + 2l_2 l_3 \cos (x_3) + l_2^2 + l_3^2\} + I_2 + I_3 + I_4;$$



Figure 3. Four-link gymnastic robot model.

$$\begin{split} \mathrm{M13} &= m_3 \big(a_3 l_1 \cos \left(x_2 + x_3 \right) + a_3 l_2 \cos \left(x_3 \right) + a_3^2 \big) + m_4 \big\{ a_4 l_1 \cos \left(x_2 + x_3 + x_4 \right) + a_4 l_2 \cos \left(x_3 + x_4 \right) \\ &+ 2 a_4 l_3 \cos \left(x_4 \right) + a_4^2 + l_1 l_3 \cos \left(x_2 + x_3 \right) + l_3 (l_2 \cos \left(x_3 \right) + l_3) \big\} + I_3 + I_4; \\ \mathrm{M14} &= m_4 \big(a_4 l_1 \cos \left(x_2 + x_3 + x_4 \right) + a_4 l_2 \cos \left(x_3 + x_4 \right) + a_4 l_3 \cos \left(x_4 \right) + a_4^2 \big) + I_4; \\ \mathrm{M22} &= m_3 \big(2 a_3 l_2 \cos \left(x_3 \right) + a_3^2 + l_2^2 \big) + m_4 \big\{ 2 a_4 l_2 \cos \left(x_3 + x_4 \right) + 2 a_4 l_3 \cos \left(x_4 \right) + a_4^2 \\ &+ 2 l_2 l_3 \cos \left(x_3 \right) + l_2^2 + l_3^2 \big\} + a_2^2 m_2 + I_2 + I_3 + I_4; \\ \mathrm{M23} &= m_3 \big(a_3 l_2 \cos \left(x_3 \right) + a_3^2 \big) + m_4 \big\{ a_4 l_2 \cos \left(x_3 + x_4 \right) + 2 a_4 l_3 \cos \left(x_4 \right) + a_4^2 \\ &+ l_2 l_3 \cos \left(x_3 \right) + l_3^2 \big\} + I_3 + I_4; \\ \mathrm{M24} &= m_4 \big(a_4 l_2 \cos \left(x_3 + x_4 \right) + a_4 l_3 \cos \left(x_4 \right) + a_4^2 \big) + I_4; \end{split}$$

-

$$\begin{split} \mathrm{M33} &= m_4 \left(2a_4 l_3 \cos\left(x_4\right) + a_4^2 + l_5^2 \right) + a_3^2 m_3 + l_3 + l_4; \\ \mathrm{M34} &= m_4 \left(a_4 l_3 \cos\left(x_4\right) + a_4^2 \right) + l_4; \\ \mathrm{M44} &= a_4^2 m_4 + l_4; \\ \mathrm{C1} &= m_3 \left\{ -a_3 l_2 \dot{q}_3 \left(2 \left(\dot{q}_1 + \dot{q}_2 \right) + \dot{q}_3 \right) \sin\left(x_3 \right) - a_3 l_1 \left(\dot{q}_2 + \dot{q}_3 \right) \left(2 \dot{q}_1 + \dot{q}_2 + \dot{q}_3 \right) \\ & \sin\left(x_2 + x_3 \right) - l_1 l_2 \dot{q}_2 \left(2 \dot{q}_1 + \dot{q}_2 \right) \sin\left(x_3 \right) - a_2 l_1 m_2 \dot{q}_2 \left(2 \dot{q}_1 + \dot{q}_2 \right) \sin\left(x_2 \right) \\ &+ m_4 \left\{ -a_4 l_3 \dot{q}_4 \left(2 \left(\dot{q}_1 + \dot{q}_2 + \dot{q}_3 \right) + \dot{q}_4 \right) \sin\left(x_4 \right) - a_4 l_2 \left(\dot{q}_3 + \dot{q}_4 \right) \left(2 \dot{q}_1 + 2 \dot{q}_2 + \dot{q}_3 + \dot{q}_4 \right) \\ & \sin\left(x_3 + x_4 \right) - a_4 l_1 \left(\dot{q}_2 + \dot{q}_3 + \dot{q}_4 \right) \left(2 \dot{q}_1 + \dot{q}_2 + \dot{q}_3 + \dot{q}_4 \right) \sin\left(x_3 \right) \\ & + l_2 l_3 \dot{q}_3^2 (-\sin\left(x_3\right)) - 2 l_2 l_3 \left(\dot{q}_1 + \dot{q}_2 \right) \dot{q}_3 \sin\left(x_3 \right) + l_2 l_3 \dot{q}_3^2 (-\sin\left(x_3\right)) - 2 l_2 l_3 \left(\dot{q}_1 + \dot{q}_2 \right) \dot{q}_3 \sin\left(x_3 \right) \\ & - 2 a_3 l_2 \dot{q}_2 \dot{q}_3 \sin\left(x_3 \right) + l_1 l_2 \dot{q}_1^2 \sin\left(x_2 + x_3 \right) - 2 a_3 l_2 \dot{q}_3 \dot{q}_1 \sin\left(x_3 \right) - a_3 l_2 \dot{q}_3^2 \sin\left(x_3 \right) \\ & - 2 a_4 l_3 \dot{q}_4 \dot{q}_1 \sin\left(x_4 \right) - 2 a_4 l_2 \dot{q}_3 \dot{q}_1 \sin\left(x_4 + a_4 l_4 \dot{q}_1^2 \sin\left(x_3 + x_4 \right) - a_4 l_3 \dot{q}_4^2 \sin\left(x_4 \right) \\ & - 2 a_4 l_3 \dot{q}_2 \dot{q}_4 \sin\left(x_4 \right) - 2 a_4 l_2 \dot{q}_3 \dot{q}_4 \sin\left(x_4 \right) - 2 a_4 l_2 \dot{q}_3 \dot{q}_4 \sin\left(x_3 + x_4 \right) - a_4 l_2 \dot{q}_4^2 \sin\left(x_3 + x_4 \right) \\ & - 2 a_4 l_3 \dot{q}_2 \dot{q}_4 \sin\left(x_4 \right) - 2 a_4 l_2 \dot{q}_2 \dot{q}_4 \sin\left(x_3 + x_4 \right) - a_4 l_2 \dot{q}_4^2 \sin\left(x_3 + x_4 \right) \\ & - 2 a_4 l_2 \dot{q}_2 \dot{q}_3 \sin\left(x_3 + x_4 \right) - 2 a_4 l_2 \dot{q}_3 \dot{q}_4 \sin\left(x_3 + x_4 \right) - a_4 l_2 \dot{q}_4^2 \sin\left(x_3 + x_4 \right) \\ & - 2 a_4 l_3 \dot{q}_2 \dot{q}_4 \sin\left(x_4 + a_4 l_2 \left(\dot{q}_1 + \dot{q}_2 \right)^2 \sin\left(x_3 + x_4 \right) - a_4 l_3 \dot{q}_4^2 \sin\left(x_3 + x_4 \right) \\ & - 2 a_4 l_3 \dot{q}_2 \dot{q}_3 \sin\left(x_4 + x_4 \right) - a_4 l_2 \dot{q}_3 \dot{q}_4 \sin\left(x_3 + x_4 \right) + a_4 l_2 \dot{q}_1^2 \sin\left(x_4 \right) \\ & - 2 a_4 l_3 \dot{q}_2 \dot{q}_3 \sin\left(x_4 + x_4 \right) + a_4 l_2 \left(\dot{q}_1 + \dot{q}_2 \right)^2 \sin\left(x_3 + x_4 \right) - a_4 l_3 \dot{q}_2 \dot{q}_3 \sin\left(x_4 \right) \\ & - 2 a_4 l_3 \dot{q}_2 \dot{q}_3 \sin\left(x_4 + x_4 \right) + a_4 l_3 \dot{q}_1^2 \sin\left(x_4 + x_4 \right) - a_4 l_3 \dot{q}_2^2 \sin\left(x_4 + x_4 \right) + a_4 l_4$$

 $G2 = g\{m_3(a_3 \sin (x_1 + x_2 + x_3) + l_2 \sin (x_1 + x_2)) + m_4(a_4 \sin (x_1 + x_2 + x_3 + x_4) + l_2 \sin (x_1 + x_2) + l_3 \sin (x_1 + x_2 + x_3)) + a_2 m_2 \sin (x_1 + x_2)\};$

$$G3 = g\{m_4(a_4 \sin (x_1 + x_2 + x_3 + x_4) + l_3 \sin (x_1 + x_2 + x_3)) + a_3 m_3 \sin (x_1 + x_2 + x_3)\};$$

$$G4 = a_4 g m_4 \sin (x_1 + x_2 + x_3 + x_4);$$

Similar to the motion equation of the two-link robot, Eq. (36) can be rewritten as follows.

$$\dot{x} = \begin{bmatrix} y \\ -M^{-1}(x)(C(x) + G(x)) \end{bmatrix} + \begin{bmatrix} 0 \\ M^{-1}(x)E \end{bmatrix} \begin{bmatrix} u_2 \\ u_3 \\ u_4 \end{bmatrix} \equiv f(x, u).$$
(37)

3.2. Multiprediction delayed feedback control

As is discussed in the reference [51], multiprediction delayed feedback control is based on a new discretization method to increase the control performance by introducing a notion of plural Poincaré maps, (P_i , P_2 ,..., P_N) which divide the first link angle $x_1 \in (-\pi, \pi)$ into N sections as shown in **Figure 4**.

By the *i*th (i = 1, 2, ..., N) Poincaré map, the four-link horizontal bar gymnastic robot system can be discretized into the following

$$x(k+1,i) = A_i x(k,i) + B_i u(k,i),$$
(38)

where *k* is the discrete time and $x(k, i)(i = 1, 2, ..., N) \in \mathbb{R}^n$, $A_i \in \mathbb{R}^{n \times n}(n = 8)$, $B_i \in \mathbb{R}^{n \times 3}$ denote, respectively, the state error, the error transfer matrix, and the input matrix in terms with the *i*th Poincaré map. Furthermore, u(k, i) is defined as



Figure 4. Image of Poincaré maps [51].

$$u(k,i) = K_i \{ x(k,i) - x(k+1,i) \} = \widehat{K}_i x(k,i).$$
(39)

Here, the equivalence of $\hat{K}_i \equiv K_i(I-A_i)$ and $K_i \equiv K_{i+N} \in \mathbb{R}^{3 \times n}$ are satisfied for each *i*. By introducing the variable of control period, τ , and defining u_0 as the control input to the system during (t_0, τ) while $x = x_0$, the input torque *u* is described as

$$u = \operatorname{rect}(t)u_0,\tag{40}$$

where rect(t) is a rectangular function as Eq. (41).

$$\operatorname{rect}(t) = \begin{cases} 1 & \text{if } (kT+t_i) < t \le (kT+t_i+\tau) & (k=0,1,2,\ldots) \\ 0 & \text{otherwise} \end{cases}$$
(41)

Here, t_i is the time at *i*th Poincaré map $P_i(i = 1, 2, ..., N)$.

Summarize the state variables at each one of the *i*th Poincaré section into one vector as follows:

$$X(k,i) = [x(k,i),...,x(k,N),x(k+1,1),...,x(k+1,i-1)]^{T} \in \mathbb{R}^{nN}$$
(42)

Therefore, the close-loop system relating to X(k, i) can be stated as the following discrete-time system with periodic *N*.

$$X(k, i+1) = F(i)X(k, i),$$
(43)

where,

$$F(i) = \begin{bmatrix} 0 & I & \dots & 0 & 0 \\ \vdots & \vdots & \ddots & \vdots & \vdots \\ 0 & 0 & \dots & 0 & I \\ A_i + B_i K_i & 0 & \dots & 0 & 0 \end{bmatrix}.$$
(44)

As it is known, the local stability of such a system can be determined by the eigenvalue of the monodromy matrix as follows:

$$\Phi_F(i+N,i) \equiv F(i+N-1)\cdots F(i+1)F(i).$$
(45)

The solution of the error transfer matrix A_i and the input matrix B_i is similar to the determination of the two-link gymnastic robot, so only the difference is shown below.

3.3. Solution for obtaining the error transfer matrix A_i

Here, let $y_1 = \dot{x}_1$, $y_2 = \dot{x}_2$, $y_3 = \dot{x}_3$, $y_4 = \dot{x}_4$. Then the state vector of four-link gymnastic robot becomes

Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic... 73 http://dx.doi.org/10.5772/65831

$$x = (x_1, x_2, x_3, x_4, y_1, y_2, y_3, y_4).$$
(46)

Consider the objective system given by the equation of motion in Eq. (37). First, given the case of u = 0, the objective system becomes

$$\dot{x} = f(x(t)) \equiv [f_1, f_2, f_3, f_4, f_5, f_6, f_7, f_8]^T$$
(47)

Assume the state *x* to be the following vector function

$$x(t) = \varphi(t, x) \equiv \left[\varphi_1, \varphi_2, \varphi_3, \varphi_4, \varphi_5, \varphi_6, \varphi_7, \varphi_8\right]^{T}.$$
(48)

where $f_{\tilde{\nu}} \varphi_{\tilde{\iota}}(\tilde{\iota} = 1 \sim 8)$ are the functions of *x*. Introducing the following definition,

$$\varphi_{x_{\bar{i},0}}^{m} = \frac{\partial \varphi_{m}}{\partial x_{\bar{i},0}} \quad (\tilde{i} = 1 \sim 8),$$

$$f_{x_{j}}^{m} = \frac{\partial f_{m}}{\partial x_{j}}, \quad f_{y_{j}}^{m} = \frac{\partial f_{m}}{\partial y_{j}} \quad (m = 1 \sim 8, j = 1 \sim 4)$$
(49)

the variational equation becomes

$$\frac{d}{dt} \begin{bmatrix} \varphi_{x_{i,0}}^{1} \\ \varphi_{x_{i,0}}^{2} \\ \varphi_{x_{i,0}}^{3} \\ \varphi_{x_{i,0}}^{4} \\ \varphi_{x_{i,0}}^{6} \\ \varphi_{x_{i,0}}^{8} \\ \varphi_{x_{i,0}}^{8}$$

where $x_{\tilde{i},0}$ refers to the solution at t = 0 of the \tilde{i} th state corresponding to the state vector x defined in Eq. (48).

Note that the following equation holds.

$$A = \begin{bmatrix} a_1 & a_2 & a_3 & a_4 & a_5 & a_6 & a_7 & a_8 \end{bmatrix}$$

$$a\tilde{i} = \begin{bmatrix} \varphi_{x_{\tilde{i},0}}^1, & \varphi_{x_{\tilde{i},0}}^2, & \varphi_{x_{\tilde{i},0}}^3, & \varphi_{x_{\tilde{i},0}}^4, & \varphi_{x_{\tilde{i},0}}^5, & \varphi_{x_{\tilde{i},0}}^6, & \varphi_{x_{\tilde{i},0}}^7, & \varphi_{x_{\tilde{i},0}}^8 \end{bmatrix}^T (\tilde{i} = 1 \sim 8).,$$
(51)

Carrying out integrals in numerical integration to the variational equations of Eq. (50) over the interval $t \in [0, T]$ by eight times, each item of the error transfer matrix, $A_i(i = 1, 2, ..., N)$, can be calculated.

3.4. Solution for obtaining the input matrix B_i

Note that Eq. (37) can be rewritten as

74 Complex Systems, Sustainability and Innovation

$$\dot{x} = \tilde{f}(x, u) = f(x) + g(x)u.$$
 (52)

Here,

$$g(x) = \begin{bmatrix} 0_{4\times3} \\ M^{-1}(x)E \end{bmatrix} = \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \\ g_{12} & g_{13} & g_{14} \\ g_{22} & g_{23} & g_{24} \\ g_{32} & g_{33} & g_{34} \\ g_{42} & g_{43} & g_{44} \end{bmatrix}.$$
(53)

Since the solution of equation at x = x(0) can be written as

$$x(t) = \varphi(t, x_0, u_0).$$
 (54)

The continuous system described by Eq. (53) will change to be a discrete system by mapping of \tilde{P}_i . Substituting Eq. (54) into Eq. (53), the following equation for the variable $\varphi(t, x_0)$ can be obtained.

$$\frac{d\varphi(t, x_0, u_0)}{dt} = f(\varphi(t, x_0, u_0)) + g(x) \operatorname{rect}(t) u_0$$
(55)

Here, consider the case of differentiation at u_0 , which can yield

$$\frac{\partial}{\partial u_0} \left(\frac{d\varphi}{dt}(t, x_0, u_0) \right) = \frac{\partial}{\partial u_0} (f(\varphi(t, x_0, u_0))) + g(x) \operatorname{rect}(t).$$
(56)

Let

$$\varphi_{u_{j,0}}^{m} = \frac{\partial \varphi_{m}}{\partial u_{j,0}} \quad (m = 1 \sim 8, j = 2, 3, 4).$$
(57)

. .

Similar to previous subsection, Eq. (53) is equivalent to the following equation. .

.

.

$$\frac{d}{dt} \begin{bmatrix} \varphi_{u_{i,0}}^{1} \\ \varphi_{u_{i,0}}^{2} \\ \varphi_{u_{i,0}}^{3} \\ \varphi_{u_{i,0}}^{4} \\ \varphi_{u_{i,0}}^{6} \\ \varphi_{u_{i,0}}^{8} \\ \varphi_{u_{i,0}}^{8} \\ \varphi_{u_{i,0}}^{8} \end{bmatrix} = \begin{bmatrix} f_{x_{1}}^{1} & f_{x_{2}}^{1} & f_{x_{3}}^{1} & f_{x_{4}}^{1} & f_{y_{1}}^{1} & f_{y_{2}}^{1} & f_{y_{3}}^{1} & f_{y_{4}}^{1} \\ f_{x_{1}}^{2} & f_{x_{2}}^{2} & f_{x_{3}}^{2} & f_{x_{4}}^{2} & f_{y_{2}}^{2} & f_{y_{3}}^{2} & f_{y_{4}}^{2} \\ f_{x_{1}}^{5} & f_{x_{2}}^{5} & f_{x_{3}}^{5} & f_{x_{4}}^{5} & f_{y_{1}}^{4} & f_{y_{2}}^{4} & f_{y_{4}}^{4} & f_{y_{4}}^{4} \\ f_{x_{1}}^{5} & f_{x_{2}}^{5} & f_{x_{3}}^{5} & f_{x_{4}}^{5} & f_{y_{2}}^{5} & f_{y_{3}}^{5} & f_{y_{4}}^{5} \\ f_{x_{1}}^{5} & f_{x_{2}}^{5} & f_{x_{3}}^{5} & f_{x_{4}}^{5} & f_{y_{1}}^{5} & f_{y_{2}}^{5} & f_{y_{3}}^{5} & f_{y_{4}}^{6} \\ f_{x_{1}}^{7} & f_{x_{2}}^{7} & f_{x_{3}}^{7} & f_{x_{4}}^{7} & f_{y_{1}}^{7} & f_{y_{2}}^{7} & f_{y_{3}}^{7} & f_{y_{4}}^{7} \\ f_{x_{1}}^{7} & f_{x_{2}}^{7} & f_{x_{3}}^{7} & f_{x_{4}}^{7} & f_{y_{1}}^{7} & f_{y_{2}}^{7} & f_{y_{3}}^{7} & f_{y_{4}}^{7} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{1}}^{8} & f_{y_{2}}^{8} & f_{y_{3}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{1}}^{8} & f_{y_{2}}^{8} & f_{y_{3}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{1}}^{8} & f_{y_{2}}^{8} & f_{y_{3}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{1}}^{8} & f_{y_{2}}^{8} & f_{y_{3}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{1}}^{8} & f_{y_{2}}^{8} & f_{y_{3}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{4}}^{8} & f_{y_{3}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{2}}^{8} & f_{x_{3}}^{8} & f_{x_{4}}^{8} & f_{y_{4}}^{8} & f_{y_{4}}^{8} & f_{y_{4}}^{8} & f_{y_{4}}^{8} \\ f_{x_{1}}^{8} & f_{x_{1}}^{8} &$$

.

.

Note that the following equation holds.

$$B = \begin{bmatrix} b_1 & b_2 & b_3 \end{bmatrix}, \\ b_{\tilde{i}} = \begin{bmatrix} \varphi_{u_{j,0}}^1, & \varphi_{u_{j,0}}^2, & \varphi_{u_{j,0}}^3, & \varphi_{u_{j,0}}^4, & \varphi_{u_{j,0}}^5, & \varphi_{u_{j,0}}^6, & \varphi_{u_{j,0}}^7, & \varphi_{u_{j,0}}^8 \end{bmatrix}^T (\tilde{i} = 1, 2, 3).$$
(59)

Carrying out integrals in numerical integration to Eq. (58) over the interval $t \in [0, T]$ by three times, it is not difficult to obtain the value of the input matrix $B_i(i = 1, 2, ..., N)$.

4. Numerical simulations

This section shows some simulation results to verify the validity of the proposed predictionbased delayed feedback control for a two-link horizontal bar gymnastic robot. The parameters of the robot model are shown in **Table 1**. The stability of giant swing motion with period T = 0.7(s) was examined.

		1st link	2nd link
Mass	m_i (kg)	39.5	20.7
Moment of inertia	I_i (kgm ²)	3.25	1.56
Link length	l_i (m)	1.2	0.879
Offset of mass center	a_i (m)	0.756	0.377

Table 1. Link parameter values.



Figure 5. Phase portrait of $X - \dot{x}$ (no error).

By means of the shooting method, an initial condition was obtained such as x (0) = (π , 0.0, 8.352, -1.19) that can achieve a free giant swing motion with a period of 0.7(s). During the first 30 s starting from this initial condition, the phase portrait of the uncontrolled system is depicted in **Figure 5** in the phase plane (X, \dot{x}) which refers to the angle and angular velocity of the link, respectively. It can be seen from **Figure 5** that the ideal phase plane orbit of the giant swing motions traces out a closed curve.

To see the stability of periodic orbits via delayed feedback control, numerical simulation was exercised from the initial condition $x(0) = (\pi, 0.1, 9.252, -2.39)$ that contained errors. First, consider the uncontrolled case, the 30 s trajectory of the orbits in the phase plane filled up a section of the phase space as in **Figure 6**. It can be confirmed that there was no periodicity without control.

Next, consider the system controlled by the proposed method, with which the error transfer matrix *A* was obtained as follows:

$A = \begin{bmatrix} 1\\ -0\\ 0\\ -1 \end{bmatrix}$	1.0054	0.0521	0.5986	0.0954	
	-0.0377	0.7685	-0.1017	0.0210	
	0.2982	1.8534	1.0041	0.0399	
	-1.7310	-10.6479	-0.0170	0.7728	

The input matrix *B* in the case of $\tau = 10$ (ms) is obtained as follows:



Figure 6. Phase portrait of $X-\dot{x}$ (with error).

Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic... 77 http://dx.doi.org/10.5772/65831

$$B = \begin{bmatrix} -0.000167\\ 0.000552\\ -0.004464\\ 0.024859 \end{bmatrix}$$

Based on the value of the matrix A and B shown in the above, the feedback gain K was obtained by assigning all of its poles of the system matrix $A + B\hat{K}$ in the unit disk. Generally, in a linear system, the poles of the system matrix $A + B\hat{K}$ are assigned as close as possible to the origin in order to decrease the convergence time; however, in such a nonlinear system as gymnastic robot, it is found that it becomes to show the divergence while the poles of $A + B\hat{K}$ is in the disk with ratio equals 0.6 approximately. Here, two cases were studied by assigning the poles larger than 0.6 as follows.

$$\begin{cases} Case \ I: \quad pole = (0.75, 0.73, 0.71, 0.69), & K = (-2242.7, -128.82, 48526, 8447.7), \\ Case \ II: \quad pole = (0.90, 0.88, 0.835, 0.815), & K = (-206.94, 354.01, 1577.1, 296.42). \end{cases}$$

In Case I, the numerical simulation results via calculated feedback gain *K* are shown in **Figure 7**. **Figure 7(a)** and **(b)** show the trajectory of the orbits in the phase plane (x, \dot{x}), of which **Figure 7(a)** depicts the trajectory of the first 50 s, while **Figure 7(b)** plots that of 50 ~ 60



Figure 7. Phase portrait and joint torque (Case I): (a) Phase portrait of $X - \dot{x}$ (t=0~50s); (b) Phase portrait of $X - \dot{x}$ (t=50~60s); (c) Joint torque.



Figure 8. Phase portrait and joint torque (Case II): (a) Phase portrait of $X-\dot{x}$ (t=0~150s); (b) Phase portrait of $X-\dot{x}$ (t=150~160s); (c) Joint torque.

(s). Meanwhile, the time history of control input is shown in **Figure 7(c)** from which it can be seen that the control input diminished with time. It is obvious that from **Figure 7(c)**, the control input *u* began to vanish from about t = 30(s), and from **Figure 7(b)** the periodicity began to appear from approximately t = 50(s).

However, on further analysis conducted, it was found that one can change the type of giant swing type by changing the pole placement of the matrix $A + B\hat{K}$. This phenomenon is shown in **Figure 8** which depicts the change histories of link angle, angular velocity, and input control with 160 s in Case II. The first 150 s trajectory of the orbits in the phase plane (x, \dot{x}) is depicted in **Figure 8(a)** and the remaining is shown in **Figure 8(b)**. Moreover, the time history of input torque is shown in **Figure 8(c)**.

It could be said that in Case II, the proposed method showed the ability of controlling the giant-swing motion to an unknown periodic orbit, which is known as one of the advantages of delayed feedback control.

Moreover, **Figures 9** and **10** depict the stick diagrams based on one period data after the giant swing motion converging to a stable orbit in Case I and Case II, in which the figures of (a), (b),

Stabilization Control for the Giant Swing Motion of the Horizontal Bar Gymnastic... 79 http://dx.doi.org/10.5772/65831



Figure 9. Angle, angular velocity, and stick (Case I).



Figure 10. Angle, angular velocity, and stick (Case II).

and (c) plot, respectively, the angle, angular velocity, and stick figure. From the above two figures, it can be seen that the difference between the assigned poles caused periodic giant swing motion to change to a different type.

5. Conclusion

This chapter studied the behavior of the giant swing motions of a two-link and a four-link horizontal bar gymnastic robot systems via delayed feedback control. First, it has been discussed as a solution to calculate analytically the error transfer matrix and the input matrix by which the stability of DFC for the original two-link gymnastic robot system is equivalent to that of the corresponding discrete system. Moreover, it was introduced as a method by which a feedback gain to ensure the stability of original system can be determined. Second, a modified DFC method, MDFC, has been extended to a four-link gymnastic robot. Plural Poincaré maps were defined so that the stability of the close-loop system can be evaluated based on the theory of monodromy matrix. Finally, the simulation results showed its effectiveness.

Acknowledgments

This work is supported in part by the Project Sponsored by the Scientific Research Foundation for the Returned Overseas Chinese Scholars, State Education Ministry, and in part by the Scientific Research Foundation for Young Teachers, Shanghai JiaoTong University.

Author details

Dasheng Liu

Address all correspondence to: dsliu@sjtu.edu.cn

Department of Instrument Science and Engineering, School of Electronic Information and Electrical Engineering, Shanghai JiaoTong University, Shanghai, China

References

- E. N. Lorenz. Deterministic nonperiodic flow. *Journal of the Atmospheric Sciences*, 20:130– 141, 1963.
- [2] R. M. May. Simple mathematical models with very complicated dynamics. *Nature*, 261:459–467, 1976.
- [3] W. Charlotte. What are the new implications of chaos for unpredictability? *The British Journal for the Philosophy of Science*, 60:195–220, 2009.
- [4] S. Boccaletti, C. Grebogi, Y. C. Lai, H. Mancini, and D. Maza. The control of chaos: theory and applications. *Physics Reports*, 329(3):103–197, 2000.
- [5] F. T. Arecchi, S. Boccaletti, M. Ciofini, R. Meucci, and R. Meucci. The control of chaos: theoretical schemes and experimental realizations. *International Journal of Bifurcation and Chaos*, 8(8):1643–1655, 1998.
- [6] E. Ott, C. Grebogi, and J. A. Yorke. Controlling chaos. *Physical Review Letters*, 64(11):1196– 1199, 1990.
- [7] K. Pyragas. Continuous control of chaos by self-controlling feedback. *Physics Letters A*, 170(6):421–428, 1992.
- [8] K. Pyragas. Control of chaos via extended delay feedback. *Physics Letters A*, 206(5–6):323–330, 1995.
- [9] M. E. Bleich and J. E. S. Socolar. Stability of periodic orbits controlled by time-delay feedback. *Physics Letters A*, 210(1–2):87–94, 1996.

- [10] S. Rajasekar, K. Murali, and M. Lakshmanan. Control of chaos by nonfeedback methods in a simple electronic circuit system and the Fitzhugh-Nagumo equation. *Chaos, Solitons* and Fractals, 8:1545–1558, 1997.
- [11] A. Zhalnin. Control of chaos in nonautonomous systems with quasiperiodic excitation. *Technical Physics Letters*, 25:662–664, 1999. DOI:10.1134/1.1262590.
- [12] P. J. Aston and C. M. Bird. Using control of chaos to refine approximations to periodic points. *International Journal of Bifurcation & Chaos*, 10:227–235, 2000.
- [13] M. Basso, R. Genesio, and A. Tesi. Stabilizing periodic orbits of forced systems via generalized pyragas controllers. *IEEE Transactions on Circuits and Systems I: Fundamental Theory and Applications*, 44(10):1023–1027, 1997.
- [14] K. Pyragas. Control of chaos via an unstable delayed feedback controller. *Physical Review Letters*, 86(11):2265–2268, 2001.
- [15] Y. P. Tian. Controlling chaos using invariant manifolds. International Journal of Control, 72:258–266, 1999.
- [16] F. T. Arecchi and S. Boccaletti. Adaptive strategies for recognition, noise filtering, control, synchronization and targeting of chaos. *Chaos*, 7:621–634, 1997.
- [17] K. Konishi and H. Kokame. Observer-based delayed-feedback control for discrete-time chaotic systems. *Physics Letters A*, 248(5–6):359–368, 1998.
- [18] T-D Eom, M Sugisaka, and J-J Lee. New skill learning paradigm using various kinds of neurons. *Applied Mathematics and Computation*, 91(1):9–22, 1998.
- [19] O. Calvo and J. H. E. Cartwright. Fuzzy control of chaos. International Journal of Bifurcation and Chaos, 8(8):1743–1747, 1998.
- [20] T Useo and S Yamamoto. A generalization of delayed feedback control in chaotic discrete-time systems. *IEICE Technical Report. Nonlinear Problems: The Institute of Electronics, Information and Communication Engineers*, 97(592):17–22, 1998. (In Japanese.)
- [21] S. Mizutani, T. Sano, T. Uchiyama, and N. Sonehara. Controlling chaos in chaotic neural networks. *IEICE Technical Report. Nonlinear Problems: The Institute of Electronics, Information and Communication Engineers*, 95(47):41–48, 1995. (In Japanese.)
- [22] A. L. Fradkov and R. J. Evans. Control of chaos: methods and applications in engineering. *Annual Reviews in Control*, 29(1):33–56, 2005.
- [23] E. Scholl, K. Pyragas, D. Cooper, and R. Dottling. Tuning of semiconductor oscillators by chaos control. *Semiconductor Science and Technology*, 9(5):559–563, 1994.
- [24] T. Hikihara and T. Kawagoshi. An experimental study on stabilization of unstable periodic motion in magneto-elastic chaos. *Physics Letters A*, 211(1):29–36, 1996.

- [25] Y. Nakamura, T. Suzuki, and M. Koinuma. Nonlinear behavior and control of a nonholonomic free-joint manipulator. *IEEE Transactions on Robotics and Automation*, 13 (6):853–862, 1997.
- [26] S. Bielawski, M. Bouazaoui, D. Derozier, and P. Glorieux. Stabilization and characterization of unstable steady states in a laser. *Physical Review A*, 47(4):3276–3279, 1993.
- [27] P. Parmananda, R. Madrigal, M. Rivera, L. Nyikos, I. Z. Kiss, and V. Gaspar. Stabilization of unstable steady states and periodic orbits in an electrochemical system using delayedfeedback control. *Physical Review E*, 59:5266–5271, 1999.
- [28] H. Qammar, K. R. Seshadhri, R. Gomatam, and A. Venkatesan. Control of a chaotic polymerization reaction using linear and nonlinear controllers. *The Chemical Engineering Journal and the Biochemical Engineering Journal*, 64(1):141–148, 1996.
- [29] J. A. Holyst and K. Urbanowicz. Chaos control in economical model by time-delayed feedback method. *Physica A: Statistical Mechanics and its Applications*, 287(3–4):587–598, 2000.
- [30] A. Tamaševičius, G. Mykolaitis, V. Pyragas, and K. Pyragas. Delayed feedback control of periodic orbits without torsion in nonautonomous chaotic systems: theory and experiment. *Physical Review E*, 76(2):026203, 2007.
- [31] M. Brucoli, D. Cafagna, L. Carnimeo, and G. Grassi. Synchronization of hyperchaotic circuits via continuous feedback control with application to secure communications. *International Journal of Bifurcation & Chaos*, 8:2031–2040, 1998.
- [32] A. Sano. Dynamic biped walking by using skillfully a gravity field (challenge to a human walking). *Journal of the Robotics Society of Japan*, 11(3):52–57, 1993. (In Japanese.)
- [33] K. Ono, R. Takahashi, A. Imadu, and T. Shimada. Self-excitation control for biped walking mechanism, Proceedings of the 2000 IEEE/RSJ Int. Conf. on Intelligent Robots and Systems, pp. 1149–1154.
- [34] J. Nakanishi, T. Fukuda, and D. E. Koditschek. Study on the control of a two-link brachiating robot: brachiation on a ladder with irregular intervals via target dynamics. *Trans*actions of the Japan Society of Mechanical Engineers. C, 65(639):4387–4394, 1999. (In Japanese.)
- [35] H. Nishimura, K. Takasaki, K. Funaki, and T. Totani. Motion control of brachiation robot by means of final-state control with error learning. *Transactions of the Japan Society of Mechanical Engineers. C*, 63(605):182–189, 1997. (In Japanese.)
- [36] S. Takashima. A robot performing gymnastics on the bar. *Journal of the Society of Biomechanisms*, 15(4):190–197, 1991. (In Japanese.)
- [37] M. W. Spong. The swing up control problem for the actobot. *IEEE Control Systems*, 15:49– 55, 1995.

- [38] A. Imadu and K. Ono. Optimum trajectory planning method for a system that includes passive joints: 1st report, proposal of a function approximation method. *Transactions of the Japan Society of Mechanical Engineers*. *C*, 64(618):516–522, 1998. (In Japanese.)
- [39] A. Imadu, K. Ono, and K. Yamamoto. Optimum trajectory planning method for a system that includes passive joints: 2nd report, analysis of the giant swing motions and their systematic search method based on optimum trajectory planning. *Transactions of the Japan Society of Mechanical Engineers. C*, 65(637):3583–3591, 1999. (In Japanese.)
- [40] N. Ogino, H. Kinjo, E. Uezato, and T. Yamamoto. Design of nonholonomic control system using neuro controller evolved by GA training. *IEICE Technical Report. Neurocomputing: The Institute of Electronics, Information and Communication Engineers*, 105(131):41–44, 2005. (In Japanese.)
- [41] M. Sampei. Vehicle control and nonlinear control theory: Control for nonholonomic systems. The Transactions of the Institute of Electrical Engineers of Japan. D, A Publication of Industry Applications Society, 114(10):955–958, 1994. (In Japanese.)
- [42] K. Senda, Y. Murotsu, and M. Ozaki. A method of attitude control for space robots: an approach using a neural network. *Transactions of the Japan Society of Mechanical Engineers*. *C*, 57(539):2356–2362, 1991. (In Japanese.)
- [43] Y. Nakamura, M. Koinuma, and T. Suzuki. Chaotic behavior and nonlinear control of a two-joint planer arm with a free joint-control of nonholonomic mechanisms with drift. *Journal of the Robotics Society of Japan*, 14(4):602–611, 1996. (In Japanese.)
- [44] H. Arai. Nonholonomic control of underactuated manipulators. Systems, Control and Information: Journal of the Institute of Systems, Control and Information Engineers, 43 (10):553–560, 1999. (In Japanese).
- [45] A. Imadu, K. Yamamoto, and K. Ono. Analysis of free giant swing motions of a 2-DOF link mechanism. *Transactions of the Japan Society of Mechanical Engineers*. C, 66(644):1102– 1109, 2000. (In Japanese.)
- [46] K. Yamamoto, K. Ono, and A. Imadu. Control of giant swing motions of a 2-DOF link mechanism. *Transactions of the Japan Society of Mechanical Engineers*. C, 66(651):3591–3599, 2000. (In Japanese.)
- [47] H. Yamaura, K. Ono, and H. Suzuki. Giant-swing motions of a 3-DOF link mechanism. Dynamics and Design Conference: Proceedings of Dynamics, Measurement and Control: D, 2002:150, 2002. (In Japanese.)
- [48] H. Yamaura, H. Suzuki, and K. Ono. Giant-swing motions and control of a 3-DOF link mechanism. *Transactions of the Japan Society of Mechanical Engineers*. C, 71(707):2289–2296, 2005. (In Japanese.)
- [49] H. Yamaura and M. Yanai. A realization method of giant-swing motion of 3-DOF link mechanism. *Transactions of the Japan Society of Mechanical Engineers*. C, 72(721):2812–2820, 2006. (In Japanese.)

- [50] D. Liu and H. Yamaura. Realization of Giant Swing Motions of a two-link horizontal bar gymnastic robot Using Delayed Feedback Control, Transactions of the Japan Society of Mechanical Engineers, 76-767, C(2010), pp.1700–1707. (In Japanese)
- [51] D. Liu and H. Yamaura. Stabilization control for giant swing motions of 3-link horizontal bar gymnastic robot using multiple-prediction delayed feedback control with a periodic gain. *Journal of System Design and Dynamics*, 5:42–54, 2011.
- [52] K. Ono, A. Imadu, and T. Sakai. Optimal motion trajectory of giant swing. *Transactions of the Japan Society of Mechanical Engineers*. C, 62(599):2640–2647, 1996. (In Japanese.)

Quantitative Methods in System-Based Drug Discovery

Yu-Chen Lo , Ren Gui , Hiroshi Honda and

Jorge Z. Torres

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/65833

Abstract

Modern pharmaceutical industries have faced significant challenges to deliver safe and effective medicines because of significant toxicity and severe side effects of discovered drugs. On the other hand, recent developments and advances in system-based pharmacology aim to address these challenges. In this chapter, we provide an overview of quantitative methods for system-based drug discovery. System-based drug discovery integrates chemical, molecular, and systematic information and applies this knowledge to the designing of small molecules with controlled toxicity and minimized side effects. First, we discuss current approaches for drug discovery and outline their advantages and disadvantages. Next, we introduce basic concepts of systems pharmacology with an emphasis on ligand-based drug discovery and target identification. This is followed by a discussion on structure-based drug design and statistical tools for pharmaceutical research. Finally, we provide an overview of future directions in systems pharmacology that will guide further developments.

Keywords: systems pharmacology, drug discovery, chemoinformatics

1. Introduction

The discovery of effective medicines has been a long-term human endeavor aimed at curing illnesses and improving physiological conditions. Early drug discovery methods often relied on serendipitous findings. For example, penicillin, a substance released by mold, was accidentally discovered to inhibit bacterial growth. However, with the advances of molecular cloning techniques, X-ray crystallography, robotics, and computational aided technology, drugs can now be rationally designed. Additionally, the marriage of combinatorial chemistry and robotics has also created a new drug discovery approach called "high-throughput



© 2016 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. screening." In a high-throughput screening campaign, a library consisting of millions of molecules is tested against a disease-relevant target to identify potential drug candidates.

However, the application of modern drug discovery methods has not directly translated into increases in the output of new drugs. Although many of these "designer" drugs have been optimized for binding and specificity, they are all low-weight molecular ligands and are likely to interact with multiple off-targets, which contribute to severe side effects in patients taking these medicines. Consequently, a system drug discovery approach that simultaneously optimizes drug binding, target promiscuity, and safety profile has been proposed. In particular, "poly-pharmacology" is a new drug discovery paradigm that aims to study the interactions of many drugs to many targets as many body problems. The binding profile of a compound can be used to modify a ligand structure to maximize on-target binding while minimizing off-target interactions. More recently, "structural poly-pharmacology" has also been proposed, which utilizes the structural data of proteins or drugs to gain a mechanistic understanding of drug action and side effects [1, 2].

System-based drug discovery can be classified as ligand-based and structure-based drug discovery approaches. In the ligand-based drug discovery approach, ligand structure information takes center stage, and only ligand information is used to derive its multitude of chemical and biological properties. On the other hand, the structure-based approach utilizes the structure of the receptor to identify shape-complementary ligands with optimal interactions. Given a validated disease target with a known crystal structure, the structure-based approach can be utilized to discover ligands that bind to the receptor of interest. On the other hand, the ligand-based approach is useful when the target structure is unknown or when the crystal structure of the target is difficult to obtain. In some cases, the system-based drug discovery approach combines both ligand-based and structure-based approaches to facilitate the drug design and discovery process [3].

2. Ligand-based approaches to drug design

2.1. Introduction

Ligand-based drug design, also known as "knowledge-based" drug design, extracts essential chemical features from drugs to construct a learning model to predict drug properties. It has been proposed that a ligand structure contains all the necessary information to accurately infer its mechanism of action. Several chemical descriptors such as molecular weight and lipophilicity are predictors of important pharmacodynamic and pharmacokinetic properties of ligands. A well-known example is Lipinski's rule of five, which describes a set of chemical properties and rules that can be used to differentiate drugs from nondrug molecules [4]. Likewise, the chemical similarity principle has been widely applied in similarity-based drug design. The chemical similarity principle assumes that if two molecules share similar structures, then they will likely have similar biological properties [5]. This concept is the underlying principle of modern chemical database search techniques used to identify similar compounds with improved bioactivities.

2.2. Outline of ligand-based drug design

The chemical similarity search is an established approach for ligand-based drug discovery. Given a compound with known biological properties, it is possible for a drug designer to identify similar compounds with improved biological properties. To compare two ligand structures, it is essential to develop appropriate data structure representations for chemical structure comparisons. Mathematically, the chemical structure can be represented by graphs where atoms represent vertices and edges represent chemical bonds [6]. Several chemoinformatics algorithm can be used to extract essential characteristic from the chemical graph such as the number of vertices, the number of bonds, path, connectivity, and others. These properties then become chemical features that can be used in feature engineering using complex machine learning techniques for similarity comparison.

The most direct chemical search approach is the nearest neighbor where the chemical feature of a ligand, also known as a "chemical fingerprint," is used to search a compound database to identify the most similar compounds using a predefined distance measure [7]. The most commonly used chemical fingerprints include path-based and substructure-based fingerprints. Using path-based fingerprints, such as Daylight fingerprints or Obabel FP2 fingerprints, potential paths at different bond lengths in a molecular graph of a molecule are used as features for the similarity comparison. On the other hand, substructure-based fingerprints such as MACCS keys use predefined substructures and characterize each molecule based on the presence or absence of a particular substructure using a binary array. Overall, path-based fingerprints offer higher search specificity due to the unique path dependency of the molecular graph. However, substructure-based fingerprints can be used to identify scaffold hopping ligands since the fingerprints do not impose requirements on the connectivity of the scaffolds and functional groups [8]. To quantify the chemical similarity between two fingerprints, several distance metrics have been proposed. The most commonly used chemical similarity metric is the Tanimoto index, which computes the shared feature bits between two fingerprints in the range of 0–1. Although there is no predefined threshold to define the similarity level, a value of 0.7–0.8 has been commonly adapted in many chemical similarity search programs.

2.3. Ligand-based drug design (LBDD) process

The similarity-based drug design process is as follows (Figure 1):

- 1. The target molecule is used as a query for the chemical search.
- 2. Similar ligands with similar biological properties are identified.
- 3. Original ligands are modified to suggest new molecules with improved activities.

The advantages of LBDD are as follows:

- 1. Does not necessitate receptor structures.
- 2. Low computational intensity and fast database searching.
- 3. Allows for large-scale similarity drug design and target prediction.



Figure 1. Ligand-based drug design (LBDD) process.

2.4. Target prediction of drugs

While similarity-based methods are prevalently applied in modern drug discovery programs as an efficient way to transition a hit to a lead, the molecular mechanism of the drug is often unknown, and the adverse reaction cannot be predicted. Consequently, drug-target prediction becomes an important follow-up step. Drug-target prediction can be classified as ligand-based or structure-based methods [9]. In ligand-based target prediction, the molecular target of a drug can be inferred from the target-annotated ligand sharing the highest chemical similarity. Chemical bioactivity databases such as ChEMBL, PubChem and DrugBank, and BindingDB have been developed for this application [10]. However, one major limitation of ligand-based target prediction is that there is no natural cutoff for chemical similarity that clearly defines biological similarity, also known as bioactivity cliffs. Approaches such as similarity ensemble approach (SEA) aim to remedy this by calculating similarity values against a random background using an algorithm similar to BLAST [11]. On the other hand, structure-based target prediction methods identify molecular targets based on the structure of the receptor binding sites. For example, panel docking is a common structure-based approach to identify the most probable target based on the docking score. Alternatively, binding site similarity methods that compare the receptor environment of the target ligand to a database of receptor pockets have also proven to be an effective target prediction approach [12].

More recently, network poly-pharmacology has been proposed as a more comprehensive approach to analyze drug-target interactions. Network pharmacology goes beyond the one drug one target hypothesis to multiple drugs multiple targets hypothesis [13]. The goal of this new paradigm is not only to accurately identify on-targets but also other off-targets. One such approach is the drug-target network, which utilizes a bipartite network to analyze complex drug gene interactions. Alternatively, drug-drug networks or chemical similarity networks have also been proposed [8]. The chemical similarity network clusters drugs based on their structure similarity. This approach can be applied for large-scale compound analysis by clustering diverse chemical structures into distinct scaffolds known as chemotypes. Consequently, each chemotype can be correlated with specific molecular targets. Using a consensus statistics scheme similar to that used for functional prediction in protein-protein interaction networks, chemical similarity networks have proven useful for target identification from chemical screening campaigns. In addition, structural poly-pharmacology has also gained substantial attention due to the possibility of correlating structural variations to clinical side effects. One example is CSNAP3D, which uses 3D ligand structure similarity to identify simplified scaffold hopping compounds of complex natural products to suggest new drugs with improved pharmacokinetic properties [1].

2.5. Side effect prediction of drugs

Since severe clinical side effects have contributed to drug failure in the late stages of clinical trials, side effect prediction will need to become an integrated part of the drug design process. Knowing the binding affinity of a drug to an array of proteins makes it possible to predict side effects using several statistical methods such as canonical component analysis (CCA), which identify a set of parameters that optimize the correlation between drug binding features and side effect features [14]. In addition, side effect predictions based on chemoinformatics analysis of the compound structures have also been developed. Many of these approaches provide an accurate prediction of the drug side effects and have been applied in the early stages of clinical drug development.

3. Structure-based approaches to drug design

Modern medicinal chemistry methods and molecular modeling have been employed as powerful tools for the study of structure-activity relationships (SAR) [15]. Structure-based drug design is a drug discovery approach by which synthetic compounds are designed from detailed structural knowledge of the active sites of protein targets associated with particular diseases. This field has involved the integrated application of traditional biology and medicinal chemistry along with advances in biomolecular spectroscopic methods such as X-ray crystallography, nuclear magnetic resonance (NMR), combinatorial chemistry, computer modeling of molecular structure, and protein biophysical chemistry, to focus on the three-dimensional molecular structure and active site characterization of the proteins that control cellular biology. Structure-based drug design is an improvement over traditional drug screening techniques. By identifying the target protein in advance and by discovering the chemical and molecular structure of the protein, it is possible to design a more optimal drug to interact with the protein.

3.1. Basic concept of structure-based drug design

Enzymes are a subset of receptor-like proteins that are directly responsible for catalyzing the biochemical reactions that sustain life. For example, digestive enzymes act to break down the nutrients of our diet. DNA polymerase and related enzymes are crucial for cell division and replication. Enzymes are genetically programmed to be specific for their appropriate molecular targets. Any errors could have grave consequences. Enzymes ensure the specificity of their targets by forming a molecular environment that excludes interactions with inappropriate molecules. The analogy most often mentioned is that of a lock and key. The enzyme is a

molecular lock, which contains a keyhole that exhibits a very specific and consistent size and shape. This molecular keyhole is termed the active site of the enzyme and allows interaction with only the appropriate molecular targets. Just as a typical lock is much bigger than the keyhole, the receptor is usually much larger than the active site. The receptor, as specified by our DNA, is a folded protein whose major purpose is to form and maintain the size and shape of the active site.

The most important concept in drug design is to understand the methods by which the active site of the receptor selectivity restricts the binding of inappropriate structures. Any potential molecules that can bind to a receptor are called ligands. In order for a ligand to bind, it must contain a specific combination of atoms that present the correct size, shape, and charge composition needed to bind and interact with the receptor. In brief, the ligand must possess the molecular key that binds the receptor lock.

Computer-aided drug design has played an important role in drug discovery and drug development and has become an indispensable tool in the drug industry. For the purposes of discovering and optimizing biologically active compounds, various types of computer-aided drug design software and resources have been used by computational medicinal chemists. Unsurprisingly, many chemical compounds were discovered and optimized by computer-aided drug design methodologies and have reached the late stages of clinical trials.

3.2. Outline of structure-based drug design

Structure-based drug design is a cyclic process which consists of stepwise procedures (**Figure 2**). It begins with a known target structure, and then, *in silico* studies are conducted in order to identify potential ligands. These molecular modeling procedures are followed by the synthesis of the most promising compounds [16]. Next, using diverse experimental platforms, biological properties such as potency, affinity, and efficacy are evaluated [17]. In the end, given that active compounds are identified, the three-dimensional structure of the ligand-receptor complex can be solved. The available structure allows the visualization of the intermolecular features that support the process of molecular recognition. Structural descriptions of ligand-receptor complexes are useful for the investigation of binding conformations, characterization of key intermolecular interactions, characterization of unknown binding sites, mechanistic studies, and the elucidation of ligand-induced conformational changes [18].

3.3. Process of structure-based drug design (SBDD)

The structure-based drug design process is as follows:

- 1. An enzyme that is important in a particular pathological condition is chosen.
- **2.** The three-dimensional structure of the active site of the enzyme is determined, often by X-ray crystallography.
- **3.** A chemical is prepared to fit the active site of the enzyme, which can alter the properties of the enzyme, that is, inactivate the enzyme.

The advantages of structure-based drug design are as follows:

- 1. Useful results are obtained faster than by traditional drug design methods.
- 2. The process is less expensive than other drug design methods.
- **3.** The compounds are more specific for the active site and potentially less toxic than compounds prepared by other approaches.



Figure 2. Structure-based drug design (SBDD) process.

3.4. Synthesis of lead compound

The initial drug design phase is followed by the synthesis of the lead compound, quantitative measurements of its ability to interact with the target protein, and X-ray crystallographic analysis of the compound-target complex. This analysis reveals important, empirical information on how the compound actually binds to the target, and the nature and extent of changes induced in the target by the binding. These data, in turn, suggest ways to refine the lead compound to improve its binding to the target protein. The refined lead compound is then synthesized and complexed with the target, and further refined in a reiterative process. If lead compounds are available from other studies, such as screening of combinatorial libraries, these compounds may serve as starting points for this optimization cycle using structure-based drug design.

Once a sufficiently potent compound has been designed and optimized, its activity is evaluated in a biological system to establish the compound's ability to function in a physiological environment. If the compound fails at any stage of the biological evaluation, the design team reviews the structural model and uses crystallography to adjust structural features of the compound to overcome the difficulty. This process continues until a designed compound exhibits the desired properties. The compound is then evaluated in an experimental disease model. If the compound fails, the reasons for failure (e.g., adverse metabolism, plasma binding, distribution, etc.) are determined, and again new modified compounds are designed to overcome the deficiencies without interfering with their ability to interact with the active site of the target protein. The experimental drug is then ready for conventional drug development (e.g., studies in safety assessment, formulation, clinical trials, etc.). This reiterative analysis and compound modification are possible because of the structural data obtained by X-ray crystallographic analysis at each stage. This capability renders structure-based drug design, a powerful tool for rapid and efficient development of drugs that are highly specific for particular protein target sites.

3.5. Docking methodologies

Several docking procedures exist in the literature, from the use of interactive graphics to manipulate the position of the ligand to completely automated procedures, which are becoming increasingly powerful to screen databases of molecules. Many docking algorithms follow a similar pattern. Usually, the first stage is to represent the molecules by their solvent-accessible surfaces. Beginning with a number of different relative orientations, the two molecules can then be brought together. More sophisticated methods carry out their moves by using a Monte Carlo algorithm to direct both rotations and translations. Rapid convergence to a minimum can be achieved by gradually cooling the simulation temperature as the molecule appears to be descending into a potential well.



Figure 3. The molecular docking methodology. (A) Prepare the ligand structure, (B) Prepare the receptor structure, (C) Dock the ligand into the receptor surface with multiple potential conformations. (D) The most likely binding mode was identified based on intermolecular interaction between the ligand and the receptor surface. The protein backbone is represented as a cartoon. The ligand and active site residue are shown in stick representation.

Molecular docking is one of the most frequently used methods in structure-based drug design because of its ability to predict, with a substantial degree of accuracy, the conformation of small-molecule ligands within the appropriate target binding site (**Figure 3**) [19]. Following the development of the first algorithms in the 1980s, molecular docking became an essential tool in drug discovery [20]. Highly intensive investigations involving crucial molecular events, including ligand-binding modes and the corresponding intermolecular interactions which stabilize the ligand-receptor complex, could be conveniently performed. Furthermore, molecular docking algorithms execute quantitative predictions of binding energetics, given docked compounds, based on the binding affinity of ligand-receptor complexes [21].

3.6. Computer simulation in drug design

The development of a new drug starts with the design of suitable candidate compounds socalled "ligands" that are selected according to observations about how these compounds are recognized by the target protein and how they bind to it. It is important to know that proteins have dynamic properties: they change their shape in much the way as a machine tool needs to in order to fulfill its function. Now, it is being realized how important it is to have techniques available for studying protein dynamics. Performing experiments is not only expensive but also very time-consuming and, moreover, cannot answer all relevant questions. An alternative is computer-aided simulation of the dynamics of molecules [molecular dynamics (MD) simulations], which is becoming increasingly important to identify the molecular properties that are important and to determine the detailed molecular interactions that are critical for binding.

In some instances, high-performance computing (HPC) is required to cross the threshold where MD simulation becomes a valuable tool for industry. However, in most pharmaceutical companies, HPC is something very new, and supercomputers are simply not available to industrial researchers. With the arrival of affordable high-performance multiprocessor machines and corresponding developments of parallel software, it now becomes possible for industrial researchers to undertake more realistic calculations that were previously out of reach. Scientists at Novo Nordisk, a large Danish pharmaceutical company, are convinced that this new capability will dramatically change the acceptance of MD simulation as a tool in the design of new ligands. During Europort-D, they, for the first time, studied the dynamics of the complex molecular interactions critical for recognition of ligands by their target proteins.

4. Statistical tool for drug design

4.1. Introduction to SPASS for pharmaceutical research

As a sophisticated statistical software, Statistical Package for Social Science (SPSS) can help researchers and users realize their complex statistical tests with result interpretation and the access to data figures, tables, and graphs, which can be quickly and easily displayed in the output view. Even though additional training is usually required before users can maximize its features and the graphing feature is not as simple as an excel spreadsheet, SPSS is still being

used by more and more researchers from other fields for applications specific to their research areas, for example, the application of SPSS in clinical trials, with power and sample size calculation functions, comprehensive statistical tests and graphical tools, such as predictive data trends, forecasting and report generation, and the analysis of complex drug assays, etc. [22].

SPSS is being used majorly in pharmaceutical and medical research, where it can provide techniques for the design, implementation, and development of drug campaigns that are important to pharmaceutical research organizations and also provides data mining solutions based on their existing data on quality control and manufacturing, regulatory safety testing, clinical trials, data mining in drug discovery, compliance, and validation [23]. As many pharmaceutical research organizations have experienced that even small improvements in research or processing affect the efficiency and success of the project, SPSS provides a sophisticated analytical platforms to decrease development costs and unnecessary preclinical and clinical trials by integrating with existing data sources.

Overall, the tools in SPSS provide an efficient mechanism to automatically validate the routine analysis reports. In other words, SPSS empowers researchers with tools, which might bring the rapid and solid return on investments in the near future.

4.2. Outlines of the SPASS package

SPSS is a Windows-based program, which can be used to perform data entry and data analysis by tables and graphs (**Figure 4**). Because SPSS is capable of dealing with huge amounts of data with different modules, it is commonly used in the business world and in the Social Sciences [24]. Familiarity with this software will be useful in the field of drug research where big data,



Figure 4. SPSS package in pharmaceutical data analytics application.

such as clinical trials data ranging from simple bioassays and dose-response experiments to long-term survival, and carcinogenicity studies need to be integrated. Unlike Excel, SPSS software can support screen transfers between the data entry view and the output view, which display the results and summarize the data.

4.3. Features of SPSS package

Whether its small and medium enterprises (SME) or large-scale enterprises (LSE), the application of SPSS can provide data accessing in rational data sets, which allows accessing inflexible resources and practices and real-time processing, and mapping of the database [25]. Although other commercial softwares, such as SAS, MATLAB, and others, can also perform such functions, SPSS is a more sophisticated application in statistics analysis. SPSS conducts statistical analysis from basic data, descriptive statistics, such as average and prevalence, and advanced inferential statistics, such as t test, regression model, one-way and factorial analysis of variance (ANOVA), one-way and factorial analysis of covariance (ANCOVA), one-way and factorial multivariate analysis of variance (MANOVA), one-way and factorial multivariate analysis of covariance (MANCOVA), factor analysis, path analysis, and logistic regression [26].

Normally, researchers use SPSS as a tool to collect and analyze data. The data entry screen in SPSS looks like an excel spreadsheet, and users can input variables and quantitative data and save the file as a data file [27]. After data are collected and entered into the data sheet in SPSS, users can also create an output file from the data they used. Then, the users can edit or organize the data in SPSS to check the running results after they choose the module they want. For example, users can check out the frequency distributions of the data to see whether the data set is normally distributed. Furthermore, the researchers can download the tables or graphs directly from the data output view. Because SPSS has the statistical tests built-in to the program, users do not need to do any math calculation or equations to get the results.

5. Conclusion and future direction

Systems pharmacology represents a new paradigm in drug research where ligand and protein information are combined to produce new methods for drug discovery and design. Traditionally, ligand-based approaches utilize information contained within the chemical structure to predict the biological properties of the drug. In particular, chemical similarity database searches can be used to identify molecules with improved activity. However, to be able to design compounds with satisfactory safety profiles, the drug targets will also need to be determined. Target prediction can be performed using ligand or structure data. Both of these approaches require a bioactivity database with pre-characterized activities or functions. If the target structure of the ligand is known, then the structure-based drug discovery approach can be applied. Computer-aided drug design techniques such as molecular docking and molecular dynamic simulation are capable of accurately predicting the binding sites of ligand, docking pose, and binding affinity based on the geometry of the receptor surface. The prediction can then be validated using *in vitro* biochemical assays and X-ray crystallography to validate ligand

binding. Thus, a cyclic drug design process is then continued until the strongest binder is found.

Currently, drug discovery has been mainly focused on in vitro optimization. One future direction in system-based drug discovery will necessarily shift to ADMET prediction to assess drug performance *in vivo*. Although several ADMET properties such as drug-likeness can be predicted by simple rules, a more detailed classification of drug properties can be achieved by more advanced machine learning techniques. Likewise, a holistic drug design process that simultaneously optimizes the drug properties based on *in vitro* and *in vivo* data will also hold promise to optimize drug safety and minimize adverse reactions. Consequently, integration of multilevel data from structure, tissue, and the whole organism will be required to achieve a more accurate prediction. In conclusion, we have presented the essential quantitative methods in system-based drug discovery, and the approaches presented here will stimulate further efforts in the progress of drug discovery and design to engineer safer and more effective drugs.

Author details

Yu-Chen Lo1*, Ren Gui2, Hiroshi Honda2 and Jorge Z. Torres3

*Address all correspondence to: bennylo@stanford.edu

1 Department of Bioengineering, Stanford University, The United States

2 Department of Bioengineering, Northwestern Polytechnic University, The United States

3 Department of Chemistry and Biochemistry, University of California, Los Angeles, The United States

References

- Lo YC, Senese S, Damoiseaux R, Torres JZ. 3D chemical similarity networks for structure-based target prediction and scaffold hopping. ACS Chemical Biology. 2006;11(8):2244–53. doi:10.1021/acschembio.6b00253. PMID: 27285961.
- [2] Barelier S, Sterling T, O'Meara MJ, Shoichet BK. The recognition of identical ligands by unrelated proteins. ACS Chemical Biology. 2015;10(12):2772–2784. doi:10.1021/ acschembio.5b00683. PMID: 26421501; PMCID: PMCPMC4684825.
- [3] Shoichet BK, Walters WP, Jiang H, Bajorath J. Advances in computational medicinal chemistry: A reflection on the evolution of the field and perspective going forward.
Journal of Medicinal Chemistry. 2016;59(9):4033–4034. doi:10.1021/acs.jmedchem. 6b00511. PMID: 27054949.

- [4] Lipinski CA. Rule of five in 2015 and beyond: Target and ligand structural limitations, ligand chemistry structure and drug discovery project decisions. Advanced Drug Delivery Reviews. 2016;101:34–41. doi:10.1016/j.addr.2016.04.029. PMID: 27154268.
- [5] Yan X, Liao C, Liu Z, Hagler AT, Gu Q, Xu J. Chemical structure similarity search for ligand-based virtual screening: Methods and computational resources. Current Drug Targets. 2015:17(14):1580–85. PMID: 26521773.
- [6] Le SQ, Ho TB, Phan TT. A novel graph-based similarity measure for 2D chemical structures. Genome Informatics. 2004;15(2):82–91. PMID: 15706494.
- [7] Raymond JW, Willett P. Effectiveness of graph-based and fingerprint-based similarity measures for virtual screening of 2D chemical structure databases. Journal of Computer-Aided Molecular Design. 2002;16(1):59–71. PMID: 12197666.
- [8] Lo YC, Senese S, Li CM, Hu Q, Huang Y, Damoiseaux R, et al. Large-scale chemical similarity networks for target profiling of compounds identified in cell-based chemical screens. PLoS Computational Biology. 2015;11(3):e1004153. doi:10.1371/journal.pcbi. 1004153. PMID: 25826798; PMCID: PMCPMC4380459.
- [9] Chen X, Yan CC, Zhang X, Zhang X, Dai F, Yin J, et al. Drug-target interaction prediction: Databases, web servers and computational models. Briefings in Bioinformatics. 2016;17(4):696–712. doi:10.1093/bib/bbv066. PMID: 26283676.
- [10] Davies M, Nowotka M, Papadatos G, Dedman N, Gaulton A, Atkinson F, et al. ChEMBL web services: Streamlining access to drug discovery data and utilities. Nucleic Acids Research. 2015;43(W1):W612–W620. doi:10.1093/nar/gkv352. PMID: 25883136; PMCID: PMCPMC4489243.
- [11] Keiser MJ, Roth BL, Armbruster BN, Ernsberger P, Irwin JJ, Shoichet BK. Relating protein pharmacology by ligand chemistry. Nature Biotechnology. 2007;25(2):197–206. doi:10.1038/nbt1284. PMID: 17287757.
- [12] Liu T, Altman RB. Using multiple microenvironments to find similar ligand-binding sites: Application to kinase inhibitor binding. PLoS Computational Biology. 2011;7(12):e1002326. doi:10.1371/journal.pcbi.1002326. PMID: 22219723; PMCID: PMCPMC3248393.
- [13] Westerhoff HV. Network-based pharmacology through systems biology. Drug Discovery Today: Technologies. 2015;15:15–16. doi:10.1016/j.ddtec.2015.05.001. PMID: 26464085.
- [14] Liu T, Altman RB. Relating essential proteins to drug side-effects using canonical component analysis: A structure-based approach. Journal of Chemical Information and

Modeling. 2015;55(7):1483–1494. doi:10.1021/acs.jcim.5b00030. PMID: 26121262; PMCID: PMCPMC4875781.

- [15] Hughes JP, Rees S, Kalindjian SB, Philpott KL. Principles of early drug discovery. British Journal of Pharmacology. 2011;162(6):1239–1249. doi:10.1111/j.1476-5381.2010.01127.x. PMID: 21091654; PMCID: PMCPMC3058157.
- [16] Wilson GL, Lill MA. Integrating structure-based and ligand-based approaches for computational drug design. Future Medicinal Chemistry. 2011;3(6):735–750. doi: 10.4155/fmc.11.18. PMID: 21554079.
- [17] Fang Y. Ligand-receptor interaction platforms and their applications for drug discovery. Expert Opinion on Drug Discovery. 2012;7(10):969–988. doi: 10.1517/17460441.2012.715631. PMID: 22860803.
- [18] Kahsai AW, Xiao K, Rajagopal S, Ahn S, Shukla AK, Sun J, et al. Multiple ligand-specific conformations of the beta2-adrenergic receptor. Nature Chemical Biology. 2011;7(10): 692–700. doi:10.1038/nchembio.634. PMID: 21857662; PMCID: PMCPMC3404607.
- [19] Meng XY, Zhang HX, Mezei M, Cui M. Molecular docking: A powerful approach for structure-based drug discovery. Current Computer-Aided Drug Design. 2011;7(2):146– 157. PMID: 21534921; PMCID: PMCPMC3151162.
- [20] Lopez-Vallejo F, Caulfield T, Martinez-Mayorga K, Giulianotti MA, Nefzi A, Houghten RA, et al. Integrating virtual screening and combinatorial chemistry for accelerated drug discovery. Combinatorial Chemistry and High Throughput Screening. 2011;14(6): 475–487. PMID: 21521151.
- [21] Huang SY, Zou X. Advances and challenges in protein-ligand docking. International Journal of Molecular Sciences. 2010;11(8):3016–3034. doi:10.3390/ijms11083016. PMID: 21152288; PMCID: PMCPMC2996748.
- [22] Williams F. Reasoning with Statistics: How to Read Quantitative Research. 4th ed. Fort Worth: Harcourt Brace Jovanovich College Publishers; 1992, Vol. xiii, 220 p.
- [23] Sprinthall RC. Basic Statistical Analysis. 6th ed. Boston: Allyn and Bacon; 2000, Vol. xiv, 626 p.
- [24] Norusis MJ, SPSS Inc. SPSS 8.0 Guide to Data Analysis. Upper Saddle River, N.J.: Prentice Hall; 1998, Vol. xviii, 563 p.
- [25] Harris MB. Basic Statistics for Behavioral Science Research. 2nd ed. Boston: Allyn and Bacon; 1998, Vol. xvi, 576 p.
- [26] Asher HB. Causal Modeling. 2nd ed. Beverly Hills: Sage Publications; 1983, 96 p.
- [27] George D, Mallery P. SPSS for Windows Step by Step: A Simple Guide and Reference 9.0 Update. 2nd ed. Boston: Allyn and Bacon; 2000, 357 p.

Stochastic Agent-Based Simulation of the Role of Labor in the Economy

Alexander Kindler, Natasa Golo and Sorin Solomon

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/64698

Abstract

We present a platform for stochastic agent-based simulation of the role of labor in the economy (SABLE). The platform facilitates heterogeneous agents: a large number of firms, workers, and banks are represented as individual agents. It allows the agents to have fixed or random reactions to the other agents and to pursue goals through decisions with bounded rationality. Its purpose is to clarify the feedback effects between the individuals as workers and consumers in a complex environment. SABLE is meant to be a quantitative precise tool to express classical, neoclassical and heterodox scenarios and to document their outcomes. In particular, questions regarding policies dealing with minimum wages, unemployment wages, hiring regulation, and the role of finance on unemployment can be studied in a new, interactive, and transparent way, inspired among others by the critical thoughts in.

Keywords: unemployment, complex systems, agent-based model, stochastic process, econophysics

1. Introduction

Unemployment is one of the main concerns of any society, because of its effect on economic, social, and political stability of countries. One of its aspects that is relatively well known and explained is job search. The main approach to understand job search is known as search and matching modeling [1, 2]. Search and matching models represent a stochastic process of joining of unemployed individuals and firms. In the existing literature, this joining is called the aggregate matching function [1], which proved successful at predicting quantitative and qualitative features of employment. The assumption of aggregate matching functions in labor



© 2016 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. markets has tested using a network configuration model for directed multigraphs in [3] and the results suggest the need for theoretical frameworks that take into account the structure of labor market frictions.

Also, the models of aggregate matching are often based on a number of assumptions that are specific to a certain case and related with specific market conditions. For example, in [4], the assumption is that newly created jobs are the most profitable in the market. Such an assumption is less valid if job creation is by new start-ups. Our approach, stochastic agent-based simulation of the role of labor in an economic system (SABLE) is unique in the sense that it allows for testing of any such assumption with relative ease and high efficiency.

Our agent-based stochastic macroeconomic model also includes the interaction between credit and labor markets. Under the assumption that the investment is dependent on the credit, and that the access to credit is dependent of firms performance, it is expected that the credit market plays a very important role in the dynamics of the labor markets. In this respect, particularly the new firms can be credit constrained, as in [5].

To describe the underlying concept of stochastic agent-based simulations, and its design concepts specific to our method of implementation, we wrote the paper along the general guidelines of [6]. We conclude with several examples of obtained output and their short discussion.

1.1. General approach

Economic heterogeneous interactive agents are on the workbench of an ever-increasing community for already 20 years (at least the age of the Workshop with the same name WEHIA). In parallel with the finance-focused platforms of the LLS [7] and Santa Fe type [8], they have been used to create stylized facts in business cycles, power laws of wealth distribution, fractal time series, etc. as shown in [9–11].

In spite of the limitations of agent-based simulations, they are recognized as a new mathematical tool to tackle concrete problems and connect them easily with the data.

The agent-based platform SABLE offers a standard environment for implementing and comparing specific mechanisms focused on problems in the labor market suggested by the empirical observations or the theoretical discourse within the academic or practitioner economics community.

Moreover, SABLE can be perceived as a neutral scaffold to express in a quantitative precise way various scenarios of current or generic economic situations and processes. In particular, it allows illustrating the possible outcomes of implementing the solutions proposed by various neoclassical and heterodox schools to current economic debates.

1.2. Agents

The basic groups in the current version of the SABLE platform are workers and firms. Each firm and each worker is represented individually, having a bank account, income, and some

other specifications. Further entities such as banks, government, and external markets are implemented in the current version as single/representative agents.

The main interactions between real sector firms are supplier-client ones. As such, the firms are related through a "transactions matrix," which is a microeconomic similar of the Leontief matrix. It is an antisymmetric matrix representing money transfer between firms (in return to products/services).

The speed by which this matrix changes has a crucial effect on the dynamics and resilience of the economy. Too fast changing of the matrix makes the economy unstable, too slow—it makes it vulnerable to external (e.g., financial sector) change, as seen in [12].

The next important interaction is between firms and workers. It includes two aspects:

- Workers as employees: receiving money from the employer firm;
- Workers as customers spending their income on buying products.

As such, the workers are crucial in closing the flow/causal chain in the economic system.

Among other things, the double role of the workers allows for very significant feedback loops that dominate the SABLE system: these feedback loops may stabilize or destabilize the system and are a key to understanding how capitalist economy works.

1.3. Activities

Workers, firms, the bank, and the government interact through a set of routines that include the following:

- Workers are employed in firms or unemployed. Correspondingly,
 - they get a salary from the firm where they work in return for the labor they produce for the firm, or
 - they get an unemployment wage from the government (for "free").
- Workers spend their income in the market, thus returning money back to the firms.
- Firms pay workers in exchange for the labor they get from them.
- Besides, they do business with other firms, paying for services and products.
- Firms draw income from the sales to their market.
- Firms also compete between themselves over their share of the whole market.
- Firms get loans from banks and repay them back later.
- Taxes are paid by firms and workers to the government.

These relations are summarized in Figure 1.



Figure 1. The major activities that are modeled in the basic version of SABLE.

2. Design concepts

In this section, we describe the previously introduced general features in more detail and we introduce the specifications of the code underlying the platform.

2.1. Time, activity patterns, and activation schemes

The basic sequence of events consists of several operations performed by agents as outlined in **Figure 1**:

- Create new firms.
- Initialize/update related parameters (e.g., the "transaction matrix (TM)").
- Update income of each firm according to its market share.
- Update the loan portfolios of the firms.
- Update the bankruptcy status of firms.
- Rescale the relative market shares of the remaining firms accordingly.
- Enforce the limits on the effective market share on each firm according to the maximal amount of products and services that it can receive from suppliers.
- Introduce new workers and initialize their respective parameters.
- Firms hire or fire workers, according to labor requirements and the salaries level.

- Enforce the limits on the effective market share on each firm according to its current available labor.
- Update workers' wages: workers get raises or cuts in their wages (according to their employment history and status).
- Pay salaries (and unemployment wages).
- Workers spend a certain fraction of their income in the market.
- Their aggregate spending generates the income for firms at the next time step.
- Firms compete between themselves over market share potential (not their effective hold on the market, which is limited by required labor and services of other firms).
- Update interest rate.
- Other events may take place, such as government intervention at different stages of the above dynamics, according to the desired effect.

According to the above list, one iteration of the platform corresponds to roughly a month of "real-world" economy progress, since salaries and loan returns are paid monthly.

2.2. Interaction protocols and information flows

Generally, interactions between agents are on the individuals' level, meaning that different decisions and actions are made by each firm and by each worker (on some occasions, the whole pool of firms or workers acts in coordination; e.g., collective labor contracts):

- Pairs of firms may agree to particular transactions.
- Loans are decided by pairs of firms or banks (according to a set of rules which takes into account the firm's features, the interest rate, etc.).
- The hiring and firing of workers is decided by the firms that employ them.
- In the present implementation, workers do not resign of their own volition.
- Initial wages are determined by labor initial capability (a given parameter).
- Further wages are fixed by previous wages and seniority.

In the present implementation, the end consumer market equals the total labor income (after taxes and interest).

2.3. Levels of randomness

SABLE has a core of random processes, which govern the dynamics of the entire economy as follows:

- The initial wealth of the new companies is random.
- For each firm, the relevant elements of the transaction matrix may evolve through a random process.

- Loans are granted with a certain probability depending on firm income, firm wealth, interest rate, etc.
- Hiring, firing, and offered wages have a random element too.
- The competition between companies for market shares is decided in a random way (with bias toward the larger companies).

3. Functional specifications

In this section, we describe in detail the implementation of the features discussed above, specifically agents and their interactions. We abstain from giving a full parametric description of each entity since that would flood the reader with a multitude of minor details, and instead we will give a general description, supported with mathematical formulation when essential only.

We will not list the different parameters and their uses, but rather give a more detailed description of the mechanisms underlying the interaction protocols and activities listed earlier.

In **Table 1**, we list these actions, interactions, and calculations.

Name	Туре	Description
Determines whether employed workers get raises, or	Action	Get raise and drop wages
whether unemployed workers lower their wage.		
Updates the interest rate.	Action	Update interest
Generates new firms and workers.	Action	Create new firms and workers
Determines whether a firm gets a loan. Size is a fraction	Interaction	Get loan
of the firm's cash.		
Determines if and how much labor a firm hires or fires.	Interaction	Hire and fire
Varies the amount of money transferred between firms.	Interaction	Vary transaction matrix
Determine which firms compete over market shares potential,	Interaction	Competition
and determine the outcome of the competition.		
Calculate how much of the market share potential	Calculation	Effective market share
a firm effectively utilizes.		
Calculate how much income firms get.	Calculation	Firm income

Table 1. Actions, interactions, and important calculations.

Now, we briefly describe the mathematics behind each action, interaction, and calculation as applied in our model.

Get raise and drop wages: An employed individual has a probability to increase his wage each step he is employed by a certain percentage of his current wage (get a raise). Vice versa, an

unemployed individual has a probability to decrease the wage for which he is willing to work each step that he is unemployed, in order to help himself get a job that is better paying than the unemployment wage.

Update interest: Interest rates may also rise and fall with respect to demand in loans. We apply a single interest rate among all firms in this version of SABLE. Every given time period, the proportion of the total debt is used to rescale the interest rate: if debt is growing, then interest rates will fall and vice versa.

Create new firms and workers: Both growth effects are assumed to be Poisson processes, with a mean related to the initial amount of workers and firms.

Get loan: Table 2 lists a means for determining the likelihood of a firm getting a loan.

	Low income	High income
Low cash Low debt	Likely	Very likely
High cash Low debt	Unlikely	Likely
Low cash High debt	Unlikely	Likely
High cash High debt	Very unlikely	Unlikely

Table 2. Likelihood of a loan.

This could be conveniently formulated in probability terms, P_{cash} and P_{income} , which represent the probability of a firm to get a loan regarding its cash and income, respectively.

Since we would like a probability function that rises with income and falls with increase in cash, we propose as follows:

$$P_{\text{cash}} = \left\{ \tan^{-1} \left[A - \cosh \right] + \pi / 2 \right\} / \pi$$
 (1a)

and similarly

$$P_{\text{income}} = \left\{ \tan^{-1} \left[B - \text{income} \right] + \pi / 2 \right\} / \pi$$
 (1b)

where *A* and *B* are parameters. Since P_{cash} and P_{income} should be independent, in general, we begin constructing the probability for a loan with

$$P_{\text{loan}}^{"} = 1 - \left(1 - P_{\text{cash}}\right) \cdot \left(1 - P_{\text{income}}\right)$$
⁽²⁾

Next, we must account for the interest rate—when it is high, all firms are less likely to loan money, and so we multiply Eq. (2) by a factor of $(1-conf \cdot I)$, where conf is a "confidence"

parameter and *I* is the interest rate. Certainly, $conf \cdot I < 1$, and the smaller conf is the less impact interest rate has on the probability of getting loans. This results in the following step in the loan probability function:

$$P'_{\text{loan}} = 1 - (1 - P_{\text{cash}}) \cdot (1 - P_{\text{income}}) \cdot (1 - \text{conf} \cdot I)$$
(3)

The last argument to be added is the current debt, since the more indebted the firm, the less likely it is for a bank to increase that debt. Plainly, we multiply Eq. (3) by $[C/(C + \text{debt})]^D$ where *C* and *D* are parameters. This gives us the definitive loan probability function:

$$P_{\text{loan}} = 1 - (1 - P_{\text{cash}}) \cdot (1 - P_{\text{income}}) \cdot (1 - \text{conf} \cdot I) \cdot \left[C / (C + \text{debt})\right]^{D}$$
(4)

This formula allows for achieving the desired control over the financing mechanism we have created (by scaling the parameters), through a random process.

Hire and fire: Firms need labor in order to utilize their market share, and do business with other firms. On the other hand, the salaries they pay are a burden, therefore resulting in a search for optimality. We assume that firms attempt to increase or decrease a percentage of currently paid salaries for each step, since the salaries they pay are the only objective measure firms usually have of their labor effectiveness. This too is a random process, where the probability of hiring, firing, or doing neither is controlled through a similar mechanism as in Eqs. (1a) and (1b):

$$P_{\text{hire}} = 1 - \tan^{-1} \left(\frac{\text{Salaries}_i}{E_i} \right) \cdot 2 / \pi$$
(5)

where E_i is a firm-specific parameter. Firing will happen in a similar fashion, leaving a probability for neither effect occurring as well.

Vary transaction matrix: We remind that the transaction matrix represents the trade of products and services between firms. To account for changes in firm behavior, we assume that the absolute values of the transaction matrix vary over time with some probability. However, we maintain the client-supplier relations fixed.

Competition: We model competition between firms through exchange in market shares, MS. Currently our model permits only one-on-one competition between firms.

The larger the market share of the firm, the less likely it is to lose to a smaller firm. So, if we have $MS_i > MS_j$ as respective market shares, the probability of shares going from firm *j* to firm *i* is as follows:

Stochastic Agent-Based Simulation of the Role of Labor in the Economy 107 http://dx.doi.org/10.5772/64698

$$P_{j \to i} = 1 - 0.5 \cdot \exp\left(-\left|\mathrm{MS}_{i} - \mathrm{MS}_{j}\right| / T\right)$$
(6)

where *T* is a coefficient. The amount of shares traded is a function of the difference in respective market shares, $\delta = |MS_i - MS_j|$. Certainly, the total shares of a smaller firm may be traded this way, leaving it totally out of the market. Not all firms compete at any given time step, and the amount of competing firm pairs is a parameter of the model. New firms start without market shares, but to compensate, they always compete when they are created.

Effective market share: If a firm has a certain MS_i, then it could only fully exploit a percentage of it due to not having enough products and services traded from other firms, or enough labor. Thus, all firms reach their potential market shares only asymptotically. Here, we implement the feedback between the transferring of money from firms to its peers and to its workers on its effectiveness to draw income from sales. The following equation shows the effective market share of a firm:

$$MS_{i}^{eff} = MS_{i} \cdot \tan^{-1} \left(\frac{\text{Employed Labor}_{i}}{E_{i}} \right) \cdot \tan^{-1} \left(\frac{\text{Products Services}_{i}}{F_{i}} \right) \cdot \left(\frac{2}{\pi} \right)^{2}$$
(7)

The measures of Employed Labor_{*i*} and Products Services_{*i*} are just as they appear — the amount of them a given firm has paid for. E_i and F_i are parameters of efficiency. We should stress here that firms hire and fire labor according to the salaries paid, and not according to the labor procured from each worker, but what really counts on efficiently utilizing market shares is the labor itself.

Firm income: Firms earn (and spend) money through a total of five features, best shown through an equation:

$$Income_{i} = (Market \cdot MS_{i}^{eff} - Salaries_{i} + \Sigma_{i}TM_{ii} - Loan Returns_{i}) \cdot (1 - Tax)$$
(8)

The Market represents the entire flux of money in the economy, either from consumer spending (again, the feedback) or from exogenous sources, such as external markets or government spending. This way, each firm utilizes a certain fraction of the entire economic input. Salaries are straightforward such as the loan returns and the tax reduction. The sum on the transaction matrix represents the trade balance of the firm.

This ends the description of the core features of SABLE.

4. Results

First, let us show the number of total and active firms at each time step. We see that at first the number of active firms will drop to a semi-stable (or slowly varying) amount, since the

initialization is unstable. This result is valid when counting all firms, and when counting only firms that have been active at least several time steps Results are plotted after the initialization of the system is complete (about 50 time steps) (**Figure 2**):



Figure 2. Top: The active number of firms (black asterisks), measured each 12 steps. Bottom: The change in bankruptcies (blue circles) and the added new firms (red pluses) between each 12 steps.

We plot the results on a 12-step interval to indicate yearly changes.

This result is reasonable enough: if economy is growing, it does not necessarily mean that every firm succeeds, but that the overall number of firms increases. If economy is receding, then new firms may still emerge though their chances of success are low, since even established firms are closing.

Moreover, almost any realistic set of parameters produces more or less the same variation in the number of active firms, while sometimes that number grows or recedes. This is a very robust product of SABLE.

We see that changes in the amount of firms vary mostly in the range between $\pm 5\%$ of the total amount of active firms at that step, meaning that these changes are rather stable: the median is 1.5%, the 80 percentile is 3% (absolute values of change).

For different parameter sets, economy can either grow or shrink, but changes in the amount of active firms remain similar to those depicted above.

We sort the firms according to their capital (cash), earnings (the positive figures from Eq. (8)), the relative amount of employees, and employed labor, at the final step of the simulated run. In order to study that distribution, we plot the results on a log-log scale (**Figure 3**).



Figure 3. Left: Money distribution in between active firms at the end of the simulation run, on a log-log scale. Middle: Earnings distribution in between active firms at the end of the simulation run, on a log-log scale. Right: Employees (blue dots) and labor (red circles) distribution in between active firms at the end of the simulation run, on a log-log scale.

We see that the distribution of the above variables roughly satisfies a power-law distribution. That is reasonable since the large capital accumulation is unavoidable for large firms: our model does not include either shareholders or a financial market where the cash could have been reinvested (except in the expansion of production). The dependence in the lower part might be characterized by another power law: the splitting of the distribution into two parts is expected [13] but largely exaggerated: most firms are indeed small (in all aspects) and only a few firms should be able to have large funds, income, and labor.

Also, we see that the relative amount of employees (blue dots) and employed labor (red circles) on the right-most plot are strongly correlated in this example.

It is a reassuring feature of SABLE, since this power law in capital accumulation is expected. Similar power-law dependencies in the firms' financial indicators have been often empirically validated by various authors [14–16] and were spontaneously achieved.

We inspect unemployment rates, and see that it is in the rational vicinity of up to 10% most of the time. There are, however, crises (besides the initialization effects), where unemployment soars, but then returns to low values (**Figure 4**).



Figure 4. The total labor available (labor market—black line), total salaries paid (blue line), and unemployment ratio (green line) at each step, cropping the initialization phase.

The "peaks" in unemployment result from two effects, which may overlap: a wave of bankruptcies causing several firms to close and fire their workers during the same step, or the closing of major firms.

We see that these crises are highly correlated with the dips in aggregate salaries as expected.

We have truncated the initialization steps, since the unemployment is extremely high and, respectively, very confusing when visualized.

Although it may seem that at some intervals unemployment is going down to zero (when not discussing the peaks), it is in fact just a low percentage: firms constantly go bankrupt and lose their workers, and firms may sometimes fire workers due to financial problems. However, when small firms go bankrupt the global effect on unemployment is usually insignificant because these firms hold a very low percentage of labor.

This small percentage is, however, important because in periods of growth when unemployment is very low new firms still emerge with man-power requirements, which cannot be met without available labor that has been released from current firms.

The slope of the rise in aggregate salaries is a product of input parameters, mainly the probability of giving raises, the raise amount relative to current employee wage, and the

growth of new workers. Also, since we have a maximum wage (in the default setting equals 20 times the minimum wage), should that wage be achieved for all current workers further raises would become impossible, and the only growth in the aggregate salaries would be due to the introduction of new workers. This effect is inevitable when regarding sensible labor procedures: if unemployment is low, most people stay at their jobs (or move mostly to better-paying jobs, though it is not part of SABLE). Eventually, a person who stays employed will get a raise, though his/her productive ability remains more or less the same on the scope in which we deal. After enough such raises, his/her salary will exceed his/her labor capability.

Next, we discuss the variation in firms' debt. In the current version of SABLE, we do not deal with firms' debt in great detail. In particular, the debt of the firms that are going bankrupt is not being dealt with: it is simply being ignored. This approximation is satisfying for the current level of implementation when our goal is to test the basic functions of economy; however, for a more profound result that can be compared with the empirical analysis [17], it will be necessary to refine the model. The light blue color indicates bankruptcy, as opposed to the dark blue color which indicates no debt (trivially true for nonexistent firms). The end of initialization is indicated by a thick red horizontal line (**Figure 5**).



Figure 5. The logarithm of the debt of each firm (*x*-axis) at each time step (*y*-axis, red line signifies the end of initialization). The light blue color represents the time when firms are bankrupt.

Overall, debt per firm slowly grows in this example (more frequent loans). This happens mostly due to the fact that interest is low, so that firms are expected to get loans. This allows the firms who get loans better chances of success. Then, new firms emerge and get loans before older firms have paid off their current debts, so that momentarily the total debt increases.

Moreover, since more firms are active at each time, the chances that they would loan additional money increase, causing an increase in the normalized debt.

Correlation between behavior in debt and other features has not yet been thoroughly established. However, a preliminary analysis indicates that economy and debt grow together, meaning that when unemployment is low and firms succeed more than they fail—debt increases—and vice versa.

We are also inspecting the number of time steps that firms have been active before going bankrupt. We see that most firms are active for only a single step, and that a very small number of firms live more than 20 steps. This is also quite expected: we do not distinguish between the entrepreneurs and firms; thus the model makes the decision on which of the entrepreneurs starting new businesses will develop until the level of real firms. The growth opportunities are equal for all new firms and not related with the products they are bringing to the market, that is, we are neglecting any industrial classification in this version of the model, that is, the luck is the main ingredient needed to the survival of the small firms and entrepreneurs (similar as in the model of [18]) (**Figure 6**).



Figure 6. The distribution of firms' life span: blue dots represent bankrupt firms, and red dots represent active firms at the end of the simulation run.

This result is sensible enough, since most firms eventually will close, and most firms will be short-lived. Generally, it does not mean that older firms are more powerful in terms of available cash or income.

5. Discussion

The generic runs we performed until now show that a series of properties of the economic systems and of their models are very nonspecific and do not depend on the details of the implementation:

- We obtain relatively stable economies with a number of active firms in the market that fluctuates around an average value corresponding to the economy size, and yet have firms go bankrupt and new firms are created. This is a rather realistic feature of the economy [16].
- The power-law distribution, in particular in the distribution of wealth of the firms, their life span, and the labor they employ, emerges spontaneously without any fine-tuning of the parameters (in particular, this means that most firms are small, and only a few are large).
- We reach sensible unemployment levels, with fluctuations in unemployment coupled to the market fluctuations.
- In particular, when unemployment is low, wages are higher than the labor they buy, meaning that workers work for more than they are actually worth.
- The life span of firms also agrees with the empirical known facts. We obtain a power-law distribution, meaning that most firms are very short-lived, and only a few may stay active for long time periods.
- The current ratio of active firms that resulted from the default run agrees with the typical empirical data [19].

Representing economic reality in a simple computer platform seems to be quite a torturous path toward an elusive target.

However, this effort to capture the essentials of the economic system in a quantitative, intuitive, and conceptually meaningful way is not a void academic exercise. Rather, it offers an interactive and transparent way of revealing the mechanisms that make an economy function, such as the emergence of collective macroscopic phenomena out of individual interactions.

The fact that this is not a barren effort is proven daily by the strong controversies in the news between different economist schools about virtually every social and policy item.

We may hope at some stage to have the quantitative platforms such as SABLE as central participants to these debates by "commenting numerically" and supporting quantitatively the various proposals and verbal arguments raised in the media.

One does not expect a machine to be a definitive judge between the views and principles introduced by different schools or interest groups. Conversely, one can positively expect to check whether the logical conclusions and quantitative relations within a certain theory of narrative add up: presenting simulations that parallel a verbal/policy argument and confirm or contradict a certain position/prediction.

As such, we expect SABLE to contribute to an informed, precise, yet not "academic jargon" analysis: exactly what the society is yearning for in the last years!

At the conceptual level, one may hope to repeat in the economic field the success of natural sciences in the last century: understanding and predicting the macroscopic world (including this time society, economics, collective human behavior) in terms of "microscopic" simple rules acting at the level of the individual agent.

Author details

Alexander Kindler^{1*}, Natasa Golo² and Sorin Solomon³

- *Address all correspondence to: alexanderkindler@gmail.com
- 1 Hebrew University of Jerusalem, Ramat-Gan, Israel
- 2 Labex-Refi, Paris, France
- 3 University of Jerusalem, Jerusalem, Israel

References

- [1] Keen, S. Debunking economics: The naked emperor of the social sciences. 2nd ed. Australia: Pluto Press; 2001.
- [2] Petrongolo, B. and Pissarides, C. Looking into the black box: A survey of the matching function. Journal of Economic Literature. 2001;39:390–431.
- [3] Rogerson, R., Shimer, R. and Wright, R. Search-theoretic models of the labor market: A survey. Journal of Economic Literature. 2005;43:959–988.
- [4] Guerrero, O.A. and López, E. Firm-to-firm labor flows and the aggregate matching function: A network-based test using employer—employee matched records. Economics Letters. 2015;136:9–12.
- [5] Mortensen, D. and Pissarides, C. Job creation and job destruction in the theory of unemployment. The Review of Economic Studies. 1994;61:397–415.
- [6] Wasmer, E. and Weil, P. The macroeconomics of labor and credit market imperfections. American Economic Review. 2004;94(4):944–963.
- [7] Wolf, S., Bouchaud, J.P., Cecconi, F., Cincotti, S., Dawid, H., Gintis, H., van der Hoog, S., Jaeger, C.C., Kovalevsky, D.V., Mandel, A., Paroussos, L. Describing economic agentbased models—Dahlem ABM documentation guidelines. Complexity Economics. 2013;2(1):63–74.

- [8] Muchnik, L. and Solomon, S. Markov nets and the NetLab platform: Application to continuous double auction. Complexity Hints for Economic Policy. 2007(1);157–180.
- [9] Gilbert, N. and Terna, P. How to build and use agent-based models in social science. Mind & Society. 2000;1(1):57–72.
- [10] Levy, H., Levy, M. and Solomon, S. Microscopic simulation of financial markets: From investor behavior to market phenomena. 1st ed. California: Academic Press; 2000.
- [11] Lux, T. The stable Paretian hypothesis and the frequency of large returns: An examination of major German stocks. Applied Financial Economics. 1996;6:463–75.
- [12] Muchnik, L., Louzoun, Y., and Solomon, S. Agent based simulation design principles; applications to stock market. In: Hideki Takayasu, editor. Practical fruits of econophysics. 1st ed. Tokyo: Springer; 2006.
- [13] Golo, N., Bree, D.S., Kelman, G., Ussher, L., Lamieri, M., Solomon, S. Too dynamic to fail: Empirical support for an autocatalytic model of Minsky's financial instability hypothesis. Journal of Economic Interaction and Coordination. 2015;special issue:1–25.
- [14] Levy, M. and Solomon, S. New evidence for the power-law distribution of wealth. Physica A: Statistical Mechanics and its Applications. 1997;242(1):90–94.
- [15] Mantegna, R.N. and Stanley, H.E. Introduction to econophysics: Correlations and complexity in finance. 1st ed. Cambridge: Cambridge University Press; 2000.
- [16] Gabaix, X. Power laws in economics and finance. Annual Review of Economics. 2009;1:255–294.
- [17] Axtell, R.L. Zipf distribution of US firm sizes. Science. 2001;293(5536):1818–1820.
- [18] Bottazzi, G. and Secchi, A. Explaining the distribution of firm growth rates. The RAND Journal of Economics. 2006;37(2):235–256.
- [19] Guy, K., Natasa, G., Marco, L., Leanne, U., David, B., Magda, F., and Sorin, S. Using appearance, disappearance, growth, contraction and accounting ratios of small firms as early warnings for the state of the economy. In: Thomas Gilbert and Gregoire Nicolis, editors. European conference on complex systems; 2–7 September, Brussels; 2012.



Edited by Ciza Thomas

The book on complex systems, sustainability, and innovation explores a broad set of ideas and presents some of the state-of-the-art research in this field concisely in six chapters. In a complex system, it is difficult to know exactly how the individual components contribute to an observed behavior and the extent of each component's contributions. It is the interactions of the individual components that determine the emergent functionalities. This makes it difficult to understand and predict the behavior of complex systems and hence the effects of any innovations in this field. This necessitates for the emergence of a new age of innovations with the main focus on user orientation and sustainability. This book explores some of the complex systems and their dependence on the environment to provide a long-term perspective, aiding innovations and supporting a sustainable society. The intended audience of this book will mainly consist of researchers, research students, and practitioners in the field of complex systems and sustainability.







IntechOpen